United States SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 8-K

Current Report Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

September 29, 2019

Date of Report (Date of earliest event reported)

Health Sciences Acquisitions Corporation

(Exact Name of Registrant as Specified in its Charter)

001-38906

Delaware 83-2771572 (State or other jurisdiction (Commission File Number) (I.R.S. Employer of incorporation) Identification No.) 412 West 15th Street, Floor 9 New York, NY 10011

(Address of Principal Executive Offices)

Registrant's telephone number, including area code: (646) 343-9280

N/A

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

Written communications pursuant to Rule 425 under the Securities Act

 \times Soliciting material pursuant to Rule 14a-12 under the Exchange Act

Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act

Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Units, each consisting of one share of Common Stock, \$0.0001 par value, and one Warrant entitling the holder to receive one half share of Common Stock	HSACU	The Nasdaq Stock Market LLC
Shares of Common Stock, \$0.0001 par value, included as part of the Units	HSAC	The Nasdaq Stock Market LLC
Warrants included as part of the Units	HSACW	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (17 CFR §230.405) or Rule 12b-2 of the Securities Exchange Act of 1934 (17 CFR §240.12b-2).

Emerging growth company 🗵

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

(Zip Code)

Participants in the Solicitation

Immunovant Sciences Ltd. ("Immunovant"), Health Sciences Acquisitions Corporation ("HSAC"), and their respective directors, executive officers and employees and other persons may be deemed to be participants in the solicitation of proxies from the holders of shares of HSAC common stock in respect of the proposed transaction described herein. Information about HSAC's directors and executive officers and their ownership of HSAC common stock is set forth in HSAC's Prospectus dated May 9, 2019 (the "Prospectus") filed with the Securities and Exchange Commission (the "SEC"), as modified or supplemented by any Form 3 or Form 4 filed with the SEC since the date of such filing. Other information regarding the interests of the participants in the proxy solicitation will be included in the proxy statement pertaining to the proposed transaction when it becomes available. These documents can be obtained free of charge from the sources indicated below.

Additional Information and Where To Find It

In connection with the transaction described herein, HSAC will file relevant materials with the SEC, including a proxy statement on Schedule 14A. Promptly after filing its definitive proxy statement with the SEC, HSAC will mail the definitive proxy statement and a proxy card to each stockholder entitled to vote at the special meeting relating to the transaction. INVESTORS AND SECURITY HOLDERS OF HSAC ARE URGED TO READ THESE MATERIALS (INCLUDING ANY AMENDMENTS OR SUPPLEMENTS THERETO) AND ANY OTHER RELEVANT DOCUMENTS IN CONNECTION WITH THE TRANSACTION THAT HSAC WILL FILE WITH THE SEC WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT HSAC, IMMUNOVANT AND THE TRANSACTION. The definitive proxy statement, the preliminary proxy statement and other relevant materials in connection with the transaction (when they become available), and any other documents filed by HSAC with the SEC, may be obtained free of charge at the SEC's website (www.sec.gov) or by writing to Health Sciences Acquisitions Corporation, 412 West 15th Street, Floor 9, New York, NY 10011.

Forward-Looking Statements

This Current Report on Form 8-K and the documents incorporated by reference herein (this "Current Report") contain certain "forward-looking statements" within the meaning of "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995. Forward-looking statements can be identified by words such as: "target," "believe," "expect," "will," "shall," "may," "anticipate," "estimate," "would," "positioned," "future," "forecast," "intend," "plan," "project" and other similar expressions that predict or indicate future events or trends or that are not statements of historical matters. Examples of forward-looking statements include, among others, statements made in this Current Report regarding the proposed transactions contemplated by the share exchange agreement (the "Share Exchange Agreement") among HSAC, Immunovant, Roivant Sciences Ltd., and the stockholders of HSAC (the "Business Combination"), including the anticipated initial enterprise value and post-closing equity value, the benefits of the Business Combination, integration plans, expected synergies and revenue opportunities, anticipated future financial and operating performance and results, including estimates for growth, the expected management and governance of the combined company, and the expected timing of the Business Combination. Forward-looking statements are neither historical facts nor assurances of future performance. Instead, they are based only on HSAC and Immunovant managements' current beliefs, expectations and assumptions. Because forward-looking statements relate to the future, they are subject to inherent uncertainties, risks and changes in circumstances that are difficult to predict and many of which are outside of our control. Actual results and outcomes may differ materially from those indicated in the forward-looking statements. Therefore, you should not rely on any of these forward-looking statements. Important factors that could cause actual results and outcomes to differ materially from those indicated in the forward-looking statements include, among others, the following: (1) the occurrence of any event that could give rise to the termination of the Share Exchange Agreement; (2) the outcome of any legal proceedings that may be instituted against HSAC, the combined company, or others following the announcement of the Business Combination and the Share Exchange Agreement; (3) the inability to complete the Business Combination due to the failure to obtain approval of HSAC's stockholders or to satisfy other conditions to closing in the Share Exchange Agreement; (4) changes to the proposed structure of the Business Combination that may be required or appropriate as a result of applicable laws; (5) the ability to meet the Nasdaq Stock Market LLC ("Nasdaq") listing standards following the consummation of the Business Combination; (6) the risk that the Business Combination disrupts current plans and operations of Immunovant as a result of the announcement and consummation of the Business Combination; (7) the ability to recognize the anticipated benefits of the Business Combination, which may be affected by, among other things, competition, the ability of the combined company to grow and manage growth profitably, maintain relationships with third parties and partners, obtain adequate supply of raw materials and retain its management and key employees; (8) costs related to the Business Combination; (9) changes in applicable laws or regulations; (10) the possibility that Immunovant or the combined company may be adversely affected by other economic, business, regulatory, and/or competitive factors; (11) Immunovant's estimates of expenses; (12) the impact of foreign currency exchange rates and interest rates fluctuations on the results of Immunovant or the combined company; and (13) other risks and uncertainties indicated in the proxy statement of HSAC to be filed by HSAC with the SEC in connection with the Business Combination, including those under "Risk Factors" therein, and other documents filed or to be filed from time to time with the SEC by HSAC.

A further list and description of risks and uncertainties can be found in HSAC's Prospectus dated May 9, 2019 filed with the SEC and in the proxy statement on Schedule 14A that will be filed with the SEC by HSAC in connection with the proposed transaction, and other documents that the parties may file or furnish with the SEC, which you are encouraged to read. Any forward-looking statement made by us in this Current Report is based only on information currently available to HSAC and Immunovant and speaks only as of the date on which it is made. HSAC and Immunovant undertake no obligation to publicly update any forward-looking statement, whether written or oral, that may be made from time to time, whether as a result of new information, future developments or otherwise, except as required by law.

Item 1.01. Entry Into a Material Definitive Agreement

On September 29, 2019, Health Sciences Acquisitions Corporation ("HSAC") entered into a share exchange agreement (the "Share Exchange Agreement") with Immunovant Sciences Ltd. ("Immunovant"), a biopharmaceutical company focused on enabling normal lives for patients with autoimmune diseases, and shareholders of Immunovant, including Roivant Sciences Ltd. ("Roivant").

Acquisition of Immunovant; Acquisition Consideration

Upon the closing of the transactions contemplated in the Share Exchange Agreement (the "Closing"), HSAC will acquire all of the issued and outstanding shares of Immunovant, and Immunovant will become a wholly owned subsidiary of HSAC (the "Business Combination"). Upon the Closing, HSAC will change its name to "Immunovant, Inc."

As a result of the Business Combination, an aggregate of approximately 43,000,000 shares of HSAC common stock will be issued (or reserved for issuance pursuant to currently exercisable options (on a treasury stock method basis)) and 10,000 shares of Series A preferred stock of HSAC to be issued to Roivant) in respect of shares of Immunovant capital stock and exercisable options (on a treasury stock method basis) that are issued and outstanding as of immediately prior to the Closing, subject to pre-closing adjustment for certain indebtedness of Immunovant (other than indebtedness convertible into Immunovant capital stock).

On the closing date of the Business Combination, each option to purchase Immunovant's common shares (each, a "Company Option") that is outstanding under Immunovant's equity incentive plan immediately prior to the Closing, and each option or restricted stock units, whether vested or unvested, will, automatically and without any required action on the part of any holder or beneficiary thereof, be assumed by HSAC and converted into an option to purchase shares of HSAC common stock (each, a "Converted Option"), subject to pre-closing adjustment for certain indebtedness of Immunovant (other than indebtedness convertible into Immunovant's common shares). Each Converted Option shall continue to have and be subject to the same terms and conditions as were applicable to such Company Option immediately before the Closing (including expiration date, vesting conditions, and exercise provisions).



Immediately after the Closing, the combined company's board of directors will consist of seven directors, with six directors identified by Roivant and one director identified by HSAC.

The holder(s) of a majority of outstanding shares of Series A Preferred Stock will be entitled to elect: (i) four directors (the "Series A Preferred Directors"), as long as the holder(s) of Series A Preferred Stock hold 50% or more of the voting power of all then-outstanding shares of capital stock of the combined company entitled to vote generally at an election of directors, (ii) three Series A Preferred Directors, as long as the holder(s) of Series A Preferred Stock hold 40% or more but less than 50% of the voting power of all then-outstanding shares of capital stock of the combined company entitled to vote generally at an election of directors, and (iii) two Series A Preferred Directors, as long as the holder(s) of Series A Preferred Stock hold 25% or more but less than 40% of the voting power of all then-outstanding shares of capital stock of the combined company entitled to vote generally at an election of directors, and (iii) two Series A Preferred Directors, as long as the holder(s) of Series A Preferred Stock hold 25% or more but less than 40% of the voting power of all then-outstanding shares of capital stock of the combined company entitled to vote generally at an election of directors;

Earnout Shares

Pursuant to the Share Exchange Agreement, HSAC agreed to issue Immunovant shareholders the following number of additional earnout shares in the aggregate subject to the achievement of the conditions specified below (each, a "Milestone"):

- a. Following the Closing, if the daily volume-weighted average price of a share of HSAC common stock in any 20 trading days within a 30-trading day period prior to March 31, 2023 is greater than or equal to \$17.50 per share, then HSAC shall issue 10,000,000 shares of HSAC common stock to Immunovant shareholders.
- b. Following the Closing, if the daily volume-weighted average price of a share of HSAC common stock in any 20 trading days within a 30-trading day period prior to March 31, 2025 is greater than or equal to \$31.50 per share, then HSAC shall issue 10,000,000 shares of HSAC common stock Immunovant shareholders.

In the event that after Closing and prior to March 31, 2025 there is an Acceleration Event (as defined in the Share Exchange Agreement), then any earnout shares that have not been previously issued by HSAC (whether or not previously earned) shall be deemed earned and due by HSAC to the Immunovant shareholders, unless in a change of control, the value of the consideration to be received in exchange for a share of HSAC common stock is lower than the applicable Milestone share price thresholds described above.

HSAC Stockholder Approval

Prior to the Closing, the holders of (i) a majority of the issued and outstanding shares of HSAC common stock must approve HSAC's charter amendment and (ii) a majority of the issued and outstanding shares of HSAC common stock present in person or by proxy and entitled to vote at the HSAC special meeting must approve the Business Combination and the 2019 HSAC Equity Incentive Plan and the issuance of more than 20% of the issued and outstanding shares of HSAC common stock pursuant to the Share Exchange Agreement, pursuant to Nasdaq requirements (collectively, the "Stockholder Approval").

In addition, HSAC must prepare and file with the SEC a Proxy Statement on Schedule 14A (the "Proxy Statement") that will be mailed to all stockholders entitled to vote at the HSAC special meeting. HSAC must call a special meeting of its common stockholders as promptly as reasonably practicable following the filing of the definitive Proxy Statement.

Representations and Warranties and Covenants

The Share Exchange Agreement contains customary representations, warranties and covenants made by the each party, including covenants relating to obtaining the requisite Stockholder Approval, indemnification of directors and officers, limitations on the solicitation of alternative proposals and change of board recommendations, and HSAC's and Immunovant's conduct of their respective businesses between the date of signing of the Share Exchange Agreement and the Closing (or the termination of the Share Exchange Agreement). In addition, between the date of the signing of the Share Exchange Agreement and the Closing (or the termination of the Share Exchange Agreement), neither Immunovant nor HSAC is allowed to enter into a financing transaction or any agreement relating to the sale of such party's assets or equity securities, or a merger or change of control agreement with respect to such party or its assets, without the prior written consent of the other party, other than certain Immunovant permitted financings and licensing by Immunovant in the ordinary course of business.

The representation and warranties contained in the Share Exchange Agreement will not survive the Closing, other than for the sole purpose of recovery under the representation and warranty insurance policy further described below.

Conditions to Closing

General Conditions

The obligation of HSAC and Immunovant to consummate the Business Combination is conditioned on, among other things: (a) the absence of any law, order, stay, judgment or decree by any government agency restraining or prohibiting or imposing any condition on the closing of the Business Combination; (b) the absence of any action brought by a governmental agency seeking to enjoin or otherwise restrict the consummation of the Business Combination; (c) expiration of any waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder; and (d) payment of the transaction expenses of each of HSAC and Immunovant.

HSAC's Additional Conditions to Closing

In addition, the obligation of HSAC to consummate the Business Combination is conditioned upon, among other things: (a) Immunovant's performance its obligations under the Share Exchange Agreement in all material respects; (b) accuracy of the representations and warranties of Immunovant subject to material adverse effect qualification; (c) absence of a material adverse effective on Immunovant that is continuing; (d) HSAC's receipt of any updated schedules from Immunovant; and (e) the effectiveness of the representation and warranties insurance policy pursuant to its terms.

Immunovant's Additional Conditions to Closing

In addition, the obligation of Immunovant to consummate the Business Combination is conditioned upon, among other things: (a) HSAC's performance of its obligations under the Share Exchange Agreement in all material respects and the accuracy of the representations and warranties of HSAC; (b) forfeiture and cancellation of HSAC's private warrants; (c) a minimum of \$65 million cash on HSAC's balance sheet after any redemption of shares of HSAC common stock; (d) receipt of a letter from Nasdaq indicating that the combined company and the transaction securities have been approved for listing; (e) appointment of the post-Closing board of directors; (f) amendment of the certificate of incorporation of HSAC; and (g) the Stockholder Approval.

Representation and Warranty Insurance Policy

In conjunction with entering into the Share Exchange Agreement, HSAC entered into a binder agreement for a representation and warranty insurance policy concurrently with executing the Share Exchange Agreement. The insurance policy was bound by Indian Harbor Insurance Company, with a policy limit of no less than \$10 million and a retention amount no greater than approximately \$900,000.



Termination

The Share Exchange Agreement may be terminated at any time prior to the closing by:

- HSAC or Immunovant, if the closing has not occurred on or prior to January 31, 2020 (the "Outside Closing Date"); provided, however, that a party shall not be permitted to
 provide notice of termination if the failure of the closing to occur prior to the Outside Closing Date is attributable to the failure on the part of such party to perform in any
 material respect any covenant or obligation in the Share Exchange Agreement required to be performed by such party;
- HSAC or Immunovant, in the event an governmental authority shall have issued an order, having the effect of permanently restraining, enjoining or otherwise prohibiting the Business Combination, which order is final and non-appealable; provided, however, that a party shall not be permitted to so terminate if such event is attributable to the failure on the part of such party to perform in any material respect any covenant or obligation in the Share Exchange Agreement required to be performed by such party;
- HSAC or Immunovant, in the event that HSAC fails to receive the Stockholder Approval at the HSAC stockholder meeting (subject to any adjournment or recess of such special meeting); provided that HSAC shall not be permitted to terminate if the failure to obtain such Stockholder Approval is proximately caused by any action or failure to act of HSAC that constitutes a breach of the Share Exchange Agreement;
- the mutual written agreement of Immunovant and HSAC;
- HSAC, if: (i) Immunovant shall have breached any representation, warranty, agreement or covenant contained in the Share Exchange Agreement to be performed on or prior to the Outside Closing Date, which has rendered the satisfaction of any of the applicable closing conditions impossible; and (ii) such breach shall not be cured by the earlier of the Outside Closing Date and thirty (30) days following receipt by Immunovant of a written notice from HSAC describing in reasonable detail the nature of such breach, except HSAC will not be allowed to so terminate if it is then in material breach of any representation, warranty, agreement or covenant;
- Immunovant, if: (i) HSAC shall have breached any of its covenants, agreements, representations, and warranties contained herein to be performed on or prior to the Outside Closing Date, which has rendered the satisfaction of any of the applicable closing conditions impossible; and (ii) such breach shall not be cured by the earlier of the Outside Closing Date and thirty (30) days following receipt by HSAC of a written notice from Immunovant describing in reasonable detail the nature of such breach, except Immunovant will not be allowed to so terminate if it is then in material breach of any representation, warranty, agreement or covenant; or
- Immunovant or HSAC, if a Triggering Event with respect to the other party shall have occurred. A "Triggering Event" shall include (i) a change of board recommendation,
 (ii) HSAC failing to convene or hold the HSAC stockholder meeting, (iii) breach of the non-solicitation clause; and (iv) failure of the HSAC board of directors to reaffirm the its recommendation under certain circumstances.

The foregoing summary of the Share Exchange Agreement does not purport to be complete and is qualified in its entirety by reference to the actual Share Exchange Agreement, which is filed as Exhibit 2.1 hereto, and which is incorporated by reference into this Current Report. Terms used herein as defined terms and not otherwise defined herein shall have the meanings ascribed to them in the Share Exchange Agreement.

The Share Exchange Agreement has been included to provide investors and stockholders with information regarding its terms. It is not intended to provide any other factual information about HSAC, Immunovant or any other parties thereto. The Share Exchange Agreement contains representations and warranties that the parties thereto made to, and solely for the benefit of, each other. The assertions embodied in such representations and warranties are qualified by information contained in the confidential disclosure schedules that each may have delivered to the other party in connection with signing the Share Exchange Agreement. Accordingly, investors and stockholders should not rely on such representations and warranties as characterizations of the actual state of facts or circumstances, since they were only made as of the date of the Share Exchange Agreement and are modified in important part by the underlying disclosure schedules. Moreover, information concerning the subject matter of such representations and warranties may change after the date of the Share Exchange Agreement, which subsequent information may or may not be fully reflected in HSAC's public disclosures.

Additional Agreements

In addition to the Share Exchange Agreement:

- 1. HSAC and its sponsor, Health Sciences Holdings, LLC ("Sponsor") entered into a Sponsor Restricted Stock Agreement pursuant to which concurrently with the Closing, the Sponsor will (a) forfeit a number of shares of HSAC common stock equal to: (A) 1,800,000, multiplied by (B) (i) the number of shares of HSAC common stock validly redeemed by holders thereof in connection with the Business Combination as reflected in the records of HSAC's transfer agent, divided by (ii) 11,500,000 (such number of shares, the "Canceled Shares"), and (b) subject a number of shares of HSAC common stock equal to 1,800,000 minus the Canceled Shares (the "Sponsor Earnout Shares") to potential forfeiture in the event that the Milestones are not achieved. In the event of an Acceleration Event (as defined in the Share Exchange Agreement), all of the Sponsor Earnout Shares shall vest and no longer be subject to forfeiture, unless in a change of control, the value of the consideration to be received in exchange for a share of HSAC common stock is lower than the applicable Milestone share price thresholds described above. Any Sponsor Earnout Shares that have not vested on or prior to March 31, 2025 will be forfeited by the Sponsor after such date.
- 2. Each Immunovant shareholder has entered into a Lock-up Agreement with HSAC, in substantially the form attached to the Share Exchange Agreement, with respect to shares of HSAC common stock (or any securities convertible into, or exchangeable for, or representing the rights to receive shares of HSAC common stock) to be received by it in the Business Combination or during the Lock-up Period (as defined below) (such shares, the "Lockup Shares"). In such Lock-up Agreement, each Immunovant shareholder has agreed that during the Lock-up Period, it will not offer, sell, contract to sell, pledge or otherwise dispose of, directly or indirectly, any Lock-up Shares), enter into a transaction that would have the same effect, or enter into any swap, hedge or other arrangement that transfers, in whole or in part, any of the economic consequences of ownership of Lock-up Shares, whether any of these transactions are to be settled by delivery of any Lock-up Shares, in cash or otherwise, publicly disclose the intention to make any offer, sale, pledge or disposition, or to enter into any transaction, swap, hedge or other arrangement, or engage in any short sales with respect to any security of HSAC.

The "Lock-up Period" means: (i) with respect to 50% of the Lock-up Shares, the shorter of (A) the period commencing on the date of Closing and ending on the date that is six months thereafter; and (B) the period commencing on the date of Closing and ending on the date on which the last reported closing price of the HSAC Shares on the Nasdaq (or such other exchange on which the HSAC Shares are then listed) equals or exceeds \$12.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days during any 30 trading day period thereafter; and (ii) with respect to the remaining 50% of the Lock-up Shares, the period commencing on the date of Closing and ending on the date that is six months thereafter. In addition, if within six months after the date of Closing , there is a Change of Control (as defined in the Share Exchange Agreement), then upon the consummation of such Change of Control, all Lock-up Shares shall be released from the foregoing restrictions.

3. HSAC and Immunovant stockholders and the Sponsor have entered into an amended and restated registration rights agreement, in substantially the form attached to the Share Exchange Agreement (the "Registration Rights Agreement"). Under the Registration Rights Agreement, the Immunovant shareholders and the Sponsor hold registration rights that obligate HSAC to register for resale under the Securities Act, all or any portion of the HSAC Shares issued under the Share Exchange Agreement, including any Earnout Payments, as well as HSAC Shares held by the Sponsor or issuable upon the exercise of warrants held by the Sponsor. Each of the Sponsor and Roivant, as well as the stockholders holding a majority-in-interest of all such registrable securities will be entitled to make a written demand for registration under the Securities Act of all or part of their registrable securities, so long as such shares are not then restricted under the Securities Act with respect to certain exceptions, if any time after the Closing, the combined company proposes to file a registration statement under the Securities Act with respect to its securities, under the Registration Rights Agreement, the combined company after the Business Combination shall give notice to the Immunovant shareholders and the Sponsor as to the proposed filing and offer such stockholders an opportunity to register the sale of such number of their registrable securities as they request in writing. In addition, subject to certain exceptions, such stockholders are entitled under the Registration Rights Agreement to request in writing that HSAC register the resale of any or all of their registrable securities on Form S-3 and any similar short-form registration statement that may be available at such time.

Except as described above, no consideration was paid by HSAC in connection with the agreements described above.

In addition:

- HSAC entered into voting agreements with holders of 4,547,000 shares of HSAC common stock pursuant to which such stockholders, including but not limited to the RTW Investments, LP, Adage Capital Management, Cormorant Asset Management, Eventide Asset Management, LLC, and Perceptive Advisors, agreed to vote in favor of the transactions contemplated by the Share Exchange Agreement and to not redeem or sell their shares.
- 2. HSAC entered into agreements with other investors that agreed to purchase up to 2,374,400 shares of HSAC common stock at HSAC's request and not to redeem such shares of HSAC common stock in connection with the closing of the Business Combination.

Item 3.02. Unregistered Sales of Equity Securities

The disclosure set forth above in Item 1.01 of this Current Report on Form 8-K is incorporated by reference herein. The securities of HSAC to be issued in connection with the Business Combination and additional agreements will not be registered under the Securities Act in reliance upon the exemption provided in Section 4(a)(2) of the Securities Act.

Item 7.01. Regulation FD Disclosure

Attached as Exhibit 99.1 to this Current Report on Form 8-K and incorporated into this Item 7.01 by reference is a copy of the press release issued October 2, 2019 announcing the proposed transaction.



Attached as Exhibit 99.2 to this Current Report on Form 8-K and incorporated into this Item 7.01 by reference is the investor presentation dated October 2, 2019 that will be used by HSAC in making presentations to certain existing and potential stockholders of HSAC with respect to the proposed transaction.

Exhibits 99.1 and 99.2 are being furnished pursuant to Item 7.01 and shall not be deemed to be filed for purposes of Section 18 of the Exchange Act or otherwise be subject to the liabilities of that section, nor shall it be deemed to be incorporated by reference in any filing under the Securities Act or the Exchange Act.

Item 9.01. Financial Statements and Exhibits

	(d)	Exhibits.
Exhibit No.		Description
2.1*		Share Exchange Agreement, dated as of September 29, 2019, by and among Immunovant Sciences Ltd., the stockholders of Immunovant Sciences Ltd., and Health Sciences Acquisitions Corporation.
99.1**		Press Release dated October 2, 2019
99.2**		Investor Presentation dated October 2, 2019

*Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The registrant hereby undertakes to furnish copies of any of the omitted schedules and exhibits upon request by the U.S. Securities and Exchange Commission.

**Furnished but not filed.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated October 2, 2019

HEALTH SCIENCES ACQUISITIONS CORPORATION

By: /s/ Roderick Wong

Name:Roderick Wong, MDTitle:Chief Executive Officer

SHARE EXCHANGE AGREEMENT

dated

September 29, 2019

by and among

Immunovant Sciences Ltd., a Bermuda exempted limited company,

the stockholders of the Company,

Roivant Sciences Ltd., a Bermuda exempted limited company,

and

Health Sciences Acquisitions Corporation, a Delaware corporation

TABLE OF CONTENTS

		Page
ARTICLE I I	DEFINITIONS	1
ARTICLE II	SHARE EXCHANGE	9
2.1	Share Exchange	9
2.2	Stock Option Conversion	10
2.3	Closing	10
2.4	Board of Directors of Purchaser	10
2.5	Taking of Necessary Action; Further Action	10
2.6	Section 368 Reorganization	10
2.7	Purchaser Charter Documents	10
ARTICLE II	I CONSIDERATION	11
3.1	Closing Payment Shares	11
3.2	Pre-Closing Adjustment for Closing Indebtedness	11
3.3	Earnout Payment	12
ARTICLE IV	REPRESENTATIONS AND WARRANTIES OF THE COMPANY	13
4.1	Corporate Existence and Power	13
4.2	Authorization	13
4.3		13
	Governmental Authorization	
4.4	Non-Contravention	14
4.5	Capitalization	14
4.6	Charter Documents	14
4.7	Corporate Records	14
4.8	Third Parties	14
4.9	Subsidiaries	15
4.10	Financial Statements	15
4.11	Absence of Certain Changes	16
4.12	Properties; Title to the Company's Assets	16
4.13	Litigation	16
4.14	Contracts	17
4.15	Insurance	18
4.16	Licenses and Permits	18
4.17	Compliance with Laws	18
4.18	Intellectual Property	18
4.19	Employees	19
4.20	Employees Employment Matters	19
4.20	Withholding	20
4.22		20
4.23	Employee Benefits and Compensation Real Property	21
4.23	Tax Matters	21
4.25	Environmental Laws	22
4.26	Finders' Fees	22
4.27	Preclinical Development and Clinical Trials	22
4.28	FDA Approvals	22
4.29	FDA Regulation	23
4.30	Information Supplied	23

	Page
ARTICLE V REPRESENTATIONS AND WARRANTIES OF PURCH.	ASER 23
5.1 Corporate Existence and Power	23
5.2 Corporate Authorization	23
5.3 Governmental Authorization	24
5.4 Non-Contravention	24
5.5 Finders' Fees	24
5.6 Issuance of Shares	24
5.7 Capitalization	24
5.8 Information Supplied	25
5.9 Trust Fund	25
5.10 Listing	25
5.11 Board Approval	25
5.12 Purchaser SEC Documents and Financial Statements	25
5.13 Compliance with Laws	26
5.14 OFAC	26
5.15 Litigation	27
5.16 Absence of Certain Changes	27
5.17 Employees and Employee Benefit Plans	27
5.18 Properties	27
5.19 Contracts	27
5.20 Insurance	27
5.21 Taxes	28
5.22 Independent Investigation	28
ARTICLE VI REPRESENTATIONS AND WARRANTIES OF THE ST	OCKHOLDERS 28
6.1 Ownership of Stock; Authority	28
6.2 Approvals	29
6.3 Non-Contravention	29
6.4 Litigation and Claims	29
6.5 Investment Representations	29
*	
ARTICLE VII COVENANTS OF THE PARTIES PENDING CLOSING	
7.1 Conduct of the Business	29
7.2 Annual and Interim Financial Statements	32
7.3 Trust Account	32
7.4 Purchaser Public Filings	32
7.5 Form 8-K; Press Releases	32
7.6 Section 16 of the Exchange Act	32
7.7 No Solicitation	32
7.8 Transfer Taxes	33
Regulatory Compliance 7.9	33
ARTICLE VIII COVENANTS OF THE COMPANY AND PURCHASE	R 34
8.1 Access to Information	34
8.2 Exclusivity	34
8.3 Notices of Certain Events	34
ARTICLE IX COVENANTS OF ALL PARTIES HERETO	35
9.1 Proxy Statement	35
9.2 Purchaser Stockholder Meeting; Board Recommendation	on 36
9.3 Further Assurances	36
9.4 Compliance with SPAC Agreements	36

		Page
9.5	Confidentiality	36
9.6	Registration Statement	37
9.7	Indemnification; Insurance	37
9.8	R&W Insurance Policy	38
RTICLE X	CONDITIONS TO CLOSING	38
10.1	Condition to the Obligations of the Parties	38
10.2	Conditions to Obligations of Purchaser	38
10.3	Conditions to Obligations of the Company	38
RTICLE XI	DISPUTE RESOLUTION	39
11.1	Arbitration	39
11.2	Waiver of Jury Trial; Exemplary Damages	40
RTICLE XI	I TERMINATION	40
12.1	Termination Without Default	40
12.2	Termination Upon Default	41
12.3	Termination for Triggering Event	41
12.4	Effect of Termination	41
RTICLE XI	II MISCELLANEOUS	41
13.1	Notices	41
13.2	Amendments; No Waivers; Remedies	43
13.3	Arm's length bargaining; no presumption against drafter	43
13.4	Publicity	43
13.5	Expenses	44
13.6	No Assignment or Delegation	44
13.7	Governing Law	44
13.8	Counterparts; facsimile signatures	44
13.9	Entire Agreement	44
13.10	Severability	44
13.11	Construction of certain terms and references; captions	44
13.12	Further Assurances	45
13.13	Third Party Beneficiaries	45
13.14	Waiver	45
13.15	Stockholders' Representative	46
13.16	Non-Recourse	46
13.17	Acknowledgement by Purchaser	46
13.18	Specific Performance	47
13.19	Acknowledgement; Waiver of Conflicts; Retention of Privilege	47
13.20	No Survival	48

SHARE EXCHANGE AGREEMENT

This SHARE EXCHANGE AGREEMENT (the "<u>Agreement</u>"), dated as of September 29, 2019, by and among Immunovant Sciences Ltd., a Bermuda registered exempted limited company (the "<u>Company</u>"), the stockholders of the Company (each, a "<u>Stockholder</u>" and, collectively, the "<u>Stockholders</u>"), Roivant Sciences Ltd., a Bermuda exempted limited company (the "<u>Stockholders</u>"), and Health Sciences Acquisitions Corporation, a Delaware corporation (the "<u>Purchaser</u>").

WITNESETH:

A. The Company and its Subsidiaries (the "<u>Company Group</u>") are in the business of developing innovative therapies for patients suffering from debilitating autoimmune diseases (the "<u>Business</u>");

B. The Purchaser is a blank check company formed for the sole purpose of entering into a share exchange, asset acquisition, share purchase, recapitalization, reorganization or other similar business combination with one or more businesses or entities;

C. The Stockholders own all of the Company's outstanding equity securities;

D. The parties desire that the Purchaser purchase 100% of the Company's outstanding equity securities from the Stockholders in exchange for the Closing Payment Shares, subject to the terms and conditions set forth herein (the "Share Exchange");

E. Concurrently with this Agreement, the Sponsor Restricted Stock Agreement, Lock-Up Agreements and Registration Rights Agreement have been executed by the parties thereto; and

F. The parties intend for the Share Exchange to qualify for U.S. federal income tax purposes as a "reorganization" within the meaning of Section 368(a) of the Code and the Treasury Regulations promulgated thereunder to which each of the Company and the Purchaser are to be parties under Section 368(b) of the Code.

The parties accordingly agree as follows:

ARTICLE I DEFINITIONS

The following terms, as used herein, have the following meanings:

- 1.1 "AAA" has the meaning set forth in Section 3.2(c).
- 1.2 "Acceleration Event" has the meaning set forth in Section 3.3(c).

1.3 "Acquisition Proposal" means any agreement, offer, proposal or indication of interest (other than this Agreement and the transactions contemplated hereby), or any public announcement of intention to enter into any such agreement or of (or intention to make) any offer, proposal or indication of interest, relating to, or involving: (A) any acquisition or purchase by any Person or Group, directly or indirectly, of any securities of the Purchaser or any member of the Company Group, or any tender offer or exchange offer, (B) any merger, consolidation, business combination, share exchange or similar transaction involving the Purchaser or any member of the Company Group, (C) any sale, acquisition, disposition, mortgage, pledge or other transfer of any of the assets of the Purchaser or any member of the Company Group, (D) any strategic joint venture, partnership, joint development, license or similar transaction or (E) any liquidation or dissolution of the Purchaser or any member of the Company Group, or any extraordinary dividend, whether of cash or other property, in each case of the foregoing clauses, in any single transaction or series of related transactions; provided, for the avoidance of doubt, that in no event shall (i) any issuance of Company Common Shares upon exercise of Company Options, (ii) any grant of Company Options in the ordinary course of business, (iii) any issuance or incurrence of any Company Convertible Debt or the conversion to the underlying securities pursuant to the terms of such Company Convertible Debt, (iv) any equity issuance pursuant to Schedule 1.23 or (v) any other bona fide financing by the Company (subject to the Purchaser's consent, not to be unreasonably withheld, conditioned or delayed) be deemed an Acquisition Proposal (each, a "Company Permitted Financing").

1.4 "<u>Action</u>" means any legal action, suit, claim, investigation, hearing or proceeding, including any audit, claim or assessment for Taxes or otherwise, by or before any Authority.

1.5 "<u>Additional Agreements</u>" means the Registration Rights Agreement, Lock-Up Agreements, the Sponsor Restricted Stock Agreement and Indemnification Agreements.

1.6 "Advised Parties" has the meaning set forth in Section 13.19(b).

1.7 "<u>Affiliate</u>" means, with respect to any Person, any other Person directly or indirectly Controlling, Controlled by, or under common Control with such Person (but excluding, with respect to the Company, any portfolio companies of venture capital or investment funds that are, or otherwise affiliated with, Stockholders, which portfolio companies may otherwise be deemed to be under common Control with the Company).

- 1.8 "Arbitrator" has the meaning set forth in Section 11.1(a).
- 1.9 "Audited Financial Statements" has the meaning set forth in Section 4.10(a).

1.10 "Authority" means any governmental, quasi-governmental, regulatory or administrative body, agency, instrumentality, department or authority, any court, tribunal, judicial authority, administrative hearing body, arbitrator, commission or other similar dispute-resolving panel or body, or any public, private or industry regulatory authority, whether international, national, Federal, state, or local.

- 1.11 "Balance Sheet Date" has the meaning set forth in Section 4.10(a).
- 1.12 "Balance Sheet" has the meaning set forth in Section 4.10(a).

1.13 "<u>Books and Records</u>" means all books and records, ledgers, employee records, customer lists, files, correspondence, and other records of every kind (whether written, electronic, or otherwise embodied) owned or controlled by a Person in which a Person's assets, the business or its transactions are otherwise reflected, other than stock books and minute books.

- 1.14 "Business" has the meaning set forth in the Recitals.
- 1.15 "Business Combination" has the meaning set forth in Section 5.11.

1.16 "<u>Business Day</u>" means any day other than a Saturday, Sunday or a legal holiday on which commercial banking institutions in New York, New York, are authorized to close for business.

1.17 "<u>Cash Closing Requirement</u>" shall mean an amount in cash equal to \$65 million, which such amount shall include funds remaining in the Purchaser's Trust Account (net of any redemptions of Purchaser Common Stock (the "<u>Redemptions</u>")). Such amount does not include any amount raised by the Company in a bridge financing prior to the date of this Agreement.

- 1.18 "Change of Control" has the meaning set forth in Section 3.3(c).
- 1.19 "Change of Recommendation" has the meaning set forth in Section 7.7.
- 1.20 "Charter Documents" has the meaning set forth in Section 4.6.
- 1.21 "Closing Date" has the meaning set forth in Section 2.3.

1.22 "<u>Closing Indebtedness</u>" means, as of the Reference Time, the aggregate amount of all Indebtedness for borrowed money of the Company Group, on a consolidated basis, determined in accordance with U.S. GAAP in excess of the Company Base Indebtedness.

1.23 "<u>Closing Payment Shares</u>" means stock certificates representing, in the aggregate, (a) 43,000,000 shares of Purchaser Common Stock less (i) the number of shares of Purchaser Preferred Stock issued to Roivant Sciences Ltd. in connection with the Closing and (ii) the number of shares of Purchaser Common Stock to which the Company Options are entitled pursuant to <u>Section 2.2(a)</u>, as calculated on a treasury stock method, less (b) the quotient of the Stockholders' pro rata portion of the Closing Indebtedness divided by \$10.00, provided, that the foregoing calculation shall be made on an iterative basis, issuable to the Stockholders and in such amounts set forth opposite each Stockholder's name on <u>Schedule 1.23</u>, with a deemed price per share of no less than \$10.00

(which amounts shall be calculated based upon the Company's capitalization as of the date of this Agreement). Company will provide to Purchaser an updated <u>Schedule 1.23</u> at least five (5) Business Days prior to Closing (which amounts shall be calculated based upon the Company's capitalization as of the Closing).

- 1.24 "Closing Press Release" has the meaning set forth in Section 7.5(b).
- 1.25 "Closing Statement" has the meaning set forth in Section 3.2(a).
- 1.26 "Closing" has the meaning set forth in Section 2.3.

1.27 "<u>COBRA</u>" means collectively, the requirements of Sections 601 through 606 of ERISA and Section 4980B of the Code.

- 1.28 "Code" means the Internal Revenue Code of 1986, as amended.
- 1.29 "Company Base Indebtedness" means \$2.9 million.
- 1.30 "Company Board" has the meaning set forth in Section 7.7.
- 1.31 "Company Certificate" has the meaning set forth in Section 10.2(d).
- 1.32 "Company Common Shares" has the meaning set forth in Section 4.5.
- 1.33 "Company D&O Policy" has the meaning set forth in Section 9.7(b).
- 1.34 "Company Employees" means the current employees of the Company as of the Closing.
- 1.35 "Company Group" has the meaning set forth in the Recitals.
- 1.36 "Company Option" has the meaning set forth in Section 2.2(a).

1.37 "<u>Contracts</u>" means the Leases and all other contracts, agreements, leases (including equipment leases, car leases and capital leases), licenses, commitments, client contracts, statements of work (SOWs), oral or written, to which any member of the Company Group is a party or by which any of its respective assets are bound, and all rights and benefits thereunder, including all rights and benefits thereunder with respect to all cash and other property of third parties under the Company Group's dominion or control.

1.38 "Control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract, or otherwise. "Controlled", "Controlling" and "under common Control with" have correlative meanings. Without limiting the foregoing, a Person (the "Controlled Person") shall be deemed Controlled by (a) any other Person (the "10% Owner") (i) owning beneficially, as meant in Rule 13d-3 under the Exchange Act, securities entitling such Person to cast 10% or more of the votes for election of directors or equivalent governing authority of the Controlled Person; (b) an officer, director, general partner, partner (other than a limited partner), manager, or member (other than a member having no management authority that is not a 10% Owner) of the Controlled Person; or (c) a spouse, parent, lineal descendant, sibling, aunt, uncle, niece, nephew, mother-in-law, father-in-law, or brother-in-law of an Affiliate of the Controlled Person or a trust for the benefit of an Affiliate of the Controlled Person or of which an Affiliate of the Controlled Person is a trustee.

- 1.39 "Converted Option" has the meaning set forth in Section 2.2(a).
- 1.40 "Data Protection Laws" has the meaning set forth in Section 4.17(a)(i).
- 1.41 "Deal Communications" has the meaning set forth in Section 13.19(c).

1.42 "Deferred Underwriting Amount" means the portion of the underwriting discounts and commissions held in the Trust Account, which the underwriters of the IPO are entitled to receive upon the Closing in accordance with the Trust Agreement.

1.43 "Determination Date" has the meaning set forth in Section 3.2(a).

1.44 "<u>Disclosure Schedule</u>" mean the disclosure schedule that has been prepared by the Company in accordance with the requirements of this Agreement and that has been delivered by the Company to Purchaser on the date of this Agreement.

1.45 "Earnout Period" has the meaning set forth in Section 3.3(b).

1.46 "Earnout Shares" has the meaning set forth in Section 3.3(a).

1.47 "<u>Environmental Laws</u>" shall mean all Laws that prohibit, regulate or control any Hazardous Material or any Hazardous Material Activity, including the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, the Resource Recovery and Conservation Act of 1976, the Federal Water Pollution Control Act, the Clean Air Act, the Hazardous Materials Transportation Act and the Clean Water Act.

1.48 "<u>Equity Award</u>" means the options and restricted stock units outstanding under the Equity Incentive Plan.

1.49 "Equity Incentive Plan" means the Company's 2018 Equity Incentive Plan.

1.50~ "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations thereunder.

- 1.51 "Estimated Closing Indebtedness" has the meaning set forth in Section 3.2(a).
- 1.52 "Exchange Act" means the Securities Exchange Act of 1934, as amended.

1.53 "FDA" has the meaning set forth in Section 4.28.

1.54 "FDA Application Integrity Policy" has the meaning set forth in Section 4.28.

- 1.55 "Financial Statements" has the meaning set forth in Section 4.10(a).
- 1.56 "Foreign Corrupt Practices Act" has the meaning set forth in Section 4.17(a)(ii).
- 1.57 "GAAP" means generally accepted accounting principles, consistently applied.
- 1.58 "Governmental Approval" has the meaning set forth in Section 4.3.
- 1.59 "Group" has the meaning set forth in Section 3.3(c).

1.60 "<u>Hazardous Material</u>" shall mean any material, emission, chemical, substance or waste that has been designated by any Authority to be radioactive, toxic, hazardous, a pollutant or a contaminant.

1.61 "<u>Hazardous Material Activity</u>" shall mean the transportation, transfer, recycling, storage, use, treatment, manufacture, removal, remediation, release, exposure of others to, sale, labeling, or distribution of any Hazardous Material or any product or waste containing a Hazardous Material, or product manufactured with ozone depleting substances, including, any required labeling, payment of waste fees or charges (including so-called e-waste fees) and compliance with any recycling, product take-back or product content requirements.

1.62 "<u>HSR Act</u>" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

1.63 "Indebtedness" means with respect to any Person, (a) all obligations of such Person for borrowed money, or with respect to deposits or advances of any kind (including amounts by reason of overdrafts and amounts owed by reason of letter of credit reimbursement agreements), including with respect thereto, all interests, fees and costs, (b) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person under conditional sale or other title retention agreements relating to property purchased by such Person, (d) all Indebtedness of others secured by (or for which the holder of such Indebtedness has an existing right, contingent or otherwise, to be secured by) any lien or security interest on property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed, (e) all guarantees by such Person and (f) any agreement to incur any of the same. For the avoidance of doubt, Indebtedness shall not include Taxes. With respect to the Company, Indebtedness shall exclude any of the foregoing obligations that will be converted

into equity securities of the Company on or prior to the Closing and will be entitled to the Closing Payment Shares, including such obligations set forth in <u>Schedule 1.63</u> (such excluded obligations, the "<u>Company</u> <u>Convertible Debt</u>").

- 1.64 "Independent Accountant" has the meaning set forth in Section 3.2(c).
- 1.65 "Insured Parties" has the meaning set forth in Section 9.7(b).

1.66 "Intellectual Property Right" means any trademark, service mark, registration thereof or application for registration therefor, trade name, license, invention, patent, patent application, trade secret, trade dress, know-how, copyright, copyrightable materials, copyright registration, application for copyright registration, software programs, data bases, u.r.l.s., and any other type of proprietary intellectual property right, and all embodiments and fixations thereof and related documentation, registrations and franchises and all additions, improvements and accessions thereto, and with respect to each of the forgoing items in this definition, which is owned or licensed or filed by any member of the Company Group, or used or held for use in the Business, whether registered or unregistered or domestic or foreign.

- 1.67 "Interim Period" has the meaning set forth in Section 7.1.
- 1.68 "<u>IPO</u>" means the initial public offering of the Purchaser pursuant to a prospectus dated May 9, 2019.
 - 1.69 "IRS" means the U.S. Internal Revenue Service.

1.70 "Law" means any domestic or foreign, federal, state, municipality or local law, statute, ordinance, code, rule, regulation, legislation, principle of common law, edict, decree, treaty, or Order that is or has been issued, enacted, adopted, passed, approved, promulgated, made, implemented or otherwise put into effect by or under the authority of any Authority.

1.71 "Leases" means the leases set forth on <u>Schedule 1.71</u> attached hereto, together with all fixtures and improvements erected on the premises leased thereby.

1.72 "<u>Lien</u>" means, with respect to any property or asset, any mortgage, lien, pledge, charge, security interest or encumbrance of any kind in respect of such property or asset, and any conditional sale or voting agreement or proxy, including any agreement to give any of the foregoing.

1.73 "<u>Lock-Up Agreements</u>" means the Lock-Up Agreements between the Purchaser and each of the Stockholders executed as of the date hereof (to be effective upon the Closing) in the form attached hereto as <u>Exhibit A</u>.

1.74 "Material Adverse Effect" means a material adverse change or a material adverse effect, individually or in the aggregate, upon on the assets, financial condition, business or operations of the Company Group, taken as a whole; provided, however, that none of the following shall be deemed in and of themselves, either alone or in combination, to constitute, and none of the following shall be taken into account in determining whether there is, or would reasonably likely to be, a Material Adverse Effect on the Company Group: (i) any event, occurrence, violation, inaccuracy, circumstance or other matter directly resulting from the announcement or pendency of the transactions contemplated by this Agreement (other than for purposes of any representation or warranty contained in Section 4.4 but subject to disclosures in Schedule 4.4 of the Disclosure Schedule); (ii) any event, occurrence, circumstance, change or effect in the industries in which the Company Group operates or in the economy generally or other general business, financial or market conditions, except to the extent that the Company Group is adversely affected materially disproportionately relative to the other participants in such industries or the economy generally, as applicable; (iii) any event, circumstance, change or effect arising directly or indirectly from or otherwise relating to any act of terrorism, war, national or international calamity or any other similar event, except to the extent that such event, circumstance, change or effect materially disproportionately affects the Company Group relative to other participants in the industries in which the Company Group operates or the economy generally, as applicable; (iv) the failure of the Company to meet internal expectations or projections or the results of operations of the Company Group; (v) any event, occurrence, circumstance, change or effect resulting or arising from Purchaser's material breach of this Agreement that is not subsequently cured in accordance with Section 13; (vi) any event, occurrence, circumstance, change or effect arising directly or indirectly from or otherwise relating to any change in, or any compliance with or action taken for the purpose of complying with, any Law or U.S. GAAP; or (vii) any

matters disclosed in the Proxy Statement; it being understood that the exception in clause "(iv)" shall not prevent or otherwise affect a determination that the underlying cause of any such decline or failure referred to therein (if not otherwise falling within any of the exceptions provided by clauses "(i)" through "(iii)" or "(v)" through "(vii)" hereof) is or would be reasonably likely to be a Material Adverse Effect.

- "Material Contracts" has the meaning set forth in Section 4.14(a). 1.75
- 1.76 "Nasdaq" means the Nasdaq Capital Market.
- 1.77 "Nasdaq Listing Application" has the meaning set forth in Section 7.4.

"Non-Recourse Party" means, with respect to a party to this Agreement, any of such party's 1.78 former, current and future equity holders, controlling persons, directors, officers, employees, agents, representatives, Affiliates, members, managers, general or limited partners, or assignees (or any former, current or future equity holder, controlling person, director, officer, employee, agent, representative, Affiliate, member, manager, general or limited partner, or assignee of any of the foregoing); provided that, for the avoidance of doubt, no party to this Agreement will be considered a Non-Recourse Party.

- 1.79 "Notice of Objection" has the meaning set forth in Section 3.2(b).
- 1.80 "OFAC" has the meaning set forth in Section 4.17(b).

1.81 "Order" means any decree, order, judgment, writ, award, injunction, rule or consent of or by an Authority.

- 1.82 "Outside Closing Date" has the meaning set forth in Section 12.1(a).
- 1.83 "PBGC" has the meaning set forth in Section 4.22(b).
- "Permits" has the meaning set forth in Section 4.16. 1.84

1.85 "Permitted Liens" means (a) all defects, exceptions, restrictions, easements, rights of way and encumbrances affecting Real Property that are not, individually or in the aggregate material to the Business; (b) mechanics', carriers', workers', repairers' and similar statutory Liens arising or incurred in the ordinary course of business for amounts (i) that are not delinquent, (ii) that are not material to the business, operations and financial condition of the Company so encumbered, either individually or in the aggregate, and (iii) not resulting from a breach, default or violation by the Company Group of any Contract or Law; (c) liens for Taxes or assessments and similar governmental charges or levies, which either are (i) not delinquent or (ii) being contested in good faith and by appropriate proceedings, and adequate reserves have been established with respect thereto; (d) non-exclusive licenses of Intellectual Property Rights granted or received in the ordinary course of business consistent with past practice; (e) other liens arising in the ordinary course of business and not incurred in connection with the borrowing of money, the existence of which would not have a Material Adverse Effect; and (f) the Liens set forth on Schedule 1.86.

"Person" means an individual, corporation, partnership (including a general partnership, 1.86 limited partnership or limited liability partnership), limited liability company, association, trust or other entity or organization, including a government, domestic or foreign, or political subdivision thereof, or an agency or instrumentality thereof.

- 1.87 "Plan(s)" has the meaning set forth in Section 4.22(a).
- "Prior Company Counsel" has the meaning set forth in Section 13.19(a). 1.88
- 1.89 "Privileged Deal Communications" has the meaning set forth in Section 13.19(c).
- "Prospectus" has the meaning set forth in Section 13.14. 1.90
- 1.91 "Proxy Statement" has the meaning set forth in Section 9.1(a).
- 1.92 "Purchaser Board" has the meaning set forth in Section 5.11.

- 1.93 "Purchaser Board Recommendation" has the meaning set forth in Section 5.11.
- 1.94 "Purchaser Charter Amendment" has the meaning set forth in Section 9.1(a).
- 1.95 "Purchaser Charter Documents" has the meaning set forth in Section 2.7.

1.96 "Purchaser Common Stock" means the common stock, par value \$0.0001 per share, of

Purchaser.

1.97 "Purchaser Financials" has the meaning set forth in Section 5.12(b).

1.98 "<u>Purchaser Material Adverse Effect</u>" means (i) a material adverse change or a material adverse effect, individually or in the aggregate, upon on the assets, financial condition, business or operations of the Purchaser, taken as a whole, or (ii) any effect, change, event or occurrence that would individually or in the aggregate, prevent, materially delay or materially impair the ability of Purchaser to consummate the transactions contemplated by this Agreement.

1.99 "Purchaser Material Contract" has the meaning set forth in Section 5.19(a).

1.100 "<u>Purchaser Preferred Stock</u>" means the Series A Preferred Stock, par value \$0.0001 per share, of Purchaser.

1.101 "<u>Purchaser Private Warrant</u>" means each redeemable warrant issued in the private placement at the time of the consummation of the IPO, entitling the holder thereof to purchase one-half (1/2) of one share of Purchaser Common Stock at an exercise price of \$11.50 per share.

1.102 "<u>Purchaser Public Warrant</u>" means each redeemable warrant that was included as part of each Purchaser Unit in the IPO, entitling the holder thereof to purchase one-half (1/2) of one share of Purchaser Common Stock at an exercise price of \$11.50 per share.

- 1.103 "Purchaser Representatives" has the meaning set forth in Section 7.7.
- 1.104 "Purchaser SEC Documents" has the meaning set forth in Section 5.12(a).
- 1.105 "Purchaser Securities" has the meaning set forth in Section 5.10.
- 1.106 "Purchaser Stockholder Approval" has the meaning set forth in Section 9.2(a).
- 1.107 "Purchaser Stockholder Matters" has the meaning set forth in Section 9.1(a).
- 1.108 "Purchaser Stockholder Meeting" has the meaning set forth in Section 9.2(a).

1.109 "Purchaser Unit" means a unit of the Purchaser comprised of (a) one share of Purchaser Common Stock, and (b) one redeemable warrant to purchase one-half (1/2) of one share of Purchaser Common Stock at an exercise price of \$11.50 per share.

1.110 "Purchaser Warrant" means each Purchaser Private Warrant and Purchaser Public Warrant.

1.111 "<u>Real Property</u>" means, collectively, all real properties and interests therein (including the right to use), together with all buildings, fixtures, trade fixtures, plant and other improvements located thereon or attached thereto; all rights arising out of use thereof (including air, water, oil and mineral rights); and all subleases, franchises, licenses, permits, easements and rights-of-way which are appurtenant thereto.

1.112 "<u>Redemptions</u>" has the meaning set forth in <u>Section 1.17</u>.

1.113 "<u>Reference Time</u>" means, with respect to the Company, the close of business on the Closing Date (but without giving effect to the transactions contemplated by this Agreement, including any payments by the Purchaser hereunder to occur at the Closing, but treating any obligations in respect of Indebtedness or other liabilities that are contingent upon the consummation of the Closing as currently due and owing without contingency as of the Reference Time).

1.114 "Reg. D" has the meaning set forth in Section 6.5(a).

1.115 "<u>Registration Rights Agreement</u>" means the agreement executed between the Purchaser and the Stockholders executed as of the date hereof (to be effective upon the Closing) in the form attached hereto as <u>Exhibit B</u> governing the resale of the Closing Payment Shares, the Purchaser Common Stock held by Sponsor, and the Purchaser Public Warrants.

1.116 "Rejection Recommendation" has the meaning set forth in Section 1.137.

1.117 "<u>Representatives</u>" means, collectively, with respect to any Person, such Person's officers, directors, Affiliates, employees, agents or advisors, including any investment banker, broker, attorney, accountant, consultant or other authorized representative of such Person.

1.118 "<u>R&W Insurance Policy</u>" means the Representations and Warranties Insurance Policy, bound by AXA XL Insurance as of the date hereof and effective as of the Closing Date, in the form attached hereto as <u>Exhibit E</u>, with Policy Number US00094394BL19A.

- 1.119 "Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002, as amended.
- 1.120 "SEC" means the U.S. Securities and Exchange Commission.
- 1.121 "Section 16" has the meaning set forth in Section 7.5.
- 1.122 "Securities Act" means the Securities Act of 1933, as amended.
- 1.123 "Share Exchange" has the meaning set forth in the Recitals.
- 1.124 "Sponsor" means Health Sciences Holdings, LLC, a Delaware limited liability company.

1.125 "<u>Sponsor Restricted Stock Agreement</u>" means the Restricted Stock Agreement between the Purchaser and the Sponsor executed as of the date hereof (to be effective upon the Closing) in the form attached hereto as <u>Exhibit C</u>.

1.126 "<u>Standard Contracts</u>" means (a) "shrink wrap" or other licenses for generally commercially available software (including open source software) or hosted services, (b) Contracts with Company's own employees or contractors substantially on Company's standard forms, and (c) standard non-disclosure agreements, material transfer agreements and master services agreements.

1.127 "Stockholders' Representative" has the meaning set forth in Section 13.15.

1.128 "<u>Subsidiary</u>" means each entity of which at least fifty percent (50%) of the capital stock or other equity or voting securities are Controlled or owned, directly or indirectly, by the Company.

1.129 "<u>Tax Return</u>" means any return, information return, declaration, claim for refund or credit, report or any similar statement, and any amendment thereto, including any attached schedule and supporting information, whether on a separate, consolidated, combined, unitary or other basis, that is filed or required to be filed with any Taxing Authority in connection with the determination, assessment, collection or payment of a Tax or the administration of any Law relating to any Tax.

1.130 "<u>Tax(es)</u>" means any federal, state, local or foreign tax, charge, fee, levy, custom, duty, deficiency, or other assessment in the nature of a tax imposed by any Taxing Authority (including any income (net or gross), gross receipts, profits, windfall profit, sales, use, goods and services, ad valorem, franchise, license, withholding, employment, social security, workers compensation, unemployment compensation, employment, payroll, transfer, excise, import, real property, personal property, intangible property, occupancy, recording, minimum, alternative minimum, environmental or estimated tax), together with any interest, penalty, additions to tax or additional amount imposed with respect thereto.

1.131 "<u>Taxing Authority</u>" means the IRS and any other Authority responsible for the collection, assessment or imposition of any Tax or the administration of any Law relating to any Tax.

1.132 "<u>Trading Day</u>" means any day on which the principal securities exchange or securities market on which the Purchaser Common Stock are then traded is open for trading.

1.133 "<u>Transfer Taxes</u>" means all transfer, documentary, sales, use, stamp, registration, value added (to the extent irrecoverable), excise, stock transfer, filing, real property transfer, recording, and other similar Taxes and fees (including any penalties and interest) applicable to or incurred in connection with the transactions contemplated by this Agreement.

1.134 "<u>Transaction Expense</u>" of a Person means, without duplication, the aggregate expenses, fees and disbursements of all attorneys, accountants and investment bankers for which a Person is liable in connection with the negotiation, execution, delivery and performance of this Agreement through the Closing that remain unpaid or outstanding as of the Closing.

1.135 "<u>Transaction Securities</u>" means the Closing Payment Shares, the Earnout Shares and the Purchaser Common Stock issuable upon the exercise of the Converted Options.

"Triggering Event" shall be deemed to have occurred if: (A) a Change of Recommendation 1.136 shall have been effected or occurred for any reason, (B) the Purchaser shall have failed to convene or hold the Purchaser Stockholder Meeting in accordance with Section 9.2, (C) the Purchaser or the Company shall have breached any of the provisions of Section 7.7 or Section 8.2 in any material respect, (D) the Purchaser Board fails to reaffirm the Purchaser Board Recommendation within 10 Business Days after the Company requests in writing that such recommendation be reaffirmed in response to an Acquisition Proposal or material modification to an Acquisition Proposal that has been publicly announced or otherwise becomes publicly known (or if such request is delivered less than 10 Business Days prior to the Purchaser Stockholder Meeting, no later than one Business Day prior to the Purchaser Stockholder Meeting; provided that if such Acquisition Proposal is subsequently modified within such period, then the Purchaser Board shall be required to reaffirm such recommendation no later than one Business Day prior to the Purchaser Stockholder Meeting) or (E) a tender or exchange offer relating to securities of the Purchaser shall have been commenced by a Person unaffiliated with the Company and the Purchaser fails to send to its stockholders pursuant to Rule 14e-2 promulgated under the Securities Act, within 10 Business Days after such tender or exchange offer is first published, sent or given, a statement disclosing that the Purchaser unconditionally recommends rejection of such tender or exchange offer (the "Rejection Recommendation") and reaffirms the Purchaser Board Recommendation or fails to reaffirm the Rejection Recommendation in any press release published by the Purchaser (or by any of its Affiliates or Purchaser Representatives) or in any Schedule 14D-9 filed by the Purchaser with the SEC, in each case relating to such tender offer or exchange offer, at any time after the foregoing 10 Business Day period.

- 1.137 "Trust Account" has the meaning set forth in Section 5.9.
- 1.138 "Trust Agreement" has the meaning set forth in Section 5.9.
- 1.139 "Trust Fund" has the meaning set forth in Section 5.9.
- 1.140 "Trustee" has the meaning set forth in Section 5.9.
- 1.141 "U.S. GAAP" means U.S. generally accepted accounting principles, consistently applied.
- 1.142 "Unaudited Financial Statements" has the meaning set forth in Section 4.10(a).

ARTICLE II SHARE EXCHANGE

2.1 <u>Share Exchange</u>. Upon the terms and subject to the conditions set forth in this Agreement, on the Closing Date, the Purchaser shall issue to the Stockholders the Closing Payment Shares, which shall be fully paid and free and clear of all Liens other than applicable securities Law restrictions and the Lock-Up Agreements, and each Stockholder shall receive the number of Closing Payment Shares opposite such Stockholder's name on <u>Schedule 1.23</u>, which may be adjusted on a pro rata basis pursuant to the final Closing Payment Shares as determined pursuant to <u>Section 3.2</u>, in exchange for the number of Company Common Shares opposite such Stockholder's name on <u>Schedule 1.23</u>. Upon the terms and subject to the conditions set forth in this Agreement, on the Closing Date, the Purchaser shall issue to Roivant Sciences Ltd. 10,000 shares of Purchaser Preferred Stock.

2.2 Stock Option Conversion.

(a) On the Closing Date, each option to purchase Company Common Shares (each, a "<u>Company Option</u>") that is outstanding under any Equity Incentive Plan immediately prior to the Closing, and each Equity Award, whether vested or unvested, shall, automatically and without any required action on the part of any holder or beneficiary thereof, be assumed by the Purchaser and converted into an option to purchase shares of Purchaser Common Stock (each, a "<u>Converted Option</u>"). Each Converted Option shall continue to have and be subject to substantially the same terms and conditions as were applicable to such Company Option immediately before the Closing (including expiration date, vesting conditions, and exercise provisions), except that each Converted Option shall be exercisable for that number of shares of Purchaser Common Stock and have a per share exercise price for each share of Purchaser Common Stock issuable upon exercise of the Converted Option as set forth in <u>Schedule 1.23</u> opposite the name of each holder of a Company Option, which may be adjusted on a pro rata basis pursuant to the final Closing Payment Shares as determined pursuant to <u>Section 3.2</u>.

(b) At the Closing, Purchaser shall assume all of the obligations of the Company under any Equity Incentive Plan in respect of Company Options, and shall assume such outstanding awards and the obligations under the agreements evidencing such awards. Purchaser shall take all corporate action necessary to reserve for issuance a number of authorized but unissued shares of Purchaser Common Stock for delivery upon settlement of the assumed Company Options in accordance with <u>Section 2.2(a)</u>. Promptly after the Closing (but in no event more than thirty (30) calendar days thereafter), Purchaser shall file or otherwise have available a registration statement on Form S-8 (or other appropriate form) with respect to the shares of Purchaser Common Stock subject to the assumed Company Options.

2.3 <u>Closing</u>. Unless this Agreement is earlier terminated in accordance with Article XII, the closing of the Share Exchange (the "<u>Closing</u>") shall take place at the offices of Loeb & Loeb LLP, 345 Park Avenue, New York, New York, at 10:00 a.m. local time, on the second (2^{nd}) Business Day after the satisfaction or waiver (to the extent permitted by applicable Law) of the conditions set forth in <u>Article X</u> or at such other time, date and location as the Purchaser and the Company agree in writing. The parties may participate in the Closing via electronic means. The date on which the Closing actually occurs is hereinafter referred to as the "<u>Closing Date</u>".

2.4 <u>Board of Directors of Purchaser</u>. Immediately after the Closing, the Purchaser Board will consist of seven (7) directors, with six (6) directors identified by the Stockholders' Representative and one (1) director identified by the Purchaser.

2.5 <u>Taking of Necessary Action; Further Action</u> If, at any time after the Closing, any further action is necessary or desirable to carry out the purposes of this Agreement and to vest the Purchaser with full right, title and interest in, to and under, and/or possession of, all assets, property, rights, privileges, powers and franchises of the Company, the officers and directors of the Purchaser are fully authorized in the name and on behalf of the Company, to take all lawful action necessary or desirable to accomplish such purpose or acts, so long as such action is not inconsistent with this Agreement.

2.6 Section 368 Reorganization. For U.S. federal income tax purposes, the Share Exchange is intended to constitute a "reorganization" within the meaning of Section 368(a) of the Code and the Treasury Regulations promulgated thereunder to which each of the Company and the Purchaser are to be parties under Section 368(b) of the Code. The parties to this Agreement hereby (a) adopt this Agreement insofar as it relates to the Share Exchange as a "plan of reorganization" within the meaning of Section 1.368-2(g) of the United States Treasury regulations, and (c) intend to file all Tax and other informational returns on a basis consistent with such characterization.

2.7 <u>Purchaser Charter Documents</u>. Immediately upon the Closing, the certificate of incorporation and bylaws of the Purchaser ("<u>Purchaser Charter Documents</u>") shall be amended and restated in the forms attached hereto as <u>Exhibit D</u>.

ARTICLE III CONSIDERATION

3.1 Closing Payment Shares.

(a) No certificates or scrip representing fractional shares of Purchaser Common Stock will be issued pursuant to the Share Exchange, and such fractional share interests will be rounded down to the nearest whole share and will not entitle the owner thereof to vote or to any rights of a stockholder of the Purchaser. The Closing Payment Shares shall bear the legend set forth in <u>Section 3.1(b)</u> and shall be subject to terms of the Lock-Up Agreement.

(b) Legend. Each certificate issued to any holder of Company Common Shares in connection with the Share Exchange shall bear the legend set forth below, or legend substantially equivalent thereto, together with any other legends that may be required by any securities laws at the time of the issuance of the Purchaser Common Stock:

THE SHARES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "<u>ACT</u>") OR THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION, AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL (I) SUCH OFFER, SALE, TRANSFER, PLEDGE OR HYPOTHECATION HAS BEEN REGISTERED UNDER THE ACT AND THE SECURITIES LAWS OF ANY STATE OR OTHER JURISDICTION COVERING SUCH SECURITIES OR (II) THE ISSUER OF THE SHARES HAS RECEIVED AN OPINION OF COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE WITH THE ACT AND SUCH OTHER APPLICABLE LAWS.

3.2 Pre-Closing Adjustment for Closing Indebtedness.

(a) For the purposes of this Agreement, the "<u>Determination Date</u>" shall be the date that is five (5) calendar days prior to the anticipated date for Closing, as agreed upon by the Company and Purchaser at least ten (10) calendar days prior to the Purchaser Stockholder Meeting. On or prior to the Determination Date, the Company shall prepare and deliver to the Purchaser a closing statement (when finalized pursuant to this <u>Section 3.2</u>, the "<u>Closing Statement</u>"), setting forth: (i) the Closing Payment Shares and Purchaser Common Stock issuable under the Converted Options pursuant to <u>Section 2.2</u>; and (ii) the Closing Indebtedness"). The Closing Statement shall be prepared in accordance with U.S. GAAP and prepared consistently with the most recent audited financial statements of the Company made available to the Purchaser prior to the date of this Agreement, and otherwise in accordance with this Agreement.

(b) The Purchaser shall review the Closing Statement promptly after the delivery of the same by the Company. The Company shall, upon reasonable request, provide the Purchaser with applicable work papers used in the preparation of the Closing Statement. If the Purchaser does not agree with the content of the Closing Statement, the Purchaser shall raise objection by delivering a written statement of objection (the "<u>Notice of Objection</u>") to the Company within two (2) calendar days after the receipt of the Closing Statement from the Company, specifying in reasonable detail the item(s) which are disputed, the basis of such disputes and the changes the Purchaser considers necessary. The Closing Statement shall become binding and conclusive on the parties for the purpose of the determination of the number of Closing Payment Shares and Purchaser Common Stock issuable under the Converted Options (and the Closing Indebtedness) if the Purchaser does not deliver a Notice of Objection to the Company pursuant to and in accordance with this paragraph.

(c) If the Purchaser delivers a Notice of Objection in accordance with paragraph (b) above, then the parties shall first try to resolve the objected items through mutual consultation. If any objected items cannot be resolved through mutual consultation within three (3) calendar days from the date of the Notice of Objection, such dispute shall be submitted to WithumSmith+Brown, PC or such other firm mutually acceptable to the Purchaser and the Company (the "<u>Independent Accountant</u>") for final determination; provided that, if the Independent Accountant does not accept its appointment and/or if the Purchaser and the Company cannot agree on the Independent Accountant, in either case within five (5) calendar days after the date on which the dispute is submitted

to the Independent Accountant in accordance with this <u>Section 3.2(c)</u>, then either the Purchaser or the Company may require, by written notice to the other, that the Independent Accountant be selected by the New York City Regional Office of the American Arbitration Association (the "<u>AAA</u>") in accordance with the AAA's procedures. The Independent Accountant shall be an independent (i.e., no prior material business relationship with any party for the prior two (2) years) accounting firm. The parties agree that the Independent Accountant will be deemed to be independent even though a party or its Affiliates may, in the future, designate the Independent Accountant to resolve disputes of the types described in this <u>Section 3.2</u>.

(d) If any items in dispute are submitted to the Independent Accountant for final determination, (i) each of the Purchaser and the Company shall furnish to the Independent Accountant such work papers and other documents and information relating to the disputed items as the Independent Accountant may request and are available to that party, and shall be afforded the opportunity to present to the Independent Accountant any material relating to the determination and to discuss the determination with the Independent Accountant; (ii) the Independent Accountant shall be directed to, within ten (10) calendar days after submission of the issues, deliver a notice to each of the Purchaser and the Company, setting forth its adjustment or revision of the Closing Statement, and/or the resolution of all issues in dispute; (iii) the Closing Statement, as adjusted or otherwise revised and finally determined by the Independent Accountant, as set forth in its notice delivered to each of the Company and the Purchaser, shall be binding and conclusive on the parties and the parties shall delay the Closing until the resolution of the matters described in this Section 3.2(d); and (iv) the costs of the Independent Accountant shall be paid by the Company (on one hand) and the Purchaser (on the other hand) in the same proportion that the aggregate disputed amount so submitted to the Independent Accountant that is unsuccessfully disputed by each such party as finally determined by the Independent Accountant bears to the total disputed amount.

(e) Upon finalization of the Closing Statement, the parties agree that the Purchaser shall issue the Closing Payment Shares and assume the Converted Options pursuant to <u>Article II</u>.

(f) Each of the Company and the Purchaser shall bear the fees, costs and expenses of its own accountants.

3.3 Earnout Payment.

(a) In the event the volume weighted average sale price of the Purchaser Common Stock equals or exceeds the stock prices set forth below for any 20 Trading Days within a 30-day Trading Day period from and after the Closing until the applicable milestone date set forth below, the Stockholders shall be entitled to receive, and the Purchaser shall promptly (but no later than five (5) Business Days), cause to be issued, as additional consideration for the Share Exchange (and without the need for additional consideration from the Stockholders), additional shares of Purchaser Common Stock (the "Earnout Shares") on a pro-rata basis based on their ownership percentages in the Company as set forth on <u>Schedule 1.23</u>, which shares shall be fully paid and free and clear of all Liens other than applicable securities Law restrictions:

	Milestone Date	Stock Price		Additional Shares of Common Stock
Milestone #1	March 31, 2023	\$	17.50	10,000,000
Milestone #2	March 31, 2025	\$	31.50	10,000,000

All share and per share amounts shall be proportionally adjusted for stock splits, stock dividends, and similar events.

(b) At all times prior to the latest milestone date set forth in<u>Section 3.3(a)</u> (the "<u>Earnout</u> <u>Period</u>"), the Purchaser shall keep available for issuance a sufficient number of unissued shares of Purchaser Common Stock to permit the Purchaser to satisfy its issuance obligations set forth in this <u>Section 3.3</u> and shall take all actions required to increase the authorized number of shares of Purchaser Common Stock if at any time there shall be insufficient unissued shares of Purchaser Common Stock to permit such reservation.

(c) In the event that after the Closing and during the Earnout Period, (i) there is a Change of Control, (ii) any liquidation, dissolution or winding up of the Purchaser (whether voluntary or involuntary) is initiated, (iii) any bankruptcy, reorganization, debt arrangement or similar proceeding under any bankruptcy, insolvency or similar law, or any dissolution or liquidation proceeding, is instituted by or against the Purchaser, or a

receiver is appointed for the Purchaser or a substantial part of its assets or properties, or (iv) the Purchaser makes an assignment for the benefit of creditors, or petitions or applies to any Authority for, or consents or acquiesces to, the appointment of a custodian, receiver or trustee for all or substantially all of its assets or properties (each of clauses (i) through (iv), an "Acceleration Event"), then any Earnout Shares that have not been previously issued by the Purchaser (whether or not previously earned), shall be deemed earned and due by the Purchaser to the Stockholders upon such Acceleration Event unless, in the case of an Acceleration Event that is a Change of Control, the Earnout Shares for Milestone #1 or Milestone #2 shall not be accelerated if the value of the consideration to be received in exchange for a share of Purchaser Common Stock in such Change of Control is lower than the applicable stock price threshold set forth in Section 3.3(a) for Milestone #1 and Milestone #2, respectively. For purposes hereof, a "Change of Control" means the occurrence in a single transaction or as a result of a series of related transactions, of one or more of the following events: (i) a merger, consolidation, reorganization or similar business combination transaction involving the Purchaser in which the holders of all of the outstanding equity interests of the Purchaser immediately prior to the consummation of such transaction do not directly own, beneficially or of record, immediately upon the consummation of such transaction, outstanding equity interests that represent a majority of the combined outstanding voting securities of the surviving entity in such transaction or a parent of the surviving entity in such transaction; (ii) a transaction in which a majority of the Purchaser's voting securities are transferred to any Person, or any two more Persons acting as a group, and all Affiliates of such Person or Persons (each, a "Group") or (iii) the consummation of the sale of substantially all of the assets of the Purchaser to any Person or Group.

(d) All distributions of Purchaser Common Stock with respect to the Earnout Shares during the Earnout Period, including, but not limited to, shares of Purchaser Common Stock issued as a result of a stock dividend, stock split, combination of shares or otherwise, shall be deemed to be Earnout Shares and shall be set aside and not issued until the Earnout Shares have been issued to the Stockholders or, if the Earnout Shares are not earned and issued, then all such distributions declared during the Earnout Period shall be forfeited.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except (a) as set forth in the Disclosure Schedule or (b) set forth in the Proxy Statement, the Company hereby represents and warrants to the Purchaser that each of the following representations and warranties are true, correct and complete as of the date of this Agreement and as of the Closing Date (except to the extent expressly made as of an earlier date, in which case such representations and warranties shall be so true, correct and complete as of such earlier date).

4.1 <u>Corporate Existence and Power</u>. The Company is an exempted limited company duly incorporated, validly existing and in good standing under the Laws of Bermuda. The Company has all power and authority, corporate and otherwise, and all governmental licenses, franchises, Permits, authorizations, consents and approvals required to own and operate its properties and assets and to carry on the Business as presently conducted and as proposed to be conducted. The Company is duly licensed or qualified to do business and is in good standing in each jurisdiction in which the properties owned or leased by it or the operation of its Business as currently conducted makes such licensing or qualification necessary, except where the failure to be so licensed, qualified or in good standing would not have a Material Adverse Effect. Other than this Agreement and the transactions contemplated thereby, there is no Contract to which the Company is a party in respect of any merger, consolidation, sale of all or substantially all of its assets, reorganization, recapitalization, dissolution or liquidation.

4.2 <u>Authorization</u>. The execution, delivery and performance by the Company of this Agreement and the Additional Agreements and the consummation by the Company of the transactions contemplated hereby and thereby are within the corporate powers of the Company and have been duly authorized by all necessary action on the part of the Company. Assuming due authorization, execution and delivery by each other party hereto and to the Additional Agreements, this Agreement constitutes, and, upon their execution and delivery, each of the Additional Agreements will constitute, a valid and legally binding agreement of the Company enforceable against the Company in accordance with their respective terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity.

4.3 <u>Governmental Authorization</u>. Except for filings under the HSR Act or any other applicable Laws relating to antitrust and except for the requirement for permission of the Bermuda Monetary Authority under the Exchange Control Act 1972 of Bermuda and related regulations, neither the execution, delivery nor performance

by the Company of this Agreement or any Additional Agreements requires any consent, approval, license, order or other action by or in respect of, or registration, declaration or filing with, any Authority as a result of the execution, delivery and performance of this Agreement or any of the Additional Agreements or the consummation of the transactions contemplated hereby or thereby (each of the foregoing, a "<u>Governmental</u> <u>Approval</u>").

4.4 <u>Non-Contravention</u>. None of the execution, delivery or performance by the Company of this Agreement or any Additional Agreements does or will (a) contravene or conflict with the organizational or constitutive documents of any member of the Company Group, (b) contravene or conflict with or constitute a violation of any provision of any Law or Order binding upon or applicable to the Company Group, (c) constitute a default under or breach of (with or without the giving of notice or the passage of time or both) or violate or give rise to any right of termination, cancellation, amendment or acceleration of any right or obligation of the Company Group or require any payment or reimbursement or to a loss of any material benefit relating to the Business to which the Company Group are entitled under any provision of any Material Contract, (d) result in the creation or imposition of any Lien on any Other Company Group's assets, except, in the case of (b), (c), (d) and (e), as would not, individually or in the aggregate, have a Material Adverse Effect.

4.5 <u>Capitalization</u>. The Company is authorized to issue 1,000,000,000 common shares, par value \$ \$0.00001 per share (the "Company Common Shares") of which 78,906,250 Company Common Shares are issued and outstanding. The Company has 6,789,981 Equity Awards issued and outstanding, with another 2,960,019 Company Common Shares authorized for issuance under the Equity Incentive Plan. Set forth on Schedule 4.5 is the list of holders of Equity Awards and the number of such Equity Awards each such holder owns. No Company Common Shares are held in its treasury. All of the issued and outstanding Company Common Shares have been duly authorized and validly issued, is fully paid and non-assessable and has not been issued in violation of any preemptive or similar rights of any Person. All of the issued and outstanding Company Common Shares are owned of record and beneficially by the Stockholders as set forth on Schedule 1.23, free and clear of all Liens. No outstanding Company Common Shares are subject to any right of first refusal, right of first offer, preemptive right or similar restriction other than as set forth in Schedule 4.5 and provided in by-laws 82, 83, 84 and 86 of the Amended and Restated By-laws of the Company. The only Company Common Shares that will be outstanding immediately after the Closing will be the Company Common Shares owned by the Purchaser following the consummation of the Share Exchange. No other class of capital stock of the Company is authorized or outstanding. Other than as set forth in this Section 4.5, there are no: (a) outstanding subscriptions, options, warrants, rights (including "phantom stock rights"), calls, commitments, understandings, conversion rights, rights of exchange, plans or other agreements of any kind providing for the purchase, issuance or sale of any shares of the Company, or (b) agreements to which the Company is a party, or, to the knowledge of the Company, other agreements with respect to the Company Common Shares, including any voting trust, other voting agreement or proxy with respect thereto.

4.6 <u>Charter Documents</u>. Copies of the Certificate of Incorporation and the Amended and Restated By-laws of the Company (the "<u>Charter Documents</u>") have heretofore been made available to the Purchaser, and such copies are each true and complete copies of such instruments as amended and in effect on the date hereof.

4.7 <u>Corporate Records</u>. All proceedings occurring since July 6, 2018 of the board of directors of the Company, including committees thereof, and all consents to actions taken thereby, are accurately reflected in the minutes and records contained in the corporate minute books of the Company in all material respects, which are complete and have been maintained in accordance with sound business practices in Bermuda.

4.8 <u>Third Parties</u>. Except as set forth on<u>Schedule 4.8</u>, no Key Company Employees engage in any business, except through the Company, or are employees of or provide any service for compensation to, any other business concern. Other than offer letters for employment or Equity Award grant agreements or exercise notices, <u>Schedule 4.8</u> lists each Contract to which the Company, on the one hand, and any Stockholder, on the other hand, is a party, including any such Contract pursuant to which a Stockholder or any Affiliate of a Stockholder owns, directly or indirectly, in whole or in part, any tangible or intangible property (including Intellectual Property Rights) that the Company uses or the use of which is necessary for the conduct of the Business or the ownership or operation of the Company's assets. <u>Schedule 4.8</u> sets forth a complete and accurate list of the Affiliates of the Company and the ownership interests in the Affiliate of the Company and each Stockholder.

4.9 <u>Subsidiaries</u>.

(a) <u>Schedule 4.9</u> of the Disclosure Schedule sets forth each Subsidiary of the Company on the date hereof, and, except as set forth in <u>Schedule 4.9</u>, the Company does not currently own, directly or indirectly, securities or other ownership interests in any other entity. The Company owns 100% of the issued and outstanding capital stock and securities of each Person listed on <u>Schedule 4.9</u>.

(b) Each Subsidiary is a corporation duly organized, validly existing and in good standing under and by virtue of the Laws of the jurisdiction of its formation set forth by its name on <u>Schedule 4.9</u>. Each Subsidiary has all power and authority, corporate and otherwise, required to own and operate its properties and assets and to carry on the Business as presently conducted. Other than as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, each Subsidiary is duly qualified as a foreign corporation, partnership or limited liability company, as applicable, to transact the Business in each jurisdiction in which the character of the property owned or leased by any Subsidiary or the nature of its activities make qualification of such Subsidiary in any such jurisdiction necessary. No Subsidiary has entered into any agreement in respect of any merger, consolidation, sale of all or substantially all of its assets, reorganization, recapitalization, dissolution or liquidation.

4.10 Financial Statements.

(a) True and correct copies of (i) the audited consolidated financial statements of the Company as of March 31, 2019 and 2018 and for the fiscal year ended March31, 2019 and the period from December 19, 2017 to March 31, 2018 (the "<u>Audited Financial Statements</u>") consisting of the audited consolidated balance sheets as of such dates, and the audited consolidated statements of operations, comprehensive loss, equity and cash flows for such periods, and (ii) unaudited financial statements (the "<u>Unaudited Financial Statements</u>") from April 1, 2019 through June 30, 2019 (collectively, the "<u>Financial Statements</u>") included therein, the "<u>Balance Sheet</u>"), have been provided to the Purchaser.

(b) The Financial Statements (i) were prepared in accordance with U.S. GAAP consistently applied, (ii) comply with all applicable accounting requirements under the Securities Act and the rules and regulations of the SEC thereunder, and (iii) fairly present in all material respects, in conformity with U.S. GAAP applied on a consistent basis, the financial position of the Company Group as of the dates thereof and the results of operations of the Company Group for the periods reflected therein (subject to normal and recurring year-end adjustments that are not, individually or in the aggregate, material).

(c) As of the date of this Agreement, the Company Group does not have any liabilities of the type required to be disclosed in the liabilities column of a consolidated balance sheet prepared in accordance with GAAP, except for: (i) liabilities disclosed on the Balance Sheet; (ii) liabilities or obligations incurred pursuant to the terms of this Agreement; (iii) liabilities for performance of obligations of the Company Group under Contracts binding upon the applicable member of the Company Group (other than resulting from any breach or acceleration thereof) either delivered or made available to Purchaser or Purchaser's Representatives prior to the date of this Agreement or entered into in the ordinary course of business; (iv) liabilities incurred in the ordinary course of business since the date of the Balance Sheet; and (v) liabilities that individually or in the aggregate have not had and would not reasonably be expected to have a Material Adverse Effect.

(d) The Company and each of its Subsidiaries make and keep accurate Books and Records and maintain a system of internal accounting controls designed, and which the Company believes is sufficient, to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with U.S. GAAP and to maintain accountability for assets; (iii) access to assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

(e) The Company has established and maintains disclosure controls and procedures (as defined in Rules 13a-15 and 15d-15 under the Exchange Act), which (i) are designed to ensure that material information relating to the Company, including its consolidated subsidiaries, is made known to the Company's principal executive officer and its principal financial officer by others within those entities and (ii) are effective in

all material respects to perform the functions for which they were established. Since the end of the Company's most recent audited fiscal year, there have been no significant deficiencies or material weakness in the Company's internal control over financial reporting (whether or not remediated) and no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

4.11 <u>Absence of Certain Changes</u>. Since the Balance Sheet Date through the date of this Agreement, the Company Group has conducted the Business in the ordinary course consistent with past practices. Without limiting the generality of the foregoing, since the Balance Sheet Date through the date of this Agreement, there has not been:

(a) any Material Adverse Effect;

(b) (i) any redemption of, declaration, setting aside or payment of any dividend or other distribution with respect to any capital stock or other equity interests in the Company Group; (ii) any issuance by the Company Group of shares of capital stock or other equity interests in the Company Group, or (iii) any repurchase, redemption or other acquisition, or any amendment of any term, by the Company Group of any outstanding shares of capital stock or other equity interests;

(c) (i) any creation or other incurrence of any Lien (other than Permitted Liens) on the Company Common Shares or any other capital stock or securities of the Company Group or on any of the Company Group's assets, and (ii) any making of any loan, advance or capital contributions to or investment in any Person by the Company Group other than in the ordinary course of business consistent with past practice;

(d) any personal property damage, destruction, or casualty loss or personal injury loss (whether or not covered by insurance) affecting the Business or assets of the Company Group, except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect;

(e) (i) any material amendment to or termination of any Material Contract, (ii) any material amendment to any material Permit from any Authority held by the Company, and (iii) any receipt of notice of termination of any of the items referenced in (i) and (ii);

(f) any material Tax election changed or revoked by the Company Group; any material claim, audit or assessment in respect of Taxes settled or compromised by the Company Group; or any Tax allocation agreement, Tax sharing agreement, Tax indemnity agreement or closing agreement relating to any Tax (other than an ordinary commercial agreement the principal purpose of which does not relate to Taxes) entered into by the Company Group;

(g) any amendment to the Company Group's organizational documents, or any engagement by the Company Group in any merger, consolidation, reorganization, reclassification, liquidation, dissolution or similar transaction; or

(h) any commitment or agreement to do any of the foregoing.

4.12 <u>Properties: Title to the Company's Assets</u>. Except as would not be individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company has good, valid and marketable title in and to, or in the case of the Leases and the assets which are leased or licensed pursuant to Contracts, a valid leasehold interest or license in or a right to use, all of their assets reflected on the Balance Sheet. No such asset is subject to any Liens other than Permitted Liens. Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company Group's assets constitute all of the assets of any kind or description whatsoever, including goodwill, for the Company Group to operate the Business immediately after the Closing in the same manner as the Business is currently being conducted. This <u>Section 4.12</u> shall not cover Intellectual Property Rights, which is addressed in <u>Section 4.18</u>.

4.13 <u>Litigation</u>. As of the date of this Agreement, there is no Action (or any basis therefore) pending against, or to the knowledge of the Company threatened in writing against or affecting, the Company Group, any of its officers or directors with respect to the Business, the Business, or any Company Common Shares or any of the Company's Group assets, or any Contract before any court, Authority or official, or which in any manner challenges or seeks to prevent or enjoin the transactions contemplated hereby or by the Additional Agreements. As of the date of this Agreement, there are no outstanding judgments against the Company Group.

4.14 Contracts.

(a) <u>Schedule 4.14(a)</u> lists all Contracts (other than (1) any Contract that is terminable without penalty by any other party thereto on 90 days' or less notice; *provided* that penalty shall not include requirements to pay costs and expenses in connection with the termination of such agreements consisting of reimbursement of expenses incurred and reasonable wind-down costs, (2) Standard Contracts, (3) ordinary course sales or purchase orders, or (4) Plans, which shall be governed by <u>Section 4.22</u>) to which the Company Group is a party and which are in effect as of the date of this Agreement and constitute the following (collectively, "<u>Material Contracts</u>"):

(i) all Contracts that require annual cash payments to or from the Company Group, taken as a whole, of \$200,000 or more in the fiscal year ending March 31, 2020;

(ii) all Contracts creating a joint venture, strategic alliance, limited liability company and partnership agreements to which a member of the Company Group is a party and that is material to the business of the Company Group taken as a whole;

(iii) all Contracts for material licensing agreements pursuant to which any member of the Company Group grants or receives rights in or to use any Intellectual Property Rights;

(iv) all Contracts materially limiting the freedom of the Company Group to compete in any line of business or with any Person or in any geographic area;

(v) all Contracts relating to patents, trademarks, service marks, trade names, brands, copyrights, trade secrets and other Intellectual Property Rights of the Company Group;

(vi) all Contracts providing for (a) guarantees or (b) as their primary purpose, indemnification arrangements and other hold harmless arrangements made or provided by the Company Group, including all ongoing agreements for the primary purpose of repair, warranty, maintenance, service, indemnification or similar obligations, other than any such Contracts entered into in the ordinary course of business;

(vii) all Contracts with or pertaining to the Company Group to which any Affiliate of the Company Group is a party, other than any Contracts pertaining to such Affiliate's status as a Company securityholder;

(viii) all Contracts relating to property or assets (whether real or personal, tangible or intangible) in which the Company Group holds a leasehold interest (including the Leases) and which involve payments to the lessor thereunder in excess of \$500,000 per year;

(ix) all Contracts relating to outstanding Indebtedness for borrowed money, including financial instruments of indenture or security instruments (typically interest-bearing) such as notes, mortgages, loans and lines of credit;

(x) any Contract relating to the voting or control of the equity interests of the Company Group or the election of directors of the Company Group (other than the organizational documents of the Company Group); and

(xi) any Contract providing for hedging obligations, including interest rate or currency exchange swaps, collars, caps or similar hedging obligations, of the Company.

(b) Each Material Contract is a valid and binding agreement, and is in full force and effect, and neither the Company Group nor, to the knowledge of the Company, any other party thereto, is in breach or default (whether with or without the passage of time or the giving of notice or both) under the material terms of any such Material Contract. The Company Group has not assigned, delegated, or otherwise transferred any of its rights or obligations with respect to any Material Contracts, or granted any power of attorney with respect thereto or to any of the Company Group's assets.

(c) The Company has, prior to the date hereof, provided to the Purchaser true and correct fully executed copies of each written Material Contract.

(d) The Company Group is in compliance with all material covenants, including all financial covenants, in all notes, indentures, bonds and other instruments or agreements evidencing any Indebtedness.

4.15 <u>Insurance</u>. Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company Group is insured by recognized, financially sound and reputable institutions with policies in such amounts and with such deductibles and covering such risks as are customarily carried by Persons conducting a business similar to the Company Group, including, but not limited to, policies covering real and personal property owned or leased by the Company Group against theft, damage, destruction and acts of vandalism and policies covering the Company Group for product liability claims and elinical trial liability claims. The Company has no reason to believe that it or any of its Subsidiaries will not be able (i) to renew its existing insurance coverage as and when such policies expire or (ii) to obtain comparable coverage from similar institutions as may be necessary or appropriate to conduct its business as now conducted and at a cost that would not reasonably be expected to have a Material Adverse Effect. Neither the Company nor its Subsidiaries have been denied any insurance coverage which it has sought or for which it has applied.

4.16 <u>Licenses and Permits</u>. The Company Group has all licenses, franchises, permits, orders or approvals or other similar authorizations required under applicable Law (the "<u>Permits</u>") necessary to operate the Business, except for the failure to have any Permits, individually or in the aggregate, that is not reasonably likely to have a Material Adverse Effect. Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, such Permits are valid and in full force and effect, and none of the Permits will be terminated or impaired or become terminable as a result of the transactions contemplated hereby.

4.17 <u>Compliance with Laws</u>. The Company Group is not in violation of, has not violated, in each case in any material respect, and to the Company's knowledge, is neither under investigation with respect to nor has been threatened in writing to be charged with or given notice of any violation or alleged violation of, any Law, or judgment, order or decree entered by any court, arbitrator or Authority, domestic or foreign, nor, to the knowledge of the Company, is there any basis for any such charge and since July 6, 2018, the Company Group has not received any subpoenas from any Authority.

(a) Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, without limiting the foregoing paragraph, the Company Group is not in violation of, has not violated, and to the knowledge of the Company, is not under investigation with respect to nor has been threatened in writing or charged with or given notice of any violation of any provisions of:

(i) any Law applicable due to the specific nature of the Business, including Laws applicable to data privacy, data security and/or personal information ("<u>Data Protection Laws</u>") and Laws applicable to lending activities;

 (ii) the Foreign Corrupt Practices Act of 1977 (§§ 78dd-1 et seq.), as amended (the "Foreign Corrupt Practices Act");

(iii) any comparable or similar Law of any jurisdiction; or

(iv) any Law regulating or covering conduct in, or the nature of, the workplace, including regarding sexual harassment or, on any impermissible basis, a hostile work environment.

(b) Without limiting the foregoing paragraph, neither the Company Group nor, to the knowledge of the Company, any director, officer, agent, employee, Affiliate or Person acting on behalf of the Company is currently subject to any U.S. sanctions administered by the Office of Foreign Assets Control of the U.S. Treasury Department ("<u>OFAC</u>"). The Company Group has not engaged in transactions with, or exported any of its products or associated technical data (i) into (or to a national or resident of) Cuba, Iran, Iraq, Libya, North Korea, Syria or any other country to which the United States has embargoed goods to or has proscribed economic transactions with or (ii) to the knowledge of the Company, to any Person included on the United States Treasury Department's list of Specially Designated Nationals or the U.S. Commerce Department's Denied Persons List.

4.18 Intellectual Property.

(a) <u>Schedule 4.18</u> sets forth a true, correct and complete list of all Intellectual Property Rights that are subject to a patent, registration or application therefor with any Authority, specifying as to each, as

applicable: (i) the nature of such Intellectual Property Right (*.e.*, whether registered or a pending application); (ii) the owner of such Intellectual Property Right; and (iii) the jurisdictions by or in which such Intellectual Property Right has been issued or registered or in which an application for such issuance or registration has been filed.

(b) Since July 6, 2018, the Company Group has not been sued or charged in writing with or been a defendant in any Action that involves a claim of infringement of any Intellectual Property Rights, and the Company has no knowledge of any other claim of infringement by the Company Group, and no knowledge of any continuing infringement by any other Person of any Intellectual Property Rights of the Company Group.

(c) Any Intellectual Property Rights used by the Company Group in the performance of any Contract (existing as of the date of this Agreement) by the Company Group, which are purported to be owned by the Company Group prior to initiation of such performance, remain owned by the Company Group upon the performance of such Contract by the Company Group, and, to the knowledge of the Company, no third party has any claim of ownership on any such Intellectual Property Rights, other than any Permitted Liens.

(d) All employees, agents, consultants or contractors who have contributed to or participated in the creation or development of any copyrightable, patentable or trade secret material on behalf of the Company Group or any predecessor in interest thereto either: (i) is a party to a "work-for-hire" agreement under which a member of the Company Group is deemed to be the original owner/author of all property rights therein; or (ii) has executed an assignment or an agreement to assign in favor of a member of the Company Group (or such predecessor in interest, as applicable) all right, title and interest in such material.

(e) None of the execution, delivery or performance by the Company of this Agreement or any of the Additional Agreements to which the Company is a party or the consummation by the Company of the transactions contemplated hereby or thereby will cause any material item of Intellectual Property Rights owned, licensed, used or held for use by the Company Group immediately prior to the Closing to not be owned, licensed or available for use by the Company Group on substantially the same terms and conditions immediately following the Closing.

(f) The Company has taken commercially reasonable measures to safeguard and maintain the confidentiality and value of all trade secrets and other items of Intellectual Property Rights that are confidential and all other confidential information, data and materials licensed by the Company Group or otherwise used in the operation of the Business. Except as would not, individually or in the aggregate, be reasonably likely to have a Material Adverse Effect, the transactions contemplated by this Agreement will not result in the violation of any Data Protection Laws or the privacy policies of the Company Group.

4.19 Employees.

(a) No director or executive officer of the Company (i) is currently on leave, (ii) has given written notice of his or her intent to terminate his or her relationship with the Company, or (iii) has received written notice of such termination from the Company. To the actual knowledge of the Company, no Company Employee or independent contractor (but specifically excluding all executive officers) of the Company that earned an aggregate amount of compensation in excess of \$250,000 in the fiscal year ended March 31, 2019, has communicated to Company his or her intention to terminate his or her employment or the contractor relationship, as applicable, with the Company. To the knowledge of the Company, there are no proceedings, governmental investigations, or administrative proceedings against the Company regarding the Company Employees or the Company's employment practices.

(b) The Company is not a party to or subject to any collective bargaining agreements. As of the date of this Agreement, no labor union or other collective bargaining unit represents or claims to represent any of the Company Employees and, to the actual knowledge of the Company, there is no union campaign being conducted to solicit cards from employees to authorize a union to request a National Labor Relations Board certifications election with respect to the Company's or the Subsidiaries' employees.

(c) There are no pending or, to the knowledge of the Company, threatened in writing, claims or proceedings against the Company Group under any worker's compensation policy or long-term disability policy.

4.20 <u>Employment Matters</u>. To the knowledge of the Company, as of the date of this Agreement, no Company Employee, in the ordinary course of his or her duties, has breached any obligation to a former employer in

respect of any covenant against competition or soliciting clients or employees or servicing clients or confidentiality or any proprietary right of such former employer.

4.21 <u>Withholding</u>. All obligations of the Company applicable to its Company Employees, whether arising by operation of Law or contract, or attributable to payments by the Company to trusts or other funds or to any governmental agency, with respect to unemployment compensation benefits, social security benefits or any other benefits for Company Employees with respect to the employment of Company Employees through the date hereof have been paid or adequate accruals therefor have been made on the Financial Statements. All reasonably anticipated obligations of the Company with respect to Company Employees (except for those related to wages during the pay period immediately prior to the Closing Date and arising in the ordinary course of business), whether arising by operation of Law or contract for accrued salaries and holiday pay, earned bonuses and other earned forms of compensation payable to Company Employees in respect of the services rendered by any of them prior to the date hereof have been or will be paid by the Company prior to the Closing Date.

4.22 Employee Benefits and Compensation.

(a) Each "employee benefit plan" (as defined in Section 3(3) of ERISA), bonus, deferred compensation, equity-based or non-equity-based incentive, severance or other plan or written agreement relating to employee or director benefits or employee or director compensation or fringe benefits, maintained or contributed to by the Company Group and/or with respect to which the Company Group could reasonably be expected to incur any direct or indirect, fixed or contingent liability (each a "<u>Plan</u>" and collectively, the "<u>Plans</u>") is and has been maintained in all material respects in compliance with all applicable laws, including (if applicable) ERISA, and has been administered and operated in all material respects in accordance with its terms.

(b) Each Plan which is intended to be "qualified" within the meaning of Section 401(a) of the Code, has received a favorable determination letter from the IRS and, to the knowledge of the Company, no event has occurred and no condition exists which could reasonably be expected to result in the revocation of any such determination. No event which constitutes a "reportable event" (as defined in Section 4043(c) of ERISA) for which the 30-day notice requirement has not been waived by the Pension Benefit Guaranty Corporation (the "<u>PBGC</u>") has occurred with respect to any Plan. No Plan subject to Title IV of ERISA. Full payment has been made of all amounts which the Company was required under the terms of the Plans to have paid as contributions to such Plans on or prior to the date hereof (excluding any amounts not yet due) and no Plan which is subject to Part 3 of Subtile B of Title I of ERISA has incurred an "accumulated funding deficiency" (within the meaning of Section 302 of ERISA or Section 412 of the Code), whether or not waived.

(c) Neither the Company nor to the knowledge of the Company, any other "disqualified person" or "party in interest" (as defined in Section 4975(e)(2) of the Code and Section 3(14) of ERISA, respectively), has engaged in any transaction in connection with any Plan that could reasonably be expected to result in the imposition of a penalty pursuant to Section 502(i) of ERISA, damages pursuant to Section 409 of ERISA or a tax pursuant to Section 4975(a) of the Code. The Company has not maintained any Plan (other than a Plan which is intended to be "qualified" within the meaning of Section 401(a) of the Code) which provides benefits with respect to current or former employees or directors following their termination of service with the Company (other than as required pursuant to COBRA). Each Plan subject to the requirements of COBRA has been operated in substantial compliance therewith.

(d) No Company Employee will accrue or receive additional benefits, service or accelerated rights to payment of benefits as a direct result of the transactions contemplated hereby. No material liability, claim, investigation, audit, action or litigation has been incurred, made, commenced or, to the knowledge of the Company, threatened in writing, by or against any Plan or the Company with respect to any Plan (other than for benefits payable in the ordinary course and PBGC insurance premiums). No Plan or related trust owns any securities in violation of Section 407 of ERISA. With respect to each Plan which is an "employee pension benefit plan" (as defined in Section 3(2) of ERISA) as of the most recent actuarial valuation report prepared for each such Plan, the aggregate present value of the accrued liabilities thereof (determined in accordance with Statement of Financial Accounting Standards No. 35) did not exceed the aggregate fair market value of the assets allocable thereto.

(e) No Plan is a "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA) and the Company has not been obligated to contribute to any multiemployer plan. No material liability has been, or could

reasonably be expected to be, incurred under Title IV of ERISA (other than for PBGC insurance premiums payable in the ordinary course) or Section 412(f) or (n) of the Code, by the Company or any entity required to be aggregated with the Company pursuant to Section 4001(b) of ERISA and/or Section 414 (b), (c), (m) or (o) of the Code with respect to any "employee pension benefit plan" (as defined in Section 3(2) of ERISA).

(f) There is no unfunded non-tax-qualified Plan which provides a pension or retirement

benefit.

4.23 Real Property.

(a) The Company Group does not own, or otherwise have an interest in, any Real Property, including under any Real Property lease, sublease, space sharing, license or other occupancy agreement. The Company Group has good, valid and subsisting title to its respective leasehold estates in the offices described on <u>Schedule 4.23</u>, free and clear of all Liens (other than Permitted Liens). Other than as would not reasonably be expected to have a Material Adverse Effect, the Company Group has not breached or violated any local zoning ordinance, and no notice from any Person has been received by the Company Group or served upon the Company Group claiming any violation of any local zoning ordinance.

(b) With respect to each Lease: (i) it is valid, binding and in full force and effect against the Company Group, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity; (ii) except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, all rents and additional rents and other sums, expenses and charges due thereunder have been paid; (iii) the lessee has been in peaceable possession since the commencement of the original term thereof in all material respects; (iv) no waiver, indulgence or postponement of the lessee's material obligations thereunder has been granted by the lessor; (v) there exists no material default or event of default thereunder by the Company Group or, to the Company's knowledge, by any other party thereto; (vi) there exists no occurrence, condition or act which, with the giving of notice, the lapse of time or the happening of any further event or condition, would become a material default or event of default by the Company Group thereunder; and (vii) to the knowledge of the Company Group, there are no outstanding claims of breach or indemnification or notice of default or termination thereunder. The Real Property leased by the Company Group is in a state of maintenance and repair adequate and suitable for the purposes for which it is presently being used, and there are no repair or restoration works likely to be required in connection with any of the leased Real Properties, in each case except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect. The Company Group is in physical possession and actual and exclusive occupation of the whole of the leased property, none of which is subleased or assigned to another Person. The Lease leases all useable square footage of the premise located at the leased Real Property. To the Company's knowledge, the Company Group does not owe any brokerage commission with respect to any Real Property.

4.24 Tax Matters.

The Company Group has or will have timely filed, or caused to be timely filed, (a) (taking into account valid extensions) all income and other material Tax Returns required to be filed by the Company Group, except where the failure to so file would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company Group, which Tax Returns are correct and complete in all material respects, and has paid all Taxes required to be paid by the Company Group other than such Taxes for which adequate reserves in the Financial Statements have been established, and except for such Taxes the non-payment of which would not reasonably be expected to have a Material Adverse Effect on the Company Group. There are no Actions pending against the Company Group in respect of any material Tax, and the Company Group has not been notified in writing of any proposed material Tax claims or assessments against the Company Group (other than, in each case, claims or assessments that have been settled or otherwise resolved in full). There are no material Liens with respect to any Taxes upon any of the Company Group's assets, other than Permitted Liens. The Company Group has no outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes that has not been paid. There are no outstanding requests by the Company Group for any extension of time within which to file any Tax Return other than an extension requested in the ordinary course of business.

(b) The Company Group has not taken any action, and does not have any knowledge of any fact or circumstance, that could reasonably be expected to prevent the transactions contemplated hereby, including

the Share Exchange, from qualifying as a "reorganization" for U.S. federal income tax purposes within the meaning of Section 368(a)(1) of the Code.

4.25 Environmental Laws.

(a) The Company Group has not (i) received any written notice of any alleged claim, violation of or liability under any Environmental Law which has not heretofore been cured or for which there is any remaining liability; (ii) disposed of, emitted, discharged, handled, stored, transported, used or released any Hazardous Materials, arranged for the disposal, discharge, storage or release of any Hazardous Materials, or exposed any Company Employee or other individual to any Hazardous Materials so as to give rise to any liability or corrective or remedial obligation under any Environmental Laws; or (iii) entered into any agreement that may require it to guarantee, reimburse, pledge, defend, hold harmless or indemnify any other Person with respect to liabilities arising out of Environmental Laws.

(b) Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, there are no Hazardous Materials in, on, or under any properties owned, leased or used at any time by the Company Group such as could give rise to any material liability or corrective or remedial obligation of the Company Group under any Environmental Laws.

4.26 <u>Finders' Fees</u>. There is no investment banker, broker, finder or other intermediary which has been retained by or is authorized to act on behalf of the Company Group or any of Affiliates who might be entitled to any fee or commission from the Company, Purchaser or any of their Affiliates upon consummation of the transactions contemplated by this Agreement.

4.27 Preclinical Development and Clinical Trials. The studies, tests, preclinical development and clinical trials, if any, conducted by or on behalf of the Company are being conducted in all material respects in accordance with experimental protocols, procedures and controls pursuant to accepted professional and scientific standards for products or product candidates comparable to those being developed by the Company and all applicable laws and regulations, including the Federal Food, Drug, and Cosmetic Act and 21 C.F.R. parts 50, 54, 56, 58, 312, and 812. The descriptions of, protocols for, and data and other results of, the studies, tests, development and trials conducted by or on behalf of the Company that have been furnished or made available to the Purchaser or as provided in the Proxy Statement are accurate and complete in all material respects (other than to the extent certain portions thereof were redacted by the Company). The Company is not aware of any studies, tests, development and trials conducted by or on behalf of the Company, and the Company has not received any notices or correspondence from the FDA or any other governmental Authority or any Institutional Review Board or comparable authority requiring the termination, suspension or material modification of any studies, tests, preclinical development or clinical trials conducted by or on behalf of the Company.

4 28 FDA Approvals. Except as would not be, individually or in the aggregate, reasonably likely to have a Material Adverse Effect, the Company possesses all permits, licenses, registrations, certificates, authorizations, orders and approvals from the appropriate federal, state or foreign regulatory authorities necessary to conduct its business as now conducted, including all such permits, licenses, registrations, certificates, authorizations, orders and approvals required by the U.S. Food and Drug Administration ("FDA") or any other federal, state or foreign agencies or bodies engaged in the regulation of drugs, pharmaceuticals, medical devices or biohazardous materials. The Company has not received any written notice of proceedings relating to the suspension, modification, revocation or cancellation of any such permit, license, registration, certificate, authorization, order or approval. Neither the Company nor, to the Company's knowledge, any officer, employee or agent of the Company has been convicted of any crime or engaged in any conduct that has caused or would reasonably be expected to result in (A) disqualification or debarment by the FDA under 21 U.S.C. Sections 335(a) or (b), or any similar law, rule or regulation of any other governmental Authorities, (B) debarment, suspension, or exclusion under any Federal Healthcare Programs or by the General Services Administration, or (C) exclusion under 42 U.S.C. Section 1320a-7 or any similar law, rule or regulation of any governmental Authorities. Neither the Company nor to the knowledge of the Company, any of its officers, employees, contractors or agents, is the subject of any investigation by FDA pursuant to its "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities" policy as stated at 56 Fed. Reg. 46191 (September 10, 1991) (the 'FDA Application Integrity Policy') and any amendments thereto, or by any other similar governmental Authority pursuant to any similar policy. Neither the Company nor, to the Company's knowledge, any of its officers, employees, contractors, and agents has committed any act, made any
statement or failed to make any statement that would reasonably be expected to provide a basis for FDA to invoke the FDA Application Integrity Policy or for any similar governmental Authority to invoke a similar policy. Neither the Company nor to the Company's knowledge, any of its officers, employees, contractors or agents has made any materially false statements on, or material omissions from, any notifications, applications, approvals, reports and other submissions to FDA or any similar governmental Authority.

4.29 <u>FDA Regulation</u>. The Company is and has been in compliance with all applicable Laws administered or issued by FDA or any similar governmental entity, including the Federal Food, Drug, and Cosmetic Act and all other Laws regarding developing, testing, manufacturing, marketing, distributing or promoting the products of the Company, or complaint handling or adverse event reporting, except as would not reasonably be expected to have a Material Adverse Effect.

4.30 Information Supplied. None of the information supplied by Company at the request of the Purchaser for inclusion or incorporation by reference: (a) in any current report of the Purchaser on Form 8-K, and any exhibits thereto or any other report, form, registration or other filing made with any governmental Authority with respect to the transactions contemplated hereby; (b) in the Proxy Statement; or (c) in the mailings or other distributions to Purchaser's stockholders and/or prospective investors with respect to the consummation of the transactions contemplated hereby or in any amendment to any of documents identified in clauses (a) through (c), will, when filed, made available, mailed or distributed, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading, but in each case only to the extent that the untrue statement or alleged untrue statement or omission or alleged omission was made in reliance upon and in conformity with the information provided by the Company for inclusion in such documents identified in clauses (a) through (c) above to be publicly filed with the SEC. Notwithstanding the foregoing, Company makes no representation, warranty or covenant with respect to any information supplied by or on behalf of Purchaser or its Affiliates.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF PURCHASER

Except as disclosed in the Purchaser SEC Documents filed with or furnished to the SEC prior to the date of this Agreement (other than disclosures in the "<u>Risk Factors</u>" or "<u>Forward Looking Statements</u>" sections of any Purchaser SEC Document and other disclosures to the extent that such disclosure is predictive or forward-looking in nature), Purchaser hereby represents and warrants to the Company that each of the following representations and warranties are true, correct and complete as of the date of this Agreement and as of the Closing Date (except to the extent expressly made as of an earlier date, in which case such representations and warranties shall be so true, correct and complete as of such earlier date):

5.1 <u>Corporate Existence and Power</u>. Purchaser is a corporation duly incorporated, validly existing and in good standing under the laws of Delaware and has all necessary power and authority: (a) to conduct its business in the manner in which its business is currently being conducted; (b) to own and use its assets in the manner in which its assets are currently owned and used; and (c) to perform its obligations under all Contracts by which it is bound, except where any such failure would not reasonably be expected to have a Purchaser Material Adverse Effect. Purchaser has either delivered or made available to Company, including via the SEC's Electronic Data Gathering Analysis and Retrieval ("<u>EDGAR</u>") system database, accurate and complete copies of the certificate of incorporation, bylaws and other charter and organizational documents of Purchaser, including all amendments thereto. Purchaser has no subsidiaries.

5.2 <u>Corporate Authorization</u>. The execution, delivery and performance by the Purchaser of this Agreement and the Additional Agreements and the consummation by the Purchaser of the transactions contemplated hereby and thereby are within the corporate powers of the Purchaser and have been duly authorized by all necessary corporate action on the part of the Purchaser. This Agreement has been duly executed and delivered by the Purchaser and, assuming due authorization, execution and delivery by each other party hereto and to the Additional Agreements, constitutes, and upon their execution and delivery, the Additional Agreements will constitute, a valid and legally binding agreement of the Purchaser, enforceable against it in accordance with its and their terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity. (i) The affirmative vote of holders of a majority of the

outstanding shares of Purchaser Common Stock entitled to vote at the Purchaser Stockholder Meeting, assuming a quorum is present, for the approval of the Purchaser Charter Amendment and (ii) the affirmative vote of the holders of a majority of the shares of Purchaser Common Stock present in person or by proxy and entitled to vote thereon at the Purchaser Stockholder Meeting for the approval of the Purchaser Stockholder Matters (other than the Purchaser Charter Amendment) are the only votes of any of the Purchaser's capital stock necessary to adopt this Agreement and to consummate the Business Combination and the other transactions contemplated hereby.

5.3 <u>Governmental Authorization</u>. Assuming the accuracy of the representations and warranties set forth in <u>Section 4.3</u>, neither the execution, delivery nor performance of this Agreement or any Additional Agreement requires any Governmental Approval, except for filings under the HSR Act or any other applicable Laws relating to antitrust.

5.4 <u>Non-Contravention</u>. The execution, delivery and performance by the Purchaser of this Agreement and the Additional Agreements does not and will not (a) provided that holders of fewer than the number of Purchaser Common Stock specified in the Purchaser's organizational documents exercise their redemption rights with respect to such transaction, contravene or conflict with the organizational or constitutive documents of the Purchaser, (b) contravene or conflict with or constitute a violation of any provision of any Law, judgment, injunction, order, writ, or decree binding upon the Purchaser, (c) require consent, conflict with, result in a breach of, or constitute a default (with or without the giving of notice or the passage of time or both), or violate or give rise to any right of termination, cancellation, amendment or acceleration of any right or obligation of Purchaser or require any payment or reimbursement or to a loss of any material benefit on the part of Purchaser under any Contract, (d) result in the creation or imposition of any Lien on any of the Purchaser Securities, or (e) result in the creation or imposition of any Lien (except for Permitted Liens) on any of Purchaser's assets, except as would not, individually or in the aggregate, reasonably likely to have a Purchaser Material Adverse Effect.

5.5 <u>Finders' Fees</u>. Except for the Deferred Underwriting Amount, there is no investment banker, broker, finder or other intermediary which has been retained by or is authorized to act on behalf of the Purchaser or its Affiliates who might be entitled to any fee or commission from the Purchaser or any of its Affiliates upon consummation of the transactions contemplated by this Agreement or any of the Additional Agreements.

5.6 <u>Issuance of Shares</u>. The Closing Payment Shares, the Earnout Shares and the Purchaser Preferred Stock, when issued in accordance with this Agreement, will be duly authorized and validly issued, and will be fully paid and nonassessable, free and clear of all Liens, and the issuance and sale of such Purchaser Common Stock pursuant hereto will not be subject to or give rise to any preemptive rights or rights of first refusal.

5.7 Capitalization. The authorized capital stock of Purchaser consists of 30,000,000 shares of common stock, par value \$0.0001 per share, of which 14.375,000 are issued and outstanding as of the date hereof. In addition, 20,000,000 Purchaser Warrants (inclusive of Purchaser Private Warrants and Purchaser Public Warrants included in any outstanding Purchaser Units) are issued and outstanding as of the date hereof. No other shares of capital stock or other voting securities of Purchaser are issued, reserved for issuance or outstanding. All issued and outstanding shares of Purchaser Common Stock are duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the Delaware General Corporation Law, the Purchaser's organizational documents or any contract to which Purchaser is a party or by which Purchaser is bound. Except as set forth in the Purchaser's organizational documents, there are no outstanding contractual obligations of Purchaser to repurchase, redeem or otherwise acquire any shares of Purchaser Common Stock, Purchaser Preferred Stock or any capital equity of Purchaser. There are no outstanding contractual obligations of Purchaser to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any other Person. There are no outstanding contractual obligations of Purchaser to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any other Person. There are no: (a) outstanding subscriptions, options, warrants, rights (including "phantom stock rights"), calls, commitments, understandings, conversion rights, rights of exchange, plans or other agreements of any kind providing for the purchase, issuance or sale of any shares of the Purchaser, or (b) any agreement to which Purchaser is a party, or to the knowledge of the Purchaser, any other agreements, with respect to the Purchaser Common Stock or Purchaser Preferred Stock, including any voting trust, other voting agreement or proxy with respect thereto.

5.8 Information Supplied. None of the information supplied or to be supplied by the Purchaser expressly for inclusion or incorporation by reference in the filings with the SEC and mailings to Purchaser's stockholders with respect to the solicitation of proxies to approve the transactions contemplated by this Agreement and the Additional Agreements, if applicable, or in any other Purchaser SEC Documents, will, at the date of filing and/ or mailing, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading (subject to the qualifications and limitations set forth in the materials provided by Purchaser or that is included in the Purchaser SEC Documents).

Trust Fund. As of the date of this Agreement, Purchaser has at least \$116,000,000 in the trust fund established by Purchaser for the benefit of its public stockholders (the "Trust Fund") in a trust account maintained by Continental Stock Transfer & Trust Company (the "Trustee") at Morgan Stanley Bank, N.A. (the "Trust Account"), and such monies are invested in "government securities" (as such term is defined in the Investment Company Act of 1940, as amended) and held in trust by the Trustee pursuant to the Investment Management Trust Agreement, dated as of May 9, 2019, between Purchaser and the Trustee (the 'Trust Agreement"). The Trust Agreement is valid and in full force and effect and enforceable in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity, and has not been amended or modified. There are no separate agreements, side letters or other agreements or understandings (whether written or unwritten, express or implied) that would cause the description of the Trust Agreement in the Purchaser SEC Documents to be inaccurate in any material respect and/or that would entitle any Person (other than stockholders of Purchaser holding shares of Purchaser Common Stock sold in Purchaser's IPO who shall have elected to redeem their shares of Purchaser Common Stock pursuant to the Purchaser's certificate of incorporation) to any portion of the proceeds in the Trust Account. Prior to the Closing, none of the funds held in the Trust Account may be released except in accordance with the Purchaser SEC Documents, the Trust Agreement and the Purchaser's organizational documents. Purchaser has performed all material obligations required to be performed by it to date under, and is not in material default or delinquent in performance or any other respect (claimed or actual) in connection with, the Trust Agreement, and, to the knowledge of Purchaser, no event has occurred which, with due notice or lapse of time or both, would constitute such a material default thereunder. There are no claims or proceedings pending with respect to the Trust Account.

5.10 <u>Listing</u>. The Purchaser Units, Purchaser Common Stock and Purchaser Warrants (collectively, the "<u>Purchaser Securities</u>") are listed on Nasdaq, with trading tickers HSACU, HSAC and HSACW. As of the date of this Agreement, (a) Purchaser is in compliance with applicable continued listing requirements of Nasdaq, (b) Purchaser has not received any written deficiency notice from Nasdaq relating to the continued listing requirements of such Purchaser Securities, (c) there are no Actions pending or, to the knowledge of Purchaser, threatened against Purchaser by the Financial Industry Regulatory Authority or any other Person with respect to the continued listing of the Purchaser Securities on Nasdaq, including any intention by such entity to suspend, prohibit or terminate the quoting of such Purchaser Securities on Nasdaq and (d) such Purchaser Securities are in compliance with all of the applicable listing and corporate governance rules of Nasdaq.

5.11 <u>Board Approval</u>. The Purchaser's board of directors (including any required committee or subgroup of such board) (the "<u>Purchaser Board</u>") has, as of the date of this Agreement, unanimously (a) declared the advisability of the transactions contemplated by this Agreement, (b) determined that the transactions contemplated hereby are in the best interests of the stockholders of Purchaser, (c) determined that the transactions contemplated hereby constitutes a "Business Combination" as such term is defined in Purchaser's amended and restated certificate of incorporation and bylaws (a "<u>Business Combination</u>") and (d) recommended that the Purchaser's stockholders approve the Purchaser Stockholder Matters (the "<u>Purchaser Board</u> <u>Recommendation</u>").

5.12 Purchaser SEC Documents and Financial Statements.

(a) Purchaser has filed all forms, reports, schedules, statements and other documents, including any exhibits thereto, required to be filed or furnished by Purchaser with the SEC through EDGAR since Purchaser's formation under the Exchange Act or the Securities Act, together with any amendments, restatements or supplements thereto (the "<u>Purchaser SEC Documents</u>"). The Purchaser SEC Documents were prepared in all material respects in accordance with the requirements of the Securities Act, the Exchange Act, and the Sarbanes-Oxley Act, as the case may be, and the rules and regulations thereunder. The Purchaser SEC Documents did not, at the time they were filed with the SEC (except to the extent that information contained in any Purchaser

SEC Document or Purchaser SEC Document has been or is revised or superseded by a later filed Purchaser SEC Document then on the date of such filing) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. As used in this <u>Section 5.12</u>, the term "file" shall be broadly construed to include any manner in which a document or information is furnished, supplied or otherwise made available to the SEC.

(b) The financial statements and notes contained or incorporated by reference in the Purchaser SEC Documents (the "<u>Purchaser Financials</u>"), (i) were prepared in accordance with U.S. GAAP consistently applied, (ii) comply with all applicable accounting requirements under the Securities Act and the rules and regulations of the SEC thereunder, and (iii) fairly present in all material respects, in conformity with U.S. GAAP applied on a consistent basis, the financial position of the Purchaser as of the dates thereof and the results of operations of the Purchaser for the periods reflected therein (subject to normal and recurring year-end adjustments that are not, individually or in the aggregate, material).

(c) The Purchaser makes and keeps accurate Books and Records and maintain a system of internal accounting controls designed, and which the Purchaser believes is sufficient, to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorization; (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with U.S. GAAP and to maintain accountability for assets; (iii) access to assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

(d) The Purchaser has established and maintains disclosure controls and procedures (as defined in Rules 13a-15 and 15d-15 under the Exchange Act), which (i) are designed to ensure that material information relating to the Purchaser is made known to the Purchaser's principal executive officer and its principal financial officer by others and (ii) are effective in all material respects to perform the functions for which they were established. Since the Purchaser's inception, there have been no significant deficiencies or material weakness in the Purchaser's internal control over financial reporting (whether or not remediated) and no change in the Purchaser's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Purchaser's internal control over financial reporting.

5.13 <u>Compliance with Laws</u>. The Purchaser is not in violation of, has not violated, and to the Purchaser's knowledge, is neither under investigation with respect to nor has been threatened in writing to be charged with or given notice of any violation or alleged violation of, any Law, or judgment, order or decree entered by any court, arbitrator or Authority, domestic or foreign, nor is there any basis for any such charge and since the IPO, the Purchaser has not received any subpoenas from any Authority, in each case except as would not be, individually or in the aggregate, reasonably likely to have a Purchaser Material Adverse Effect. Except as would not be, individually or in the aggregate, reasonably likely to have a Purchaser Material Adverse Effect, without limiting the foregoing paragraph, the Purchaser is not in violation of, has not violated, and to the knowledge of the Purchaser, is not under investigation with respect to nor has been threatened in writing or charged with or given notice of any violation of any provisions of:

- (a) Data Protection Laws and Laws applicable to lending activities;
- (b) the Foreign Corrupt Practices Act;
- (c) any comparable or similar Law of any jurisdiction; or

(d) any Law regulating or covering conduct in, or the nature of, the workplace, including regarding sexual harassment or, on any impermissible basis, a hostile work environment.

5.14 <u>OFAC</u>. Neither the Purchaser, nor any director or officer of the Purchaser (nor, to the knowledge of the Purchaser, any agent, employee, affiliate or Person acting on behalf of the Purchaser) is currently identified on the specially designated nationals or other blocked person list or otherwise currently subject to any U.S. sanctions administered by the OFAC; and the Purchaser has not, directly or indirectly, used any funds, or loaned, contributed or otherwise made available such funds to any subsidiary, joint venture partner or other Person, in connection with any sales or operations in Cuba, Iran, Iraq, Libya, North Korea, Syria or any other country

sanctioned by OFAC or for the purpose of financing the activities of any Person currently subject to, or otherwise in violation of, any U.S. sanctions administered by OFAC in the last five (5) fiscal years.

5.15 <u>Litigation</u>. There is no Action (or any basis therefore) that would be material to the Purchaser pending against, or to the best knowledge of the Purchaser, threatened against or affecting, the Purchaser, any of its officers or directors with respect to the business of the Purchaser or any securities of the Purchaser or any of the Purchaser's assets or any material contract of the Purchaser before any court, Authority or official or which in any manner challenges or seeks to prevent or enjoin the transactions contemplated hereby or by the Additional Agreements. There are no outstanding judgments against the Purchaser that would be, individually or in the aggregate, reasonably likely to have a material adverse effect on the ability of Purchaser to enter into and perform its obligations under this Agreement.

5.16 <u>Absence of Certain Changes</u>. Since its formation, the Purchaser has (a) conducted no business other than its formation, the public offering of its securities (and the related private offerings), public reporting and its search for an initial Business Combination as described in the IPO Prospectus (including the investigation of the Company Group and the negotiation and execution of this Agreement) and related activities and (b) not been subject to a Purchaser Material Adverse Effect.

5.17 <u>Employees and Employee Benefit Plans</u>. The Purchaser does not (a) have any paid employees or (b) maintain, sponsor, contribute to or otherwise have any liability under, any Plans.

5.18 <u>Properties</u>. The Purchaser does not own, license or otherwise have any right, title or interest in any material intellectual property. The Purchaser does not own, or otherwise have an interest in, any Real Property, including under any Real Property lease, sublease, space sharing, license or other occupancy agreement.

5.19 Contracts.

(a) Other than this Agreement and the Additional Agreements, there are no Contracts to which the Purchaser is a party or by which any of its properties or assets may be bound, subject or affected, which (a) creates or imposes a liability greater than \$200,000, (b) may not be cancelled by the Purchaser on less than sixty (60) calendar days' prior notice without payment of a material penalty or termination fee or (c) prohibits, prevents, restricts or impairs in any material respect any business practice of the Purchaser as its business is currently conducted, any acquisition of material property by the Purchaser, or restricts in any material respect the ability of the Purchaser from engaging in business as currently conducted by it or from competing with any other Person (each such contract, a "<u>Purchaser Material Contract</u>"). All Purchaser Material Contracts have been made available to the Company other than those that are exhibits to the Purchaser SEC Documents.

(b) With respect to each Purchaser Material Contract: (i) the Purchaser Material Contract is legal, valid, binding and enforceable in all material respects against the Purchaser and, to the knowledge of the Purchaser, the other parties thereto, and is in full force and effect (except as may be limited by bankruptcy, insolvency, reorganization or other similar laws affecting the enforcement of creditors' rights generally and by general principles of equity); (ii) the Purchaser is not in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default in any material respect by the Purchaser, or permit termination or acceleration by the other party, under such Purchaser Material Contract; and (iii) to the knowledge of the Purchaser, no other party to any Purchaser Material Contract is in breach or default in any material respect, and no event thas occurred that with the passage of time or giving of notice or both would constitute such a breach or default in any material contract; and (iii) to the knowledge of the Purchaser, no other party to any Purchaser Material Contract is in breach or default in any material respect, and no event has occurred that with the passage of time or giving of notice or both would constitute such a breach or default by such other party, or permit termination or acceleration by the Purchaser Material Contract.

5.20 <u>Insurance</u>. Except as would not be material to the Purchaser, the Purchaser is insured by recognized, financially sound and reputable institutions with policies in such amounts and with such deductibles and covering such risks as are customarily carried by Persons conducting a business similar to the Purchaser. The Purchaser has no reason to believe that it will not be able (i) to renew its existing insurance coverage as and when such policies expire or (ii) to obtain comparable coverage from similar institutions as may be necessary or appropriate to conduct its business as now conducted and at a cost that would not reasonably be expected to have a Purchaser Material Adverse Effect. The Purchaser has not been denied any insurance coverage which it has sought or for which it has applied.

5.21 <u>Taxes</u>.

(a) The Purchaser has or will have timely filed, or caused to be timely filed (taking into account valid extensions), all income and other material Tax Returns required to be filed by it, except where the failure to so file would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the Purchaser, which Tax Returns are correct and complete in all material respects, and has paid all Taxes required to be paid by the Purchaser other than such Taxes for which adequate reserves in the Purchaser Financials have been established, and except for such Taxes the non-payment of which would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on the Purchaser. There are no Actions pending against the Purchaser in respect of any material Tax, and the Purchaser has not been notified in writing of any proposed material Tax claims or assessments against the Purchaser (other than, in each case, claims or assessments that have been settled or otherwise resolved in full). There are no Liens with respect to any Taxes upon any of the Purchaser's assets, other than Permitted Liens. The Purchaser has no outstanding waivers or extensions of any applicable statute of limitations to assess any material amount of Taxes that has not been paid. There are no outstanding requested in the ordinary course of business.

(b) The Purchaser has not taken any action, and does not have any knowledge of any fact or circumstance, that could reasonably be expected to prevent the transactions contemplated hereby, including the Share Exchange, from qualifying as a "reorganization" for U.S. federal income tax purposes within the meaning of Section 368(a)(1) of the Code.

5.22 Independent Investigation. Purchaser and its Affiliates and their respective Representatives have conducted their own independent investigation, review and analysis of the business, results of operations, prospects, condition (financial or otherwise) or assets of the Company Group, and Purchaser acknowledges that it and they have been provided adequate access to the personnel, properties, assets, premises, books and records, and other documents and data of the Company Group for such purpose. The Purchaser acknowledges and agrees that: (a) in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, it has relied solely upon its own investigation and the express representations and warranties of the Company set forth in <u>Article IV</u> (including the related portions of the Disclosure Schedules); and (b) none of the Company, its Affiliates nor their respective Representatives have made any express or implied representation or warranty as to the Company Group, or this Agreement, except as expressly set forth in <u>Article IV</u> (including the related portions of the Disclosure Schedules).

ARTICLE VI REPRESENTATIONS AND WARRANTIES OF THE STOCKHOLDERS

Each Stockholder hereby represents and warrants to the Purchaser, as to itself, that each of the following representations and warranties are true, correct and complete as of the date of this Agreement and as of the Closing Date, except to the extent expressly made as of an earlier date, in which case such representations and warranties shall be so true, correct and complete as of such earlier date.

6.1 Ownership of Stock; Authority.

(a) The Stockholder has good and valid title to its Company Common Shares, free and clear of any and all Liens other than restrictions under applicable securities Laws and Company Charter Documents.

(b) The Stockholder has full legal capacity, power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. This Agreement has been, or at Closing will be, duly executed and delivered by the Stockholder and are, or, assuming due authorization, execution and delivery by each other party thereto, upon its execution and delivery will be, valid and legally binding obligation of the Stockholder, enforceable against the Stockholder in accordance with its terms, subject to (i) laws of general application relating to bankruptcy, insolvency and the relief of debtors, or (ii) rules of law governing specific performance, injunctive relief or other equitable remedies.

(c) Neither the execution and delivery by the Stockholder of any or all of the Agreement, nor the consummation by the Stockholder of the transactions contemplated thereby, will result in the imposition of any Lien upon the Company Common Shares.

6.2 <u>Approvals</u>. Except for filings under the HSR Act or any other applicable Laws relating to antitrust, or as otherwise contemplated by this Agreement, no consent, approval, waiver, authorization or novation is required to be obtained by the Stockholder from, and no notice or filing is required to be given by the Stockholder to or made by the Stockholder with, any Authority or other Person in connection with the execution, delivery and performance by the Stockholder of this Agreement and the exchange of the Company Common Shares.

6.3 <u>Non-Contravention</u>. The execution, delivery and performance by the Stockholder of this Agreement, and the consummation of the transactions contemplated hereby, do not and will not (a) violate any provision of the organizational documents of the Stockholder (if applicable) or (b) violate or result in a breach of or constitute a default under any Law, judgment, injunction, Order, decree or other restriction of any Authority to which the Stockholder or the Company Common Shares, are subject.

6.4 <u>Litigation and Claims</u>. As of the date of this Agreement, there is no civil, criminal or administrative action, suit, demand, claim, hearing, proceeding or disclosed investigation pending or, to the knowledge of the Stockholder, threatened in writing, against the Stockholder, and the Stockholder is not subject to any Order, writ, judgment, award, injunction or decree of any Authority of competent jurisdiction or any arbitrator that would prevent consummation of the transactions contemplated hereby or materially impair the ability of the Stockholder to perform its obligations hereunder.

6.5 Investment Representations.

(a) The Stockholder is an "accredited investor" as such term is defined in Rule 501 of Regulation D ("<u>Reg. D</u>") promulgated under the Act.

(b) Except as otherwise set forth in <u>Article V</u>, Purchaser has not and is not making any representations or warranties to the Stockholder or providing any advice or information to the Stockholder. The Stockholder acknowledges that it has retained its own professional advisors to evaluate the tax and other consequences of an investment in the shares of Purchaser Common Stock.

(c) The Stockholder acknowledges that this offering of shares of Purchaser Common Stock has not been reviewed by the SEC because this is intended to be a non-public offering pursuant to Section 4(a)(2) of the Securities Act and Rule 506 under Reg. D. The shares of Purchaser Common Stock will be received by the Stockholder for investment and not for distribution or resale to others.

(d) The Stockholder understands and consents to the placement of a legend on any certificate or other document evidencing shares of Purchaser Common Stock stating that such shares of Purchaser Common Stock have not been registered under the Securities Act and setting forth or referring to the restrictions on transferability and sale thereof. Each certificate evidencing shares of Purchaser Common Stock shall bear the legends set forth below, or legends substantially equivalent thereto, together with any other legends that may be required by federal or state securities laws at the time of the issuance of the shares of Purchaser Common Stock:

THE SHARES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "<u>ACT</u>"), AND MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED, PLEDGED OR HYPOTHECATED UNLESS AND UNTIL (I) REGISTERED UNDER THE ACT OR (II) THE ISSUER OF THE SHARES (THE "<u>ISSUER</u>") HAS RECEIVED AN OPINION OF COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER THAT SUCH OFFER, SALE OR TRANSFER, PLEDGE OR HYPOTHECATION IS IN COMPLIANCE WITH THE ACT.

ARTICLE VII COVENANTS OF THE PARTIES PENDING CLOSING

7.1 <u>Conduct of the Business</u>. From the date hereof through the earlier of the termination of this Agreement in accordance with <u>Article XII</u> or the Closing Date (the "<u>Interim Period</u>"), each of Purchaser and the Company shall conduct their respective businesses only in the ordinary course, (including the payment of accounts payable and the collection of accounts receivable), consistent with past practices, and shall use its commercially

reasonable efforts to preserve intact its business relationships with key employees, material suppliers and other material third parties (provide that the Company shall be under no obligation to enter into retention agreements with employees).

(a) Without limiting the generality of the foregoing, during the Interim Period, without the Purchaser's prior written consent (which shall not be unreasonably withheld, conditioned or delayed), except as required by applicable Law or as contemplated by this Agreement, the Company shall not:

 (i) amend, modify or supplement its certificate of incorporation and bylaws or other organizational or governing documents;

 (ii) amend, waive any provision of, terminate prior to its scheduled expiration date, or otherwise compromise in any way, any Material Contract or any other material right or asset of the Company;

(iii) modify, amend or enter into any contract, agreement, lease, license or commitment, which (A) is with respect to Real Property, (B) extends for a term of one year or more or (C) obligates the payment of more than \$500,000 (individually or in the aggregate);

(iv) make any capital expenditures in excess of \$500,000 (individually or in the aggregate) outside of the capital expenditure plan made available to Purchaser on or prior to the date of this Agreement;

(v) sell, lease, license or otherwise dispose of any of the Company Group's material assets or assets covered by any Material Contract except (A) pursuant to existing contracts or commitments disclosed herein, and (B) in the ordinary course consistent with past practice;

 (vi) pay, declare or promise to pay any dividends or other distributions with respect to its capital stock or other equity securities, or pay, declare or promise to pay any other payments to any Stockholder (other than payment of salary, benefits, leases, commissions and similar payments in the ordinary course of business);

(vii) obtain or incur any loan or other Indebtedness in excess of \$500,000 other than
(a) accounts payable, (b) accrued liabilities in the ordinary course of business consistent with past practice, (c) Permitted Liens and (d) Company Convertible Debt;

(viii) suffer or incur any Lien, except for Permitted Liens, on the Company Group's assets;

(ix) delay, accelerate or cancel any material receivables or material Indebtedness owed to the Company Group or write off or make further reserves against the same, other than in the ordinary course of business;

 $(x) \quad \mbox{merge or consolidate with or acquire any other Person or be acquired by any other Person;}$

(xi) permit any material insurance policy protecting any of the Company Group's assets to lapse, other than in connection with such lapse a replacement policy having comparable deductions and providing coverage equal to or greater than the coverage under the lapsed policy for substantially similar premiums or less is in full force and effect;

(xii) adopt any severance, retention or other Plans, amend any of its Plans (except as may be required by such Plan or by applicable law) or fail to continue to make timely contributions thereto in accordance with the terms thereof, or pay any bonus to any Company Employee;

(xiii) institute, settle or agree to settle any litigation, action, proceeding or investigation before any court or governmental body in each case in excess of \$500,000 (exclusive of any amounts covered by insurance) or that imposes injunctive or other non-monetary relief on the Company;

(xiv) make any change in its accounting principles or methods;

(xv) change the place of business or jurisdiction of organization of the Company;

(xvi) extend any material loans to employees other than travel or other expense advances to employees in the ordinary course of business;

(xvii) issue, redeem or repurchase any capital stock, membership interests or other securities, or issue any securities exchangeable for or convertible into any shares of its capital stock or other securities except (a) subject to (viii) above, in connection with obtaining any loans or grants in the ordinary course of business, (b) as the result of the exercise or conversion of any Company Options outstanding as of the date hereof, (c) any equity issuance pursuant to <u>Schedule 1.23</u>, or (d) grant of Company Options in the ordinary course of business;

(xviii) make or change any material Tax election, in each case other than to comply with applicable GAAP; or

(xix) agree to do any of the foregoing.

(b) Without limiting the generality of the foregoing, during the Interim Period, without the Company's prior written consent (which shall not be unreasonably withheld, conditioned or delayed), except as required by applicable Law or as contemplated by this Agreement, Purchaser shall not:

(i) amend, modify or supplement its certificate of incorporation and bylaws or other organizational or governing documents;

(ii) amend, waive any provision or terminate prior to its scheduled expiration date, or otherwise compromise in any way, any material Contract or any other material right or asset of the Purchaser or any other Contract which (A) is with respect to Real Property, (B) extends for a term of one year or more or (C) obligates the payment of more than \$500,000 (individually or in the aggregate);

(iii) make any capital expenditures;

(iv) pay, declare or promise to pay any dividends or other distributions with respect to its capital stock or other equity securities;

(v) obtain or incur any loan or other Indebtedness other than (a) accounts payable,
(b) accrued liabilities in the ordinary course of business consistent with past practice, (c) Permitted Liens and (d) loans of up to \$400,000 (in the aggregate) for Working Capital Loans, as described and defined in the Prospectus;

(vi) suffer or incur any Lien, except for Permitted Liens, on Purchaser's assets;

(vii) merge or consolidate with or acquire (by purchasing a substantial portion of the assets of or equity in, or by any other manner) any other Person or be acquired by any other Person;

(viii) permit any material insurance policy protecting any of the Purchaser's assets to lapse, other than in connection with such lapse a replacement policy having comparable deductions and providing coverage equal to or greater than the coverage under the lapsed policy for substantially similar premiums or less is in full force and effect;

 (ix) institute, settle or agree to settle any litigation, action, proceeding or investigation before any court or governmental body in each case in excess of \$500,000 (exclusive of any amounts covered by insurance) or that imposes injunctive or other non-monetary relief on the Purchaser;

- (x) make any change in its accounting principles or methods;
- (xi) change its place of business or jurisdiction;

(xii) issue, redeem or repurchase any capital stock, or other securities, or issue any securities exchangeable for or convertible into any shares of its capital stock or other securities;

(xiii) make or change any material Tax election, in each case other than to comply with applicable GAAP;

 $({\rm xiv})$ $\,$ amend, waive or otherwise change the Trust Agreement in any manner adverse to Purchaser; or

(xv) agree to do any of the foregoing.

7.2 <u>Annual and Interim Financial Statements</u> Each of the Company Group and the Purchaser shall provide the other party with its consolidated balance sheet, income statements, and other financial information necessary for inclusion in any Purchaser SEC Document and/or the Proxy Statement no later than forty-five (45) calendar days following the end of its three-month fiscal quarterly period, all prepared in accordance with U.S. GAAP, except as otherwise indicated in such statements and subject to year-end audit adjustments and omission of notes. If either party does not deliver the requisite financial information for each three-month quarterly period as required by this <u>Section 7.2</u>, the other party shall have the right to terminate this Agreement in accordance with <u>Section 12.2</u> hereof.

7.3 <u>Trust Account</u>. Purchaser covenants that it shall make appropriate arrangements to cause the funds in the Trust Account to be disbursed in accordance with the Trust Agreement and for the payment of (a) all amounts payable to stockholders holding Purchaser Units or shares of Purchaser Common Stock who shall have validly redeemed their Purchaser Units or shares of Purchaser Common Stock upon acceptance by the Purchaser of such Purchaser Units or shares of Purchaser Common Stock (b) the expenses to the third parties to which they are owed, (c) the Deferred Underwriting Amount to the underwriter in the IPO and (d) the remaining monies in the Trust Account to Purchaser.

7.4 <u>Purchaser Public Filings</u>. During the Interim Period, Purchaser will keep current and timely file all of its public filings with the SEC and otherwise comply in all material respects with applicable securities Laws and shall use its commercially reasonable efforts to maintain the listing of the Purchaser Securities on Nasdaq. Purchaser shall (a) to the extent required by the rules and regulations of Nasdaq, prepare and submit to Nasdaq a notification form for the listing of the Transaction Securities, and to cause such shares to be approved for listing (subject to official notice of issuance), and (b) to the extent required under the rules and regulations of <u>Nasdaq</u>, to file an initial listing application for the Purchaser Common Stock on<u>Nasdaq</u> (the <u>"Nasdaq Listing Application</u>") and to cause such Nasdaq Listing Application to be approved prior to the Closing.

7.5 Form 8-K; Press Releases

(a) As promptly as practicable after execution of this Agreement, Purchaser will prepare and file a Current Report on Form 8-K pursuant to the Exchange Act to report the execution of this Agreement, which the Company may review and comment upon prior to filing. Promptly after the execution of this Agreement, Purchaser and the Company shall also issue a joint press release announcing the execution of this Agreement.

(b) Prior to the Closing, the Purchaser and the Company shall prepare a mutually agreeable press release announcing the consummation of the Share Exchange (the "<u>Closing Press Release</u>"). Concurrently with the Closing, the Purchaser shall distribute the Closing Press Release.

7.6 Section 16 of the Exchange Act. Prior to the Closing, the board of directors of the Purchaser, or an appropriate committee thereof, shall adopt a resolution consistent with the interpretive guidance of the SEC relating to Rule 16b-3(d) under the Exchange Act, such that the acquisition of Purchaser Common Stock pursuant to this Agreement by any officer or director of the Company who is expected to become a "covered person" of the Purchaser for purposes of Section 16 of the Exchange Act and the rules and regulations thereunder ("Section 16") shall be exempt acquisitions for purposes of Section 16.

7.7 <u>No Solicitation</u>. During the Interim Period, none of the Purchaser Board or the Purchaser, on the one hand, or the Company or any of its Subsidiaries, or the Company's Board of Directors (the "<u>Company</u><u>Board</u>"), on the other hand, will, nor will they authorize or permit any of their respective Representatives to, directly or indirectly, (i) solicit, initiate or knowingly encourage, support, facilitate or induce the making, submission or public announcement of any inquiry, indication of interest, proposal or offer that constitutes, or could reasonably be expected to lead to, an Acquisition Proposal, (ii) enter into, participate in, maintain or continue any communications (except solely to provide written notice as to the existence of these provisions) or negotiations

regarding, or deliver or make available to any Person any non-public information with respect to, or knowingly take any other action regarding, any inquiry, indication of interest, proposal or offer that constitutes, or could reasonably be expected to lead to, an Acquisition Proposal, (iii) agree to, accept, approve, endorse or recommend (or publicly propose or announce any intention or desire to agree to, accept, approve, endorse or recommend) any Acquisition Proposal, (iv) in the case of the Purchaser, fail to include the Purchaser Board Recommendation in (or remove from) the Proxy Statement, (v) enter into any agreement in principle, letter of intent, term sheet or any other agreement, understanding or contract (whether binding or not) contemplating or otherwise relating to any Acquisition Proposal, (vi) submit any Acquisition Proposal to the vote of any securityholders of the Purchaser, the Company or any of its Subsidiaries (as applicable), (vii) withhold, withdraw, qualify, amend or modify (or publicly propose or announce any intention or desire to withhold, withdraw, gualify, amend or modify), in a manner adverse to the other party, the approval of such party's board of directors of this Agreement and/or any of the transactions contemplated hereby, or, in the case of the Purchaser, the Purchaser Board Recommendation, (viii) approve any transaction, or any third party becoming an interested stockholder, under applicable Laws or (ix) resolve, propose or agree to do any of the foregoing (any of the actions described in clauses (iii)-(ix), a "Change of Recommendation"). Both the Company and the Purchaser shall, and shall cause their respective Representatives to, immediately cease any and all existing activities, discussions or negotiations with any Persons conducted prior to or on the date of this Agreement with respect to any Acquisition Proposal. Each of the Purchaser and the Company shall notify the other party within 48 hours of the receipt by such party or any of its Representatives of any bona fide inquiries, proposals or offers, requests for information or requests for discussions or negotiations that constitutes or could reasonably be expected to constitute an Acquisition Proposal, including a correct and complete copy thereof if in writing or a written summary of the material terms thereof if it is not in writing, and the identity of the party making such inquiry, proposal, offer or request for information. If a Representative of either party, whether in his, her or its capacity as such or in any other capacity, takes any action that such party is obligated pursuant to this Section 7.7 not to take, then such party shall be deemed for all purposes of this Agreement to have breached this Section <u>7.7</u>.

7.8 <u>Transfer Taxes</u>. All Transfer Taxes that may become due and payable as a result of the transactions contemplated in this Agreement, levied on the Purchaser or the Company, shall be paid fifty percent (50%) by the Purchaser and fifty percent (50%) by the Stockholders. The Purchaser will file all necessary Tax Returns and other documentation with respect to all such Transfer Taxes and any expenses incurred in connection with the filing of such Tax Returns or other documentation shall be paid (50%) by the Purchaser and fifty percent (50%) by the Stockholders. If required by applicable Law, the Stockholders will join the Purchaser in the execution of any such Tax Return and other documentation.

7.9 Regulatory Compliance.

(a) As promptly as practicable after execution and delivery of this Agreement, but in no event later than ten (10) Business Days thereafter, Purchaser and the Company shall execute and file, or join in the execution and filing of, any application, notification (including the provision of any required information in connection therewith) or other document that may be required under the HSR Act or any other applicable Laws relating to antitrust, in order to obtain the authorization, approval or consent of any Authority, or expiration or termination of the waiting periods under such Laws, that may be reasonably required, in connection with the consummation of the transactions contemplated by this Agreement.

(b) Purchaser and the Company shall cooperate with each other and shall each use its respective reasonable best efforts to obtain promptly all authorizations, approvals, consents, expirations or terminations of applicable waiting periods necessary or advisable for the consummation of this Agreement. Each of the parties will furnish to the other parties such information and assistance as may be reasonably requested in connection with the foregoing, including by (i) timely furnishing to each other all information required to be included in such documents; (ii) to the extent not prohibited by such Authority and practicable, promptly providing each other with copies of all written communications to or from any Authority; (iii) responding promptly to and complying with any reasonable request for additional information or documents; (iv) to the extent practicable, permitting the Company or Purchaser (as the case may be) to review in advance and consider incorporating the other party's reasonable comments in any communication given by it to any Authority; and (v) to the extent there are any meetings or substantive calls with any Authority, the Company and Purchaser shall each permit the other to attend such meetings or calls unless prohibited by such Authority.

(c) All fees and payments required to be made to any Authority in accordance with this <u>Section 7.9</u>, whether under the HSR Act or any other applicable Laws relating to antitrust, shall be paid fifty percent (50%) by the Purchaser and fifty percent (50%) by the Stockholders.

ARTICLE VIII

COVENANTS OF THE COMPANY AND PURCHASER

The Company and the Purchaser acknowledge and agree that:

8.1 Access to Information. From the date hereof until and including the Closing Date, each of the Company and the Purchaser shall (a) continue to give the other party, its legal counsel and other representatives reasonable access during normal business hours to the offices, properties and Books and Records of such party, (b) furnish to the other party, its legal counsel and other representatives such information relating to the business of the Company Group and the Purchaser as such Persons may reasonably request and (c) cause its employees, legal counsel, accountants and representatives to cooperate with the other party in its investigation of such party's business; provided no investigation pursuant to this Section 8.1 (or any investigation prior to the date hereof) shall affect any representation or warranty given by the Company or the Purchaser and, provided, further, that any investigation pursuant to this Section 8.1 shall be conducted in such manner as not to interfere unreasonably with the conduct of the business of the Company or the Purchaser during normal business hours under the supervision of appropriate personnel of the Company. Nothing herein shall require a party to disclose or provide access to any information that could be detrimental to such party's business or operations or if such disclosure could, in its reasonable discretion (i) jeopardize any attorney-client or other legal privilege (so long as such party has reasonably cooperated with the other party to permit such inspection of or to disclose such information on a basis that does not waive such privilege with respect thereto), or (ii) contravene any applicable Law or binding agreement entered into prior to the date of this Agreement (including any confidentiality agreement to which the Company or Purchaser is a party).

8.2 <u>Exclusivity</u>. During the Interim Period, neither the Company nor the Purchaser shall enter into a financing transaction or any agreement relating to the sale of such party's assets or equity securities, or a merger or change of control agreement with respect to such party or its assets, without the prior written consent of the other party, other than any Company Permitted Financing and, in the case of the Company, licensing in the ordinary course of business.

8.3 <u>Notices of Certain Events</u>. During the Interim Period, each of the Company and the Purchaser shall promptly notify the other party in writing of:

(a) any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement or that the transactions contemplated by this Agreement might give rise to any Action or other rights by or on behalf of such Person or result in the loss of any rights or privileges of the Company (or the Purchaser, post-Closing) to any such Person or create any Lien (other than Permitted Liens) on any Company Common Shares, Purchaser Common Stock or any of the Company Group's or the Purchaser's assets;

(b) any notice or other communication from any Authority in connection with the transactions contemplated by this Agreement or the Additional Agreements;

(c) any Actions commenced or threatened in writing against, involving the Company, any Stockholder, the Purchaser, the Company Common Shares, the Purchaser Common Stock or the Company's or Purchaser's assets or the Business or that relate to the consummation of the transactions contemplated by this Agreement or the Additional Agreements;

(d) the occurrence of any fact or circumstance which constitutes or results, or is reasonably be expected to constitute or result, in a Material Adverse Effect or Purchaser Material Adverse Effect; and

(e) the occurrence of any fact or circumstance that renders any representation or warranty under <u>Article IV</u> by the Company or <u>Article V</u> by the Purchaser untrue or inaccurate such that the conditions set forth in <u>Section 10.2(b)</u> or <u>10.3(a)</u>, respectively, would not be satisfied.

No such notice shall constitute an acknowledgement or admission by the party providing the notice regarding whether or not any of the conditions to the Closing have been satisfied or in determining whether or not any of the representations, warranties or covenants contained in this Agreement have been breached.

ARTICLE IX COVENANTS OF ALL PARTIES HERETO

The parties hereto covenant and agree that:

9.1 Proxy Statement.

(a) As soon as reasonably practicable (but no later than three (3) Business Days) after the date hereof, the Purchaser and the Company shall prepare and the Purchaser shall file a preliminary proxy statement (as amended, the "Proxy Statement") with the SEC for purposes of (a) approval of the Business Combination and the other transactions contemplated hereby, (b) approval of the amendment of the Amended and Restated Certificate of Incorporation of the Purchaser in the form set forth in Exhibit D attached hereto (the "Purchaser Charter Amendment"), (c) approval of the issuance of more than 20% of the issued and outstanding shares of Purchaser Common Stock pursuant to this Agreement, pursuant to Nasdaq requirements, (d) approval of the 2019 HSAC Equity Incentive Plan and (e) approval of any adjournment of the Purchaser Stockholder Meeting in the event the Purchaser does not receive the requisite vote to approve the matter set forth in clause (a) above (the approvals described in foregoing clauses (a) through (e), collectively, the "Purchaser Stockholder Matters"). The Proxy Statement and any other SEC filings shall be in a form mutually agreed by the Purchaser, the Company and the Stockholders' Representative. As promptly as reasonably practicable (but in any event within five (5) Business Days) following the later of (i) receipt and resolution of SEC comments with respect to the Proxy Statement and (ii) the expiration of the 10-day waiting period provided in Rule 14a-6(a) promulgated under the Exchange Act, the Purchaser and the Company shall cooperate to file the definitive Proxy Statement and cause the definitive Proxy Statement to be mailed to the Purchaser's stockholders. The Purchaser will cause all documents that it is responsible for filing with the SEC or other regulatory authorities in connection with the Share Exchange to (A) comply as to form with all applicable SEC requirements and (B) otherwise comply in all material respects with all applicable Law.

The Purchaser will notify the Company promptly of the receipt of any comments (b) (written or oral) from the SEC or its staff (or of notice of the SEC's intent to review the Proxy Statement) and of any request by the SEC or its staff or any other official of any Authority for amendments or supplements to the Proxy Statement or any other filing or for additional/supplemental information, and will supply the Company with copies of all correspondence between the Purchaser or any of its Representatives, on the one hand, and the SEC, or its staff or any other official of any Authority, on the other hand, with respect to the Proxy Statement or such other filing. The Purchaser shall permit the Company and its outside counsel to participate in all material discussions and meetings with the SEC and its staff relating to the Proxy Statement, this Agreement or the transactions contemplated thereby. The Purchaser shall (i) consult with the Company prior to responding to any comments or inquiries by the SEC or any other Authority with respect to any filings related to the Share Exchange, (ii) provide the Company and its Representatives with reasonable opportunity to review and comment on any such written response in advance and consider in good faith the incorporation of any changes reasonably proposed by the Company and (iii) promptly inform the Company whenever any event occurs that requires the filing of an amendment or supplement to the Proxy Statement or any other filing, and the Purchaser shall provide the Company and its Representatives with a reasonable opportunity to review and comment on any such amendment or supplement in advance and consider in good faith the incorporation of any changes reasonably proposed by the Company and its Representatives, and shall cooperate in filing with the SEC or its staff or any other official of any Authority, and/or mailing to the Purchaser's stockholders, such amendment or supplement. The Company shall promptly inform the Purchaser whenever the Company discovers any event relating to Purchaser or any of its Affiliates, officers or directors that is required to be set forth in an amendment or supplement to the Proxy Statement.

(c) In connection with any filing the Purchaser makes with the SEC that requires information about the transactions contemplated by this Agreement to be included, the Company will, and will use reasonable best efforts to cause its Representatives, in connection with the disclosure included in any such filing or the responses provided to the SEC in connection with the SEC's comments to a filing, to use reasonable best efforts to (i) cooperate with the Purchaser, (ii) respond to questions about the Company required in any filing or requested by the SEC in a timely fashion, and (iii) promptly provide any information requested by the Purchaser or the Purchaser's Representatives in connection with any filing with the SEC.

9.2 Purchaser Stockholder Meeting; Board Recommendation.

(a) Prior to the filing of a definitive Proxy Statement with the SEC, the Purchaser shall establish a record date for, duly call, give notice of, convene and hold a meeting of the Purchaser's stockholders (including any adjournment or postponement thereof, the "Purchaser Stockholder Meeting") to be held as promptly as reasonably practicable following the filing of the definitive Proxy Statement for the sole purpose of obtaining approval of the Purchaser Stockholder Matters (the "Purchaser Stockholder Approval") (including any adjournment of such meeting for the purpose of soliciting additional proxies in favor of the adoption of this Agreement) and such other matter as may be agreed by the Company. The Purchaser will use its reasonable best efforts to solicit from its stockholders proxies in favor of the adoption of this Agreement and will take all other reasonable action necessary or advisable to obtain such proxies and the Purchaser Stockholder Approval and to secure the vote or consent of its stockholders required by and in compliance with all applicable Law and its certificate of incorporation and bylaws. The Purchaser (i) shall consult with the Company regarding the record date and the date of the Purchaser Stockholder Meeting and (ii) shall not adjourn or postpone the Purchaser Stockholder Meeting without the prior written consent of Company; provided that the Purchaser may adjourn or postpone the Purchaser Stockholder Meeting (A) to the extent necessary to ensure that any supplement or amendment to the Proxy Statement that the Purchaser reasonably determines (following consultation with Company, except with respect to any Acquisition Proposal) is necessary to comply with applicable Laws, is provided to the Purchaser's stockholders in advance of a vote on the adoption of this Agreement, (B) if, as of the time that the Purchaser Stockholder Meeting is originally scheduled, there are insufficient shares of Purchaser Common Stock represented at such meeting (either in person or by proxy) to constitute a quorum necessary to conduct the business of the Purchaser Stockholder Meeting, or (C) if, as of the time that the Purchaser Stockholder Meeting is originally scheduled, adjournment or postponement of the Purchaser Stockholder Meeting is necessary to enable the Purchaser to solicit additional proxies required to obtain the Purchaser Stockholder Approval; provide that the Purchaser may postpone or adjourn on one occasion so long as the date of the Purchaser Stockholder Meeting is not postponed or adjourned more than an aggregate of 15 consecutive calendar days in connection with such postponement or adjournment.

(b) (i) the Proxy Statement shall include a statement to the effect that the Purchaser Board has unanimously recommended that the Purchaser's stockholders vote in favor of the Purchaser Stockholder Matters at the Purchaser Stockholder Meeting and (ii) neither the Purchaser Board nor any committee thereof shall withhold, withdraw, qualify, amend or modify, or publicly propose or resolve to withhold, withdraw, qualify, amend or modify, the Purchaser Board Recommendation.

9.3 <u>Further Assurances</u>. Subject to the terms and conditions of this Agreement, each party shall use its commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary or desirable under applicable Laws, and in the case of the Company, as reasonably requested by Purchaser, to consummate and implement expeditiously each of the transactions contemplated by this Agreement. The parties hereto shall execute and deliver, or cause to be executed and delivered, such other documents, certificates, agreements and other writings and take such other actions as may be necessary or desirable in order to consummate or implement expeditiously each of the transactions contemplated by this Agreement.

9.4 <u>Compliance with SPAC Agreements</u>. The Company and the Purchaser shall comply with each of the agreements entered into in connection with the IPO.

9.5 <u>Confidentiality</u>. Except as necessary to complete the Proxy Statement, the Company and the Stockholders, on the one hand, and the Purchaser, on the other hand, shall hold and shall cause their respective representatives to hold in strict confidence, unless compelled to disclose by judicial or administrative process or by other requirements of Law, all documents and information concerning the other party furnished to it by such other party or its representatives in connection with the transactions contemplated by this Agreement (except to the extent that such information can be shown to have been (a) previously known by the party to which it was furnished, (b) in the public domain through no fault of such party or (c) later lawfully acquired from other sources on a non-confidential basis, which source of any obligation of confidentiality to the other party), and each party shall not release or disclose such information to any other person, except its representatives in connection with this Agreement. In the event that any party believes that it is required to disclose any such confidential information pursuant to applicable Laws, to the extent permitted by applicable Law, such party shall give timely written notice to the other party (which, for clarity, shall in the case of the Company be to the Purchaser and in the case of the

Purchaser be to the Company) so that such party may have an opportunity to obtain a protective order or other appropriate relief, and such party shall only disclose the minimum amount of such confidential information so required to be disclosed. The parties acknowledge that some previously confidential information will be required by applicable Law to be disclosed in the Proxy Statement.

9.6 <u>Registration Statement</u>. Within 30 days from the Closing Date, the Purchaser shall file a resale registration statement on Form S-3 (or to the extent not available, on Form S-1) with respect to (a) the Closing Payment Shares; (b) the Purchaser Warrants (including any shares of Purchaser Common Stock issuable upon exercise of such Purchaser Warrants) and (c) the Purchaser Common Stock held by Sponsor.

9.7 Indemnification; Insurance.

(a) From and after the Closing Date and for a period of six years thereafter, the Purchaser shall fulfill and honor in all respects the obligations pursuant to any indemnification agreements between the Purchaser or the Company Group, on the one hand, and any current or former directors, officers and employees, as the case may be, of the Purchaser, the Company Group, on the other hand, in effect immediately prior to the Closing Date, and any indemnification provisions under the Purchaser Charter Documents, Company Charter Documents or the comparable charter or organizational documents of any of its Subsidiaries as in effect on the date hereof, in each case to the maximum extent permitted by Law, and shall not amend, repeal or otherwise modify any such provision in any manner that would adversely affect the rights of such indemnitee thereunder for any acts or omissions occurring prior to the Closing Date.

(b) Prior to the Closing Date, the Company shall enter into a directors' and officers' liability insurance policy covering the current and former directors, officers and employees, as the case may be, of the Company (the "<u>Insured Parties</u>") on customary terms that are no less favorable to the Insured Parties than those of any present directors' and officers' liability insurance policy maintained by the Company covering the Insured Parties (such policy, a "<u>Company D&O Policy</u>"), for a period of seven years after the Closing Date. All costs and expenses related to the Company D&O Policy, including the insurance premiums, shall be paid by the Company.

(c) Notwithstanding anything contained in this Agreement to the contrary, this <u>Section</u> <u>9.7</u> shall survive the consummation of the transactions contemplated by this Agreement and shall be binding, jointly and severally, on the Purchaser, the Company Group and all successors and assignees of the Purchaser and the Company Group. In the event that the Purchaser or any of its respective successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all its properties and assets to any Person, the Purchaser shall cause proper provisions to be made so that the successors and assigns of the Purchaser assume the obligations set forth in this <u>Section 9.7</u>.

(d) The obligations of the Purchaser and the Company under this <u>Section 9.7</u> shall not be terminated or modified in such a manner as to adversely affect any indemnitee and/or Insured Party to whom this <u>Section 9.7</u> applies without the express written consent of such affected indemnitee and Insured Party. It is expressly agreed that the indemnitees and/or Insured Parties to whom this <u>Section 9.7</u> applies shall be third-party beneficiaries of this <u>Section 9.7</u>.

(e) The Purchaser shall assume, be jointly and severally liable for, and shall cause its Subsidiaries to honor, in accordance with their respective terms, each of the covenants contained herein without limit as to time. The Purchaser shall pay all reasonable expenses, including reasonable attorneys' fees, that may be incurred by any indemnitee and/or Insured Party in enforcing the indemnity and other obligations provided hereunder or other applicable indemnification obligation referenced to herein. The rights of each indemnitee and/or Insured Party hereunder shall be in addition to, and not in limitation of, any other rights such Person may have under the Company Charter Documents or the comparable charter or organizational documents of any member of the Company Group, or any other indemnification arrangement or otherwise.

(f) On the Closing Date, the Purchaser shall enter into customary indemnification agreements reasonably satisfactory to the Company with the individuals set forth on <u>Schedule 9.7(f)</u> (the "<u>Indemnification Agreements</u>"), which indemnification agreements shall continue to be effective following the Closing.

9.8 <u>R&W Insurance Policy</u>. The parties acknowledge that the Purchaser has entered into the binder agreement with respect to the R&W Insurance Policy as of the date of this Agreement. Purchaser shall not amend, modify or otherwise change, terminate or waive any subrogation, or any other provision of the R&W Insurance Policy without the written consent of the Company. All costs and expenses related to the R&W Insurance Policy, including the total premium, underwriting costs, brokerage commissions and any other fees and expenses of such policy, shall be paid ninety percent (90%) by the Company and ten percent (10%) by the Purchaser.

ARTICLE X CONDITIONS TO CLOSING

10.1 <u>Condition to the Obligations of the Parties</u>. The obligations of all of the parties to consummate the Closing are subject to the satisfaction of all the following conditions:

(a) No provisions of any applicable Law, and no Order shall restrain or prohibit or impose any condition on the consummation of the Closing.

(b) There shall not be any Action brought by any governmental Authority to enjoin or otherwise restrict the consummation of the Closing.

(c) Any waiting period under the HSR Act relating to the transactions contemplated by this Agreement shall have expired or been terminated.

(d) The Transaction Expenses of each of the Purchaser and the Company shall have been paid by wire transfer of immediately available funds to the applicable third party in accordance with instructions provided by such third party.

10.2 <u>Conditions to Obligations of Purchaser</u>. The obligation of the Purchaser to consummate the Closing is subject to the satisfaction, or the waiver at the Purchaser's sole and absolute discretion, of all the following further conditions:

(a) The Company shall have performed in all material respects its obligations hereunder required to be performed by it pursuant to this Agreement at or prior to the Closing Date.

(b) All of the representations and warranties of the Company contained in this Agreement, disregarding all qualifications and exceptions contained therein relating to materiality or Material Adverse Effect, shall: (i) be true, correct and complete at and as of the date of this Agreement, or, (ii) if otherwise specified, when made or when deemed to have been made, and (iii) be true, correct and complete as of the Closing Date, except in the case of (i), (ii) and (iii) for any inaccuracies in such representations and warranties which would not in the aggregate reasonably be expected to have a Material Adverse Effect.

(c) There shall have been no event, change or occurrence that is continuing, which, individually or together with any other event, change or occurrence, would reasonably be expected to have a Material Adverse Effect.

(d) The Purchaser shall have received a certificate signed by the Chief Executive Officer of the Company to the effect set forth in clauses (a) through (c) of this <u>Section 10.2</u> (the "<u>Company Certificate</u>").

(e) The Purchaser shall have received Schedules updated as of the Closing Date, which shall be deemed to update the disclosures set forth in the Disclosures Schedule provided to the Purchaser on or prior to the date of this Agreement for all purposes of this Agreement, including for the purpose of determining the satisfaction of the conditions set forth in <u>Section 10.2</u>.

(f) The R&W Insurance Policy shall continue to be effective as of the Closing Date pursuant to its terms.

10.3 <u>Conditions to Obligations of the Company</u>. The obligations of the Company to consummate the Closing is subject to the satisfaction, or the waiver at the Company's discretion, of all of the following further conditions:

(a) (i) The Purchaser shall have performed in all material respects all of its obligations hereunder required to be performed by it pursuant to this Agreement at or prior to the Closing Date, (ii) the

representations and warranties of the Purchaser contained in this Agreement, and in any certificate or other writing delivered by the Purchaser pursuant hereto, disregarding all qualifications and expectations contained therein relating to materiality shall (1) be true, correct and complete in all respects at and as of the date of this Agreement, (2) if otherwise specified, when made or when deemed to have been made, and (3) be true, correct and complete as of the Closing Date, and (iii) the Company shall have received a certificate signed by an authorized officer of the Purchaser to the foregoing effect.

(b) The Sponsor's 10,000,000 Purchaser Private Warrants shall have been unconditionally and irrevocably forfeited and cancelled.

(c) The Purchaser shall have met the Cash Closing Requirement.

(d) The Purchaser shall have received a letter from Nasdaq indicating that the combined company and the Transaction Securities (subject to official notice of issuance) have been approved for listing.

(e) The Stockholder Representative designees (in accordance with <u>Section 2.4</u>) shall have been appointed to the Purchaser Board, effective as of the Closing, and the Purchaser nominee, if nominated, shall have been nominated in compliance with <u>Section 2.4</u>.

(f) Purchaser Charter Documents shall have been amended and restated in the forms attached hereto as <u>Exhibit D</u>.

(g) The Purchaser Stockholder Approval shall have been obtained in accordance with the provisions of the Purchaser Charter Documents and applicable Laws, including the Delaware General Corporation Law, Nasdaq rules and the Exchange Act.

ARTICLE XI DISPUTE RESOLUTION

11.1 Arbitration

(a) The parties shall promptly submit any dispute, claim, or controversy arising out of or relating to this Agreement or any Additional Agreement (including with respect to the meaning, effect, validity, termination, interpretation, performance, or enforcement of this Agreement or any Additional Agreement) or any alleged breach thereof (including any action in tort, contract, equity, or otherwise), to binding arbitration before one arbitrator (the "Arbitrator"). Binding arbitration shall be the sole means of resolving any dispute, claim, or controversy arising out of or relating to this Agreement or any Additional Agreement (including with respect to the meaning, effect, validity, termination, interpretation, performance or enforcement of this Agreement or any Additional Agreement) or any alleged breach thereof (including any claim in tort, contract, equity, or otherwise).

(b) If the parties cannot agree upon the Arbitrator, the Arbitrator shall be selected by the New York, New York chapter head of the American Arbitration Association upon the written request of either side. The Arbitrator shall be selected within thirty (30) days of the written request of any party.

(c) The laws of the State of New York shall apply to any arbitration hereunder. In any arbitration hereunder, this Agreement shall be governed by the laws of the State of New York applicable to a contract negotiated, signed, and wholly to be performed in the State of New York, which laws the Arbitrator shall apply in rendering his decision. The Arbitrator shall issue a written decision, setting forth findings of fact and conclusions of law, within sixty (60) days after he shall have been selected. The Arbitrator shall have no authority to award punitive or other exemplary damages.

(d) The arbitration shall be held in New York, New York in accordance with and under the then-current provisions of the rules of the American Arbitration Association, except as otherwise provided herein.

(c) On application to the Arbitrator, any party shall have rights to discovery to the same extent as would be provided under the Federal Rules of Civil Procedure, and the Federal Rules of Evidence shall apply to any arbitration under this Agreement; provided, however, that the Arbitrator shall limit any discovery or evidence such that his decision shall be rendered within the period referred to in <u>Section 11.1(c)</u>.

(f) The Arbitrator may, at his discretion and at the expense of the party who will bear the cost of the arbitration, employ experts to assist him in his determinations.

(g) The costs of the arbitration proceeding and any proceeding in court to confirm any arbitration award or to obtain relief as provided in <u>Section 11.1(h)</u>, as applicable (including actual attorneys' fees and costs), shall be borne by the unsuccessful party and shall be awarded as part of the Arbitrator's decision, unless the Arbitrator shall otherwise allocate such costs in such decision. The determination of the Arbitrator shall be final and binding upon the parties and not subject to appeal.

(h) Any judgment upon any award rendered by the Arbitrator may be entered in and enforced by any court of competent jurisdiction. The parties expressly consent to the non-exclusive jurisdiction of the courts (Federal and state) in the State of New York to enforce any award of the Arbitrator or to render any provisional, temporary, or injunctive relief in connection with or in aid of the arbitration. The parties expressly consent to the personal and subject matter jurisdiction of the Arbitrator to arbitrate any and all matters to be submitted to arbitration hereunder. None of the parties hereto shall challenge any arbitration hereunder on the grounds that any party necessary to such arbitration (including the parties hereto) shall have been absent from such arbitration for any reason, including that such party shall have been the subject of any bankruptcy, reorganization, or insolvency proceeding.

(i) The parties shall indemnify the Arbitrator and any experts employed by the Arbitrator and hold them harmless from and against any claim or demand arising out of any arbitration under this Agreement or any agreement contemplated hereby, unless resulting from the gross negligence or willful misconduct of the person indemnified.

(j) This arbitration section shall survive the termination of this Agreement and any agreement (including the Additional Agreements) contemplated hereby.

11.2 Waiver of Jury Trial; Exemplary Damages

(a) THE PARTIES TO THIS AGREEMENT HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVE ANY RIGHT EACH SUCH PARTY MAY HAVE TO TRIAL BY JURY IN ANY ACTION OF ANY KIND OR NATURE, IN ANY COURT IN WHICH AN ACTION MAY BE COMMENCED, ARISING OUT OF OR IN CONNECTION WITH THIS AGREEMENT OR ANY ADDITIONAL AGREEMENT, OR BY REASON OF ANY OTHER CAUSE OR DISPUTE WHATSOEVER BETWEEN OR AMONG ANY OF THE PARTIES TO THIS AGREEMENT OF ANY KIND OR NATURE. NO PARTY SHALL BE AWARDED PUNITIVE OR OTHER EXEMPLARY DAMAGES RESPECTING ANY DISPUTE ARISING UNDER THIS AGREEMENT OR ANY ADDITIONAL AGREEMENT.

(b) Each of the parties to this Agreement acknowledge that each has been represented in connection with the signing of this waiver by independent legal counsel selected by the respective party and that such party has discussed the legal consequences and import of this waiver with legal counsel. Each of the parties to this Agreement further acknowledge that each has read and understands the meaning of this waiver and grants this waiver knowingly, voluntarily, without duress and only after consideration of the consequences of this waiver with legal counsel.

ARTICLE XII TERMINATION

12.1 Termination Without Default.

(a) In the event that the Closing of the transactions contemplated hereunder has not occurred by January 31, 2020 (the "<u>Outside Closing Date</u>"), the Purchaser or the Company shall have the right, at its sole option, to terminate this Agreement; provided, however, that a party shall not be permitted to terminate this Agreement pursuant to this <u>Section 12.1(a)</u> if the failure of the Closing to occur prior to the Outside Closing Date is attributable to the failure on the part of such party to perform in any material respect any covenant or obligation in this Agreement required to be performed by such party. Such right may be exercised by the Purchaser or the Company, as the case may be, giving written notice to the other at any time after the Outside Closing Date.

(b) In the event that a court of competent jurisdiction or other Authority shall have issued an order, decree or ruling, or shall have taken any other action, having the effect of permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement, which order, decree, ruling or other action shall be final and nonappealable, the Purchaser or the Company shall have the right, at its sole option, to terminate this Agreement; provided, however, that a party shall not be permitted to terminate this Agreement pursuant to this <u>Section 12.1(b)</u> if such action by such court or Authority is attributable to the failure on the part of such party to perform in any material respect any covenant or obligation in this Agreement required to be performed by such party.

(c) In the event that the Purchaser fails to receive the Purchaser Stockholder Approval at the Purchaser Stockholder Meeting (subject to any adjournment or recess of such special meeting pursuant to <u>Section 9.2(a)</u>), the Purchaser or the Company shall have the right, at its sole option, to terminate this Agreement by written notice to the other party; <u>provided</u> that the Purchaser shall not be permitted to terminate this Agreement pursuant to this <u>Section 12.1(c)</u> if the failure to obtain such Purchaser Stockholder Approval is proximately caused by any action or failure to act of the Purchaser that constitutes a breach of this Agreement.

(d) This Agreement may be terminated at any time prior to the Closing by mutual written consent of the Purchaser and the Company.

12.2 Termination Upon Default.

(a) The Purchaser may terminate this Agreement on or prior to the Closing Date by giving written notice to the Company, without prejudice to any rights or obligations the Purchaser may have, if the Company or the Stockholders shall have materially breached any representation, warranty, agreement or covenant contained herein to be performed on or prior to the Closing Date, which would result in a failure of a condition set forth in <u>Section 10.2(a)</u> or <u>Section 10.2(b)</u> to be satisfied and cannot be cured by the earlier of the Outside Closing Date and thirty (30) days following receipt by the Company or the Stockholders' Representative, as the case may be, of a written notice describing in reasonable detail the nature of such breach; provided, however, that, the Purchaser shall not have the right to terminate this Agreement pursuant to this <u>Section 12.2(a)</u> if the Purchaser is then in material breach of any representation, warranty, agreement or covenant hereunder.

(b) The Company may terminate this Agreement on or prior to the Closing Date by giving written notice to the Purchaser, without prejudice to any rights or obligations the Company may have, if the Purchaser shall have materially breached any representation, warranty, agreement or covenant contained herein to be performed on or prior to the Closing Date, which would result in a failure of a condition set forth in <u>Section 10.3(a)</u> to be satisfied and cannot be cured by the earlier of the Outside Closing Date and thirty (30) days following receipt by the Purchaser of a written notice describing in reasonable detail the nature of such breach; provided, however, that, the Company shall not have the right to terminate this Agreement pursuant to this <u>Section 12.2(b)</u> if the Company is then in material breach of any representation, warranty, agreement or covenant hereunder.

12.3 <u>Termination for Triggering Event</u>. Either the Purchaser or the Company may, by written notice to the other party, terminate this Agreement on or prior to the Closing Date, without prejudice to any rights or obligations of such party, if a Triggering Event with respect to the other party shall have occurred.

12.4 <u>Effect of Termination</u>. If this Agreement is terminated pursuant to this <u>Article XII</u>, this Agreement shall become void and of no effect without liability of any party (or any stockholder, director, officer, employee, Affiliate, agent, consultant or representative of such party) to any other party hereto, other than liability of any party for common law fraud. The provisions of <u>Section 9.5</u>, <u>Article XII</u>, this <u>Section 12.4</u> and <u>Article XIII</u> shall survive any termination hereof pursuant to this <u>Article XII</u>.

ARTICLE XIII MISCELLANEOUS

13.1 <u>Notices</u>. Any notice hereunder shall be sent in writing, addressed as specified below, and shall be deemed given: (a) if by hand or recognized courier service, by 4:00PM on a Business Day, addressee's day and time, on the date of delivery, and otherwise on the first Business Day after such delivery; (b) if by fax or email, on the date that transmission is confirmed electronically, if by 4:00PM on a Business Day, addressee's day and time, and otherwise on the first Business Day after the date of such confirmation; or (c) five days after mailing by certified or registered mail, return receipt requested. Notices shall be addressed to the respective parties as follows (excluding

telephone numbers, which are for convenience only), or to such other address as a party shall specify to the others in accordance with these notice provisions:

if to Purchaser or the Company (following the Closing), to:

Health Sciences Acquisitions Corporation 412 West 15th Street, Floor 9 New York, NY 10011 Attention: Roderick Wong, M.D. Phone: (646) 343-9280 Email: rw@rtwfunds.com

with a copy to (which shall not constitute notice):

Loeb & Loeb LLP 345 Park Avenue New York, NY 10154 Attention: Giovanni Caruso, Esq. Phone: (212) 407-4866 Email: gcaruso@loeb.com

if to the Company (prior to the Closing)

Immunovant Sciences Ltd. Suite 1, 3rd Floor 11-12 St. James's Square London SW1Y 4LB United Kingdom Attn: Marianne L. Romeo Phone: (441) 295-1422 Email: Marianne.Romeo@roivant.com

with a copy to (which shall not constitute notice):

Immunovant, Inc. 320 West 37th Street, 3rd Floor New York, NY 10018 Attention: W. Bradford Middlekauff, General Counsel Phone: (212) 847-6204 Email: brad.middlekauff@immunovant.com

with a copy to (which shall not constitute notice):

Cooley LLP 3175 Hanover Street Palo Alto, CA 94304-1130 Attention: Frank Rahmani, Esq., John T. McKenna, Esq. Phone: (650) 843-5000 Email: rahmaniff@cooley.com, jmckenna@cooley.com

with a copy to (which shall not constitute notice):

HanAll BioPharma Co., Ltd. 12F Gyeonggi Bio-Center, 147 Kwangkyo-ro Yeongtong-gu Suwon, Kyeonggi-do Seoul Republic of Korea Attention: Hyeakyung Ahn Facsimile: +82-31-888-6523 E-mail: ahnhk@hanall.co.kr

with a copy to (which shall not constitute notice):

HPI Inc. 1 Church St., Suite 103 Rockville, MD 20850 Attention: Minjae Shin Facsimile: +1-301-738-2016 E-mail: emptymj@hanall.co.kr

with a copy to (which shall not constitute notice):

McDermott, Will & Emery 275 Middlefield Road Menlo Park CA 94025 Attention: Paul DeStefano

if to the Stockholders' Representative:

Roivant Sciences Ltd. Suite 1, 3rd Floor 11-12 St. James's Square London SW1Y 4LB United Kingdom

13.2 Amendments; No Waivers; Remedies.

(a) This Agreement cannot be amended, except by a writing signed by each party, and cannot be terminated orally or by course of conduct. No provision hereof can be waived, except by a writing signed by the party against whom such waiver is to be enforced, and any such waiver shall apply only in the particular instance in which such waiver shall have been given.

(b) Neither any failure or delay in exercising any right or remedy hereunder or in requiring satisfaction of any condition herein nor any course of dealing shall constitute a waiver of or prevent any party from enforcing any right or remedy or from requiring satisfaction of any condition. No notice to or demand on a party waives or otherwise affects any obligation of that party or impairs any right of the party giving such notice or making such demand, including any right to take any action without notice or demand not otherwise required by this Agreement. No exercise of any right or remedy, as appropriate to make the aggrieved party whole with respect to such breach, or subsequent exercise of any right or remedy with respect to any other breach.

(c) Except as otherwise expressly provided herein, no statement herein of any right or remedy shall impair any other right or remedy stated herein or that otherwise may be available.

(d) Notwithstanding anything else contained herein, neither shall any party seek, nor shall any party be liable for, diminution in value, punitive, consequential, special, indirect, or exemplary damages, under any tort, contract, equity, or other legal theory, with respect to any breach (or alleged breach) of this Agreement or any provision hereof or any matter otherwise relating hereto or arising in connection herewith.

13.3 <u>Arm's length bargaining: no presumption against drafter</u>. This Agreement has been negotiated at arm's-length by parties of equal bargaining strength, each represented by counsel or having had but declined the opportunity to be represented by counsel and having participated in the drafting of this Agreement. This Agreement creates no fiduciary or other special relationship between the parties, and no such relationship otherwise exists. No presumption in favor of or against any party in the construction or interpretation of this Agreement or any provision hereof shall be made based upon which Person might have drafted this Agreement or such provision.

13.4 <u>Publicity</u>. The parties agree that neither they nor their agents shall issue any press release or make any other public disclosure concerning the transactions contemplated hereunder without the prior approval of the other party hereto (which, for clarity, shall in the case of the Company be from the Purchaser and in the case of the Purchaser be from the Company). If a party is required to make such a disclosure as required by law, the parties will use their best efforts to cause a mutually agreeable release or public disclosure to be issued. Notwithstanding

the foregoing: (a) each party may, without such consultation or consent, make any public statement in response to questions from the press, analysts, investors or those attending industry conferences, make internal announcements to employees and make disclosures in Purchaser SEC Documents, so long as such statements are consistent with previous press releases, public disclosures or public statements made jointly by the parties (or individually, if approved by the other party); (b) a party may, without the prior consent of the other party hereto but subject to giving advance notice to the other party, issue any such press release or make any such public announcement or statement as may be required by any Law.

13.5 <u>Expenses</u>. The costs and expenses incurred by the Company, the Purchaser and HanAll Biopharma Co., Ltd. (provided that for HanAll Biopharma Co., Ltd., reasonable costs and expenses incurred of only up to an amount of \$30,000) in connection with this Agreement and the transactions contemplated hereby, including any Transaction Expenses, shall be paid by the Purchaser at the Closing; provided, for the avoidance of doubt, that if the Closing does not take place, each party shall bear its own costs and expenses in connection with this Agreement and the transactions contemplated hereby.

13.6 <u>No Assignment or Delegation</u>. No party may assign any right or delegate any obligation hereunder, including by merger, consolidation, operation of law, or otherwise, without the written consent of the other party (which, for clarity, shall in the case of the Company be from the Purchaser and in the case of the Purchaser be from the Company). Any purported assignment or delegation without such consent shall be void, in addition to constituting a material breach of this Agreement.

13.7 <u>Governing Law.</u> This Agreement shall be construed in accordance with and governed by the laws of the State of New York, without giving effect to the conflict of laws principles thereof, except that all matters relating to the fiduciary duties of the Company Board shall be subject to the laws of Bermuda.

13.8 <u>Counterparts; facsimile signatures</u>. This Agreement may be executed in counterparts, each of which shall constitute an original, but all of which shall constitute one agreement. This Agreement shall become effective upon delivery to each party of an executed counterpart or the earlier delivery to each party of original, photocopied, or electronically transmitted signature pages that together (but need not individually) bear the signatures of all other parties.

13.9 Entire Agreement. This Agreement together with the Additional Agreements, sets forth the entire agreement of the parties with respect to the subject matter hereof and thereof and supersedes all prior and contemporaneous understandings and agreements related thereto (whether written or oral), all of which are merged herein. No provision of this Agreement or any Additional Agreement may be explained or qualified by any agreement, negotiations, understanding, discussion, conduct or course of conduct or by any trade usage. Except as otherwise expressly stated herein or any Additional Agreement, there is no condition precedent to the effectiveness of any provision hereof or thereof. No party has relied on any representation from, or warranty or agreement of, any person in entering into this Agreement, prior hereto or contemporaneous herewith or any Additional Agreement, except those expressly stated herein or therein.

13.10 <u>Severability</u>. A determination by a court or other legal authority that any provision that is not of the essence of this Agreement is legally invalid shall not affect the validity or enforceability of any other provision hereof. The parties shall cooperate in good faith to substitute (or cause such court or other legal authority to substitute) for any provision so held to be invalid a valid provision, as alike in substance to such invalid provision as is lawful.

13.11 Construction of certain terms and references; captions In this Agreement:

(a) References to particular sections and subsections, schedules, and exhibits not otherwise specified are cross-references to sections and subsections, schedules, and exhibits of this Agreement.

(b) The words "herein," "hereof," "hereunder," and words of similar import refer to this Agreement as a whole and not to any particular provision of this Agreement, and, unless the context requires otherwise, "party" means a party signatory hereto.

(c) Any use of the singular or plural, or the masculine, feminine, or neuter gender, includes the others, unless the context otherwise requires; "including" means "including without limitation;" "or" means "and/or;" "any" means "any one, more than one, or all;" and, unless otherwise specified, any financial or accounting term

has the meaning of the term under United States generally accepted accounting principles as consistently applied heretofore by the Company.

(d) Unless otherwise specified, any reference to any agreement (including this Agreement), instrument, or other document includes all schedules, exhibits, or other attachments referred to therein, and any reference to a statute or other law includes any rule, regulation, ordinance, or the like promulgated thereunder, in each case, as amended, restated, supplemented, or otherwise modified from time to time. Any reference to a numbered schedule means the same-numbered section of the Disclosure Schedule.

(c) If any action is required to be taken or notice is required to be given within a specified number of days following a specific date or event, the day of such date or event is not counted in determining the last day for such action or notice. If any action is required to be taken or notice is required to be given on or before a particular day which is not a Business Day, such action or notice shall be considered timely if it is taken or given on or before the next Business Day.

(f) Captions are not a part of this Agreement, but are included for convenience, only.

(g) "knowledge" with respect to an entity shall mean with respect to any matter in question the actual knowledge of such entity's executive officers after reasonable inquiry, and as such term is used with respect to the Company, it shall mean such knowledge of Pete Salzmann, Sandeep Kulkarni and Brad Middlekauff (each, a "Kev Company Employee"). With respect to matters involving Intellectual Property Rights, knowledge does not require that any of such entity's executive officers conduct or have conducted or obtain or have obtained any freedom-to-operate opinions or similar opinions of counsel or any intellectual property clearance searches, and no knowledge of any third party intellectual property that would have been revealed by such inquiries, opinions or searches will be imputed to such executive officers.

13.12 <u>Further Assurances</u>. Each party shall execute and deliver such documents and take such action, as may reasonably be considered within the scope of such party's obligations hereunder, necessary to effectuate the transactions contemplated by this Agreement.

13.13 <u>Third Party Beneficiaries</u>. Except as provided in <u>Section 9.7</u> and <u>Section 13.16</u>, neither this Agreement nor any provision hereof confers any benefit or right upon or may be enforced by any Person not a signatory hereto.

13.14 <u>Waiver</u>. Reference is made to the final prospectus of the Purchaser, dated May9, 2019 (the "Prospectus"). Each of the Company and the Stockholders' Representative, for itself and on behalf of the Stockholders, has read the Prospectus and understands that the Purchaser has established the Trust Account for the benefit of the public stockholders of the Purchaser and the underwriters of the IPO pursuant to the Trust Agreement and that, except for a portion of the interest earned on the amounts held in the Trust Account, the Purchaser may disburse monies from the Trust Account only for the purposes set forth in the Trust Agreement. For and in consideration of the Purchaser agreeing to enter into this Agreement, each of the Company and the Stockholders' Representative, for itself and on behalf of the Stockholders, hereby agrees that it does not have any right, title, interest or claim of any kind in or to any monies in the Trust Account and hereby agrees that it will not seek recourse against the Trust Account for any claim it may have in the future as a result of, or arising out of, any negotiations, contracts or agreements with the Purchaser. Notwithstanding the foregoing, (x) nothing herein shall serve to limit or prohibit the Company's or the Stockholders' Representative's right to pursue a claim against the Purchaser for legal relief against monies or other assets held outside the Trust Account, for specific performance or other equitable relief in connection with the consummation of the transactions contemplated by this Agreement pursuant to Section 13.18 hereof (including a claim for Purchaser to specifically perform its obligations under this Agreement and cause the disbursement of the balance of the cash remaining in the Trust Account after giving effect to the redemption of Purchaser Common Stock pursuant to the Redemptions in connection with the Closing) to the Stockholders in accordance with the terms of this Agreement and the Trust Agreement, or for fraud and (y) nothing herein shall serve to limit or prohibit any claims that the Company and the Stockholders' Representative may have in the future against the Purchaser's assets or funds that are not held in the Trust Account (including any funds that have been released from the Trust Account to the Purchaser and any assets that have been purchased or acquired with any such funds). This Section 13.14 shall survive termination of this Agreement for any reason.

13.15 Stockholders' Representative. Roivant Sciences Ltd., is hereby appointed as agent and attorney-in-fact (the "Stockholders' Representative") for each Stockholder, (a) to give and receive notices and communications to Purchaser for any purpose under this Agreement and the Additional Agreements, (b) to agree to, negotiate, enter into settlements and compromises of and demand arbitration and comply with orders of courts and awards of arbitrators under Section 11.1 or other disputes arising under or related to this Agreement, (c) to act on behalf of Stockholders in accordance with the provisions of the Agreement, the securities described herein and any other document or instrument executed in connection with the Agreement and the Share Exchange and (d) to take all actions necessary or appropriate in the judgment of the Stockholders' Representative for the accomplishment of the foregoing; provided that for (a), (b), (c) and (d) as applied to HanAll Biopharma Co., Ltd ("HanAll"), only with respect to the matters or as set forth in Schedule 13.15. The Stockholders' Representative shall provide HanAll with timely notice prior to taking any of the actions set forth in the preceding sentence on behalf of HanAll. Such agency may be changed by the Stockholders from time to time upon no less than twenty (20) days prior written notice to the Purchaser, provided, however, that the Stockholders' Representative may not be removed unless holders of at least 51% of all of the Company Common Shares on an as-if converted basis outstanding immediately prior to the transaction contemplated by this Agreement agree to such removal. Any vacancy in the position of Stockholders' Representative may be filled by approval of the holders of at least 51% of all of the Company Common Shares on an as-if converted basis outstanding immediately prior to the transaction contemplated by this Agreement. Any removal or change of the Stockholders' Representative shall not be effective until written notice is delivered to Purchaser. No bond shall be required of the Stockholders' Representative, and the Stockholders' Representative shall not receive any compensation for his services. Notices or communications to or from the Stockholders' Representative shall constitute notice to or from the Stockholders. The Stockholders' Representative shall not be liable for any act done or omitted hereunder while acting in good faith and in the exercise of reasonable business judgment. A decision, act, consent or instruction of the Stockholders' Representative shall, for all purposes hereunder, constitute a decision, act, consent or instruction of all of the Stockholders of the Company and shall be final, binding and conclusive upon each of the Stockholders. The Stockholders shall severally indemnify the Stockholders' Representative and hold him harmless against any loss, liability, or expense incurred without gross negligence or bad faith on the part of the Stockholders' Representative and arising out of or in connection with the acceptance or administration of his duties hereunder.

13.16 <u>Non-Recourse</u>. Notwithstanding any provision of this Agreement or otherwise, the parties to this Agreement agree on their own behalf and on behalf of their respective Subsidiaries and Affiliates that this Agreement may only be enforced against, and any action for breach of this Agreement may only be made against, the parties to this Agreement, and no Non-Recourse Party of a party to this Agreement shall have any liability relating to this Agreement or any of the transactions contemplated herein, except with respect to common law fraud against the person who committed such common law fraud.

13.17 Acknowledgement by Purchaser.

(a) Purchaser is not relying and Purchaser has not relied on any representations or warranties whatsoever regarding the subject matter of this Agreement, express or implied, except for the representations and warranties in <u>Article IV</u> and <u>Article VI</u>, including the Disclosure Schedule. Such representations and warranties by the Company Group and the Stockholders constitute the sole and exclusive representations and warranties of the Company Group and the Stockholders in connection with this Agreement and the transactions contemplated thereby and Purchaser understands, acknowledges and agrees that all other representations and warranties of any kind or nature whether express, implied or statutory are specifically disclaimed by the Company Group and Stockholders.

(b) In connection with the due diligence investigation of the Company Group by Purchaser and its respective Affiliates, stockholders, directors, officers, employees, agents, representatives or advisors, Purchaser and its Affiliates, stockholders, directors, officers, employees, agents, representatives and advisors have received and may continue to receive after the date hereof from the Company Group and their respective Affiliates, stockholders, directors, officers, employees, consultants, agents, representatives and advisors certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Company Group and their businesses and operations and regarding its Affiliate. Purchaser hereby acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, and that Purchaser will have no claim against the Company Group, or any of their respective Affiliates, stockholders, directors, officers, employees, consultants, agents, representatives or advisors, or any other person with respect thereto unless any such information is expressly

addressed or included in a representation or warranty contained in this Agreement. Accordingly, Purchaser hereby acknowledges and agrees that neither the Company Group nor any of their respective Affiliates, stockholders, directors, officers, employees, consultants, agents, representatives or advisors, nor any other person, has made or is making any express or implied representation or warranty with respect to such estimates, projections, forecasts, forward-looking statements or business plans unless any such information is expressly addressed or included in a representation or warranty contained in this Agreement.

13.18 Specific Performance. Each party acknowledges that the rights of each party to consummate the transactions contemplated hereby are unique, recognizes and affirms that in the event of a breach of this Agreement by any party, money damages would be inadequate and the non-breaching parties would not have adequate remedy at law, and agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed by an applicable party in accordance with their specific terms or were otherwise breached. Accordingly, each party shall be entitled to an injunction or restraining order to prevent breaches of this Agreement and to seek to enforce specifically the terms and provisions hereof, without the requirement to post any bond or other security or to prove that money damages would be inadequate, this being in addition to any other right or remedy to which such party may be entitled under this Agreement, at law or in equity.

13.19 Acknowledgement; Waiver of Conflicts; Retention of Privilege.

(a) Each of the parties hereto acknowledges and agrees that Cooley LLP ("<u>Prior</u> <u>Company Counsel</u>") has acted as counsel to the Company in various matters involving a range of issues and as counsel to the Company in connection with the negotiation of this Agreement and the Additional Agreements, and the transactions contemplated hereby and thereby.

(b) In connection with any matter or dispute under this Agreement, Purchaser hereby irrevocably waives and agrees not to assert, and agree to cause the Company Group after the Closing to irrevocably waive and agree not to asset, any conflict of interest arising from or in connection with (i) Prior Company Counsel's prior representation of the Company and (ii) Prior Company Counsel's representation of the Company Group (collectively, the "<u>Advised Parties</u>") prior to and after the Closing.

(c) Purchaser further agrees, on behalf of itself and, after the Closing, on behalf of the Company Group, that all communications in any form or format whatsoever between or among any of Prior Company Counsel, the Company, any of the Advised Parties, or any of their respective Representatives that relate in any way to the negotiation, documentation and consummation of the transactions contemplated by this Agreement or, beginning on the date of this Agreement, any dispute arising under this Agreement (collectively, the "<u>Deal Communications</u>") shall be deemed to be retained and owned collectively by the Advised Parties, shall be controlled by the Stockholders' Representative on behalf of the Company and shall not pass to or be claimed by Purchaser or the Company Group after the Closing. All Deal Communications that are attorney-client privileged (the "<u>Privileged Deal Communications</u>") shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to the Stockholders' Representative and the Company, shall be controlled by the Stockholders' Representative on behalf of the Company and shall not pass to or be claimed by Purchaser or the Company Group after the Closing; provided, further, that nothing contained herein shall be deemed to be a waiver by the Purchaser or any of its Affiliates (including, after the Closing, the Company Group and its Affiliates) of any applicable privileges or protections that can or may be asserted to prevent disclosure of any such communications to any third party.

(d) Notwithstanding the foregoing, in the event that a dispute arises between Purchaser or the Company Group, on the one hand, and a third party other than the Stockholders' Representative or the Company, on the other hand, Purchaser or the Company Group may assert the attorney-client privilege to prevent the disclosure of the Privileged Deal Communications to such third party; provided, however, that neither Purchaser or the Company Group may waive such privilege without the prior written consent of the Stockholders' Representative. In the event that Purchaser or the Company Group is legally required by governmental order or otherwise to access or obtain a copy of all or a portion of the Privileged Deal Communications, Purchaser shall immediately (and, in any event, within two (2) Business Days) notify the Stockholders' Representative an seek a protective order and Purchaser agrees to use all commercially reasonable efforts to assist therewith.

(e) To the extent that files or other materials maintained by Prior Company Counsel constitute property of its clients, only the Stockholders' Representative shall hold such property rights and Prior Company Counsel shall have no duty to reveal or disclose any such files or other materials or any Privileged Deal Communications by reason of any attorney-client relationship between Prior Company Counsel, on the one hand, and the Company Group after the Closing, on the other hand so long as such files or other materials would be subject to a privilege or protection if they were being requested in a proceeding by an unrelated third party.

(f) Purchaser agrees on behalf of itself and the Company Group after the Closing, (i) to the extent that Purchaser or, after the Closing, the Company Group receives or takes physical possession of any Deal Communications, (a) such physical possession or receipt shall not, in any way, be deemed a waiver by any of the Advised Parties or any other Person, of the privileges or protections described in this section, and (b) neither Purchaser nor the Company Group after the Closing shall assert any claim that any of the Advised Parties or any other Person waived the attorney-client privilege, attorney work-product protection or any other right or expectation of client confidence applicable to any such materials or communications, (ii) not to access or use the Deal Communications, including by way of review of any electronic data, communications or other information, or by seeking to have the Stockholders' Representative or the Company Waive the attorney-client or other privilege, and (iii) not to seek to obtain the Deal Communications from Prior Company Group after the Closing has unclications would be subject to a privilege or protection if they were being requested in a proceeding by an unrelated third party.

13.20 <u>No Survival</u>. The parties hereto, intending to modify any applicable statute of limitations, agree that (a) other than for the sole purpose of recovery under R&W Insurance Policy, the representations and warranties in this Agreement and in any certificate delivered pursuant hereto shall terminate effective as of the Closing and shall not survive the Closing for any purpose, and thereafter there shall be no liability on the part of, nor shall any claim be made by, any party or any of their respective Affiliates in respect thereof, and (b) after the Closing, there shall be no liability on the part of, nor shall any claim be made by, any party or any of their respective Affiliates in respect of any covenant or agreement to be performed prior to the Closing.

[The remainder of this page intentionally left blank; signature pages to follow]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

Purchaser:		
HEALT	TH SCIENCES ACQUISITIONS CORPORATION	
By:	/s/ Roderick Wong	
Name:	Roderick Wong, M.D.	
Title:	President	
Company:		
IMMU	IMMUNOVANT SCIENCES LTD.	
By:	/s/ Marianne L. Romeo	
Name:	Marianne L. Romeo	
Title:	Authorized Signatory	
Stockholders' Representative:		
ROIVANT SCIENCES LTD.		
By:	/s/ Marianne L. Romeo	
Name:	Marianne L. Romeo	
Title:	Authorized Signatory	
	49	

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the day and year first above written.

Stockl	nolders:
ROIVA	ANT SCIENCES LTD.
By:	/s/ Marianne L. Romeo
Name:	Marianne L. Romeo
Title:	Authorized Signatory
RTW	MASTER FUND, LTD.
By:	/s/ Roderick Wong
Name:	Roderick Wong, M.D.
Title:	Director
RTW	INNOVATION MASTER FUND, LTD.
By:	/s/ Roderick Wong
Name:	Roderick Wong, M.D.
Title:	Director
HANA	ALL BIOPHARMA CO., LTD.
By:	/s/ Seung Kook Park
Name:	Seung Kook Park
Title:	Chief Executive Officer
50	

EXHIBIT A

Form of Lock-Up Agreement

LOCK-UP AGREEMENT

THIS LOCK-UP AGREEMENT (this "<u>Agreement</u>") is dated as of September 29, 2019, by and between the undersigned (the "<u>Holder</u>") and Health Sciences Acquisitions Corporation, a Delaware corporation ("<u>Health</u> <u>Sciences</u>"). Capitalized terms used and not otherwise defined herein shall have the meanings given such terms in the Share Exchange Agreement (as defined below).

BACKGROUND

A. Health Sciences has entered into that certain Share Exchange Agreement, dated as of September29, 2019 (the "<u>Share Exchange Agreement</u>"), by and among Health Sciences, Immunovant Sciences Ltd., a Bermuda exempted limited company ("<u>Immunovant</u>"), the stockholders of Immunovant (each, a "<u>Stockholders</u>" and, collectively, the "<u>Stockholders</u>"), and Roivant Sciences Ltd., a Bermuda exempted limited company.

B. Pursuant to the Share Exchange Agreement, Health Sciences shall purchase 100% of Immunovant's outstanding common shares from the Stockholders (the "<u>Transaction</u>"). As consideration for the Transaction, Health Sciences will issue shares of its common stock, par value \$0.0001 per share (the "<u>HS Shares</u>"), to the Stockholders.

C. The Holder is the record and/or beneficial owner of common shares of Immunovant and is therefore entitled to receive HS Shares pursuant to the Share Exchange Agreement.

D. As a condition of, and as a material inducement for Health Sciences to enter into and consummate the transactions contemplated by the Share Exchange Agreement, the Holder has agreed to execute and deliver this Agreement.

NOW, THEREFORE, for and in consideration of the mutual covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties, intending to be legally bound, agree as follows:

AGREEMENT

1. Lock-Up

(a) During the Lock-up Period (as defined below), the Holder irrevocably agrees that it, he or she will not offer, sell, contract to sell, pledge or otherwise dispose of, directly or indirectly, any of the Lock-up Shares (as defined below), enter into a transaction that would have the same effect, or enter into any swap, hedge or other arrangement that transfers, in whole or in part, any of the economic consequences of ownership of such Lock-up Shares, whether any of these transactions are to be settled by delivery of any such Lock-up Shares, in cash or otherwise, publicly disclose the intention to make any offer, sale, pledge or disposition, or to enter into any transaction, swap, hedge or other arrangement, or engage in any Short Sales (as defined below) with respect to any security of Health Sciences.

(b) In furtherance of the foregoing, Health Sciences will (i) place an irrevocable stop order on all Lock-up Shares, including those which may be covered by a registration statement, and (ii) notify Health Sciences' transfer agent in writing of the stop order and the restrictions on such Lock-up Shares under this Agreement and direct Health Sciences' transfer agent not to process any attempts by the Holder to resell or transfer any Lock-up Shares, except in compliance with this Agreement.

(c) For purposes hereof, "<u>Short Sales</u>" include, without limitation, all "short sales" as defined in Rule 200 promulgated under Regulation SHO under the Securities Exchange Act of 1934, as amended (the "<u>Exchange Act</u>"), and all types of direct and indirect stock pledges, forward sale contracts, options, puts, calls, swaps and similar arrangements (including on a total return basis), and sales and other transactions through non-US broker dealers or foreign regulated brokers.

(d) For purpose of this Agreement, the <u>'Lock-up Period</u>" means: (i) with respect to 50% of the Lock-up Shares, the shorter of (A) the period commencing on the Closing Date (as defined in the Share Exchange



Agreement) and ending on the date that is six (6) months thereafter; and (B) the period commencing on the Closing Date and ending on the date on which the last reported closing price of the HS Shares on the Nasdaq Capital Market (or such other exchange on which the HS Shares are then listed) equals or exceeds \$12.50 per share (as adjusted for stock splits, stock dividends, reorganizations and recapitalizations) for any 20 trading days during any 30 trading day period thereafter; and (ii) with respect to the remaining 50% of the Lock-up Shares, the period commencing on the Closing Date and ending on the date that is six (6) months thereafter.

The restrictions set forth herein shall not apply to: (1) transfers or distributions to the Holder's current or former general or limited partners, managers or members, stockholders, other equityholders or direct or indirect affiliates (within the meaning of Rule 405 under the Securities Act of 1933, as amended) or to the estates of any of the foregoing; (2) transfers by bona fide gift to a member of the Holder's immediate family or to a trust, the beneficiary of which is the Holder or a member of the Holder's immediate family for estate planning purposes; (3) by virtue of the laws of descent and distribution upon death of the Holder; or (4) pursuant to a qualified domestic relations order, in each case where such transfere agrees to be bound by the terms of this Agreement.

In addition, if within six (6) months after the Closing Date, there is a Change of Control (as defined in the Share Exchange Agreement), then upon the consummation of such Change of Control, all Lock-up Shares shall be released from the restrictions contained herein.

2. <u>Representations and Warranties</u>. Each of the parties hereto, by their respective execution and delivery of this Agreement, hereby represents and warrants to the others and to all third party beneficiaries of this Agreement that (a) such party has the full right, capacity and authority to enter into, deliver and perform its respective obligations under this Agreement, (b) this Agreement has been duly executed and delivered by such party and is the binding and enforceable obligation of such party, enforceable against such party is accordance with the terms of this Agreement, and (c) the execution, delivery and performance of such party's obligations under this Agreement will not conflict with or breach the terms of any other agreement, contract, commitment or understanding to which such party is a party or to which the assets or securities of such party are bound. The Holder has independently evaluated the merits of its decision to enter into and deliver this Agreement, and such Holder confirms that it has not relied on the advice of Health Sciences, Health Sciences' legal counsel, Immunovant or its legal counsel, or any other person.

3. <u>Beneficial Ownership</u>. The Holder hereby represents and warrants that it does not beneficially own, directly or through its nominees (as determined in accordance with Section 13(d) of the Exchange Act, and the rules and regulations promulgated thereunder), any shares of capital stock of Health Sciences, or any economic interest in or derivative of such stock, other than those HS Shares specified on the signature page hereto. For purposes of this Agreement, the HS Shares beneficially owned by the Holder on the Closing Date, together with any HS Shares acquired during the Lock-up Period, and including any securities convertible into, or exchangeable for, or representing the rights to receive HS Shares, if any, are collectively referred to as the "Lock-up Shares."

4. <u>No Additional Fees/Payment</u>. Other than the consideration specifically referenced herein, the parties hereto agree that no fee, payment or additional consideration in any form has been or will be paid to the Holder in connection with this Agreement.

5. <u>Notices</u>. Any notices required or permitted to be sent hereunder shall be sent in writing, addressed as specified below, and shall be deemed given: (a) if by hand or recognized courier service, by 4:00PM on a business day, addressee's day and time, on the date of delivery, and otherwise on the first business day after such delivery; (b) if by fax or email, on the date that transmission is confirmed electronically, if by 4:00PM on a business day, addressee's day and time, and otherwise on the first business day after the date of such confirmation; or (c) five days after mailing by certified or registered mail, return receipt requested. Notices shall be addressed to the respective parties as follows (excluding telephone numbers, which are for convenience only), or to such other address as a party shall specify to the others in accordance with these notice provisions:

(a) If to Health Sciences, to:

Health Sciences Acquisitions Corporation 412 West 15th Street, Floor 9 New York, NY 10011 Attention: Roderick Wong, M.D. Phone: (646) 343-9280 Email: rw@rtwfunds.com



with a copy (which shall not constitute notice) to:

Loeb & Loeb LLP 345 Park Avenue New York, NY 10154 Attention: Giovanni Caruso Phone: (212) 407-4866 Email: gcaruso@loeb.com

(b) If to the Holder, to the address set forth on the Holder's signature page hereto, with a copy, which shall not constitute notice, to:

Cooley LLP 3175 Hanover Street Palo Alto, CA 94304 Attention: Frank Rahmani Phone: (650) 843-5753 Email: rahmaniff@cooley.com

or to such other address as any party may have furnished to the others in writing in accordance herewith.

6. <u>Enumeration and Headings</u>. The enumeration and headings contained in this Agreement are for convenience of reference only and shall not control or affect the meaning or construction of any of the provisions of this Agreement.

7. <u>Counterparts</u>. This Agreement may be executed in facsimile and in any number of counterparts, each of which when so executed and delivered shall be deemed an original, but all of which shall together constitute one and the same agreement.

8. <u>Successors and Assigns</u>. This Agreement and the terms, covenants, provisions and conditions hereof shall be binding upon, and shall inure to the benefit of, the respective heirs, successors and assigns of the parties hereto. The Holder hereby acknowledges and agrees that this Agreement is entered into for the benefit of and is enforceable by Health Sciences and its successors and assigns.

9. <u>Severability</u>. If any provision of this Agreement is held to be invalid or unenforceable for any reason, such provision will be conformed to prevailing law rather than voided, if possible, in order to achieve the intent of the parties and, in any event, the remaining provisions of this Agreement shall remain in full force and effect and shall be binding upon the parties hereto.

10. <u>Amendment</u>. This Agreement may be amended or modified by written agreement executed by each of the parties hereto.

11. <u>Further Assurances</u>. Each party shall do and perform, or cause to be done and performed, all such further acts and things, and shall execute and deliver all such other agreements, certificates, instruments and documents, as any other party may reasonably request in order to carry out the intent and accomplish the purposes of this Agreement and the consummation of the transactions contemplated hereby.

 No Strict Construction. The language used in this Agreement will be deemed to be the language chosen by the parties to express their mutual intent, and no rules of strict construction will be applied against any party.

13. <u>Dispute Resolution</u>. Article XII of the Share Exchange Agreement regarding arbitration of disputes is incorporated by reference herein to apply with full force to any disputes arising under this Agreement.

14. <u>Governing Law</u>. The terms and provisions of this Agreement shall be construed in accordance with the laws of the State of New York.

15. <u>Controlling Agreement</u>. To the extent the terms of this Agreement (as amended, supplemented, restated or otherwise modified from time to time) directly conflicts with a provision in the Share Exchange Agreement, the terms of this Agreement shall control.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Lock-up Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

HEA	HEALTH SCIENCES ACQUISITIONS CORPORATION	
By:		
	Name: Roderick Wong, M.D.	
	Title: President	

IN WITNESS WHEREOF, the parties hereto have caused this Lock-up Agreement to be duly executed by their respective authorized signatories as of the date first indicated above.

НО	LDER
By:	
	Name:
	Title:
Address:	
55	

EXHIBIT B

Form of Registration Rights Agreement

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

THIS AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "<u>Agreement</u>") is entered into as of the 29th day of September, 2019, by and among Health Sciences Acquisitions Corporation, a Delaware corporation (the "<u>Company</u>"), and the undersigned parties listed under Investor on the signature page hereto (each, an "<u>Investor</u>" and collectively, the "<u>Investors</u>").

WHEREAS, concurrently with the execution of this Agreement, the Company is entering into that certain Share Exchange Agreement, dated as of September 29, 2019 (the "<u>Share Exchange Agreement</u>"), by and among the Company, Immunovant Sciences Ltd., a Bermuda exempted limited company ("<u>Immunovant</u>"), the stockholders of Immunovant (the "**Immunovant Stockholders**") and Roivant Sciences Ltd., a Bermuda exempted limited company, to effect the consummation of a business combination with Immunovant (the "<u>Business Combination</u>");

WHEREAS, Health Sciences Holdings, LLC (the "<u>Sponsor</u>") is party to that certain Registration Rights Agreement, dated May 9, 2019 (the "<u>Prior Agreement</u>"), pursuant to which the Company provided the Sponsor with certain rights relating to the registration of the securities held by them; and

WHEREAS, as a condition of, and as a material inducement for Immunovant to enter into and consummate the transactions contemplated by the Share Exchange Agreement, the Company and the Sponsor have agreed to amend and restate the Prior Agreement to provide certain rights relating to the registration of shares of Common Stock (as defined below) held by stockholders of Immunovant, as of and contingent upon the closing of the Business Combination.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree that the Prior Agreement is hereby amended and restated in its entirety, as of and contingent upon the closing of the Business Combination, as follows:

1. DEFINITIONS. The following capitalized terms used herein have the following meanings:

"Agreement" means this Agreement, as amended, restated, supplemented, or otherwise modified from time to time.

"<u>Commission</u>" means the Securities and Exchange Commission, or any other federal agency then administering the Securities Act or the Exchange Act.

"<u>Closing Date</u>" is the closing date of the Business Combination and has the meaning set forth in Section 2.3 of the Share Exchange Agreement.

"Common Stock" means the common stock, par value \$0.0001 per share, of the Company.

"Company" is defined in the preamble to this Agreement.

"Demand Registration" is defined in Section 2.1.1.

"Demanding Holder" is defined in Section 2.1.1.

"Exchange Act" means the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"Form S-3" is defined in Section 2.3.

"Indemnified Party" is defined in Section 4.3.

"Indemnifying Party is defined in Section 4.3.

"Investor" is defined in the preamble to this Agreement.

"Investor Indemnified Party" is defined in Section 4.1.

"Maximum Number of Shares" is defined in Section 2.1.4.

"Notices" is defined in Section 6.3.

"Piggy-Back Registration" is defined in Section 2.2.1.

"Private Warrants" means the 10,000,000 Warrants that the Investors are privately purchasing simultaneously with the consummation of the Company's initial public offering.

"Register," "Registered" and "Registration mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such registration statement becoming effective.

"Registrable Securities" means all shares of Common Stock (i) issued or issuable to Investors in connection with the Business Combination (including shares of Common Stock that may be issued after the closing of the Business Combination pursuant to the Share Exchange Agreement) and (ii) held by the Sponsor immediately after the closing of the Business Combination and underlying the Private Warrants). Registrable Securities include any Warrants, shares of capital stock or other securities of the Company issued as a dividend or other distribution with respect to or in exchange for or in replacement of such shares of Common Stock (including shares of Common Stock underlying the Private Warrants). As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when: (a) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been otherwise transferred, new certificates for them not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of them shall not require registration under the Securities Act; (c) such securities shall have ceased to be outstanding, or (d) the Registrable Securities Act without volume limitations.

"Registration Statement" means a registration statement filed by the Company with the Commission in compliance with the Securities Act and the rules and regulations promulgated thereunder for a public offering and sale of Common Stock (other than a registration statement on Form S-4 or Form S-8, or their successors, or any registration statement covering only securities proposed to be issued in exchange for securities or assets of another entity).

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations of the Commission promulgated thereunder, all as the same shall be in effect at the time.

"<u>Underwriter</u>" means a securities dealer who purchases any Registrable Securities as principal in an underwritten offering and not as part of such dealer's market-making activities.

"<u>Warrants</u>" means the warrants of the Company, each entitling the holder thereof to purchase one half of one share of Common Stock.

2. REGISTRATION RIGHTS.

2.1 Demand Registration.

2.1.1 <u>Request for Registration</u>. At any time and from time to time on or after the Closing Date of the Business Combination, either (i) the holders of a majority-in-interest of the Registrable Securities held by the Investors or their affiliates, or the transferees of the Investors, (ii) the Sponsor or its affiliates or transferees and/or (iii) Roivant Sciences Ltd. or its affiliates or transferees, may make a written demand for registration under the Securities Act of all or part of the Registrable Securities (a "Demand Registration"). Any demand for a Demand Registration shall specify the number of shares of Registrable Securities proposed to be sold and the intended method(s) of distribution thereof. The Company will notify all holders of Registrable Securities of the demand, and each holder of Registraties Securities who wishes to include all or a portion of such holder's Registration (each such holder including shares of Registrable Securities in such registration, a "Demanding Holder") shall so notify the Company within fifteen (15) days after the receipt by the holder of the

notice from the Company. Upon any such request, the Demanding Holders shall be entitled to have their Registrable Securities included in the Demand Registration, subject to Section 2.1.4 and the provisos set forth in Section 3.1.1. The Company shall not be obligated to effect more than an aggregate of two (2) Demand Registrations under this Section 2.1.1 in respect of all Registrable Securities. For the avoidance of doubt, each of (a) the holders of a majority-in-interest of the Registrable Securities held by the Investors, (b) the Sponsor and (c) Roivant Sciences Ltd. are permitted to exercise a Demand Registration pursuant to this Section 2.1.1 with respect to their respective Registrable Securities.

2.1.2 Effective Registration. A registration will not count as a Demand Registration until the Registration Statement filed with the Commission with respect to such Demand Registration has been declared effective and the Company has complied with all of its obligations under this Agreement with respect thereto; provided, however, that if, after such Registration Statement has been declared effective, the offering of Registrates pursuant to a Demand Registration is interfered with by any stop order or injunction of the Commission or any other governmental agency or court, the Registration Statement with respect to such Demand Registration is removed, rescinded or otherwise terminated, and (ii) a majority-in-interest of the Demanding Holders thereafter elect to continue the offering; provided, further, that the Company shall not be obligated to file a second Registration Statement until a Registration Statement that has been filed is counted as a Demand Registration or is terminated.

2.1.3 <u>Underwritten Offering</u>. If a majority-in-interest of the Demanding Holders so elect and such holders so advise the Company as part of their written demand for a Demand Registration, the offering of such Registrable Securities pursuant to such Demand Registration shall be in the form of an underwritten offering. In such event, the right of any holder to include its Registrable Securities in such registration shall be conditioned upon such holder's participation in such underwriting and the inclusion of such holder's Registrable Securities in the underwriting to the extent provided herein. All Demanding Holders proposing to distribute their Registrable Securities through such underwriting shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such underwriting by a majority-in-interest of the holders initiating the Demand Registration.

2.1.4 Reduction of Offering. If the managing Underwriter or Underwriters for a Demand Registration that is to be an underwritten offering advises the Company and the Demanding Holders in writing that the dollar amount or number of shares of Registrable Securities which the Demanding Holders desire to sell, taken together with all other shares of Common Stock or other securities which the Company desires to sell and the shares of Common Stock, if any, as to which registration has been requested pursuant to written contractual piggy-back registration rights held by other shareholders of the Company who desire to sell, exceeds the maximum dollar amount or maximum number of shares that can be sold in such offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of shares, as applicable, the "Maximum Number of Shares"), then the Company shall include in such registration: (i) first, the Registrable Securities as to which Demand Registration has been requested by the Demanding Holders (pro rata in accordance with the number of shares that each such Person has requested be included in such registration, regardless of the number of shares held by each such Person (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Shares; (ii) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (i), the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; and (iii) third, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (i) and (ii), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to register pursuant to written contractual arrangements with such persons and that can be sold without exceeding the Maximum Number of Shares.

2.1.5 <u>Withdrawal</u>. If a majority-in-interest of the Demanding Holders disapprove of the terms of any underwriting or are not entitled to include all of their Registrable Securities in any offering, such majority-ininterest of the Demanding Holders may elect to withdraw from such offering by giving written notice to the Company and the Underwriter or Underwriters of their request to withdraw prior to the effectiveness of the Registration Statement
filed with the Commission with respect to such Demand Registration. If the majority-in-interest of the Demanding Holders withdraws from a proposed offering relating to a Demand Registration, then such registration shall not count as a Demand Registration provided for in Section 2.1.

2.2 Piggy-Back Registration.

2.2.1 Piggy-Back Rights. If, at any time on or after the Closing Date of the Business Combination, the Company proposes to file a Registration Statement under the Securities Act with respect to an offering of equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into, equity securities, by the Company for its own account or for shareholders of the Company for their account (or by the Company and by shareholders of the Company including, without limitation, pursuant to Section 2.1), other than a Registration Statement (i) filed in connection with any employee stock option or other benefit plan, (ii) for an exchange offer or offering of securities solely to the Company's existing shareholders, (iii) for an offering of debt that is convertible into equity securities of the Company or (iv) for a dividend reinvestment plan, then the Company shall (x) give written notice of such proposed filing to the holders of Registrable Securities as soon as practicable but in no event less than ten (10) days before the anticipated filing date, which notice shall describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, of the offering, and (y) offer to the holders of Registrable Securities in such notice the opportunity to Register the sale of such number of shares of Registrable Securities as such holders may request in writing within five (5) days following receipt of such notice (a "Piggy-Back Registration"). The Company shall cause such Registrable Securities to be included in such registration and shall use its best efforts to cause the managing Underwriter or Underwriters of a proposed underwritten offering to permit the Registrable Securities requested to be included in a Piggy-Back Registration on the same terms and conditions as any similar securities of the Company and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All holders of Registrable Securities proposing to distribute their securities through a Piggy-Back Registration that involves an Underwriter or Underwriters shall enter into an underwriting agreement in customary form with the Underwriter or Underwriters selected for such Piggy-Back Registration.

2.2.2 <u>Reduction of Offering</u>. If the managing Underwriter or Underwriters for a Piggy-Back Registration that is to be an underwritten offering advises the Company and the holders of Registrable Securities in writing that the dollar amount or number of shares of Common Stock which the Company desires to sell, taken together with the shares of Common Stock, if any, as to which Registration has been demanded pursuant to written contractual arrangements with persons other than the holders of Registrable Securities hereunder, the Registrable Securities as to which Registration has been requested under this Section 2.2, and the shares of Common Stock, if any, as to which Registration has been requested pursuant to the written contractual piggyback Registration rights of other shareholders of the Company, exceeds the Maximum Number of Shares, then the Company shall include in any such Registration:

(a) If the Registration is undertaken for the Company's account: (A) first, the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the shares of Common Stock or other securities, if any, comprised of Registrable Securities, as to which Registration has been requested pursuant to the applicable written contractual piggy-back registration rights of such security holders, Pro Rata, that can be sold without exceeding the Maximum Number of Shares; and (C) third, to the extent that the Maximum Number of shares has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to Register pursuant to written contractual piggy-back registration rights with such persons and that can be sold without exceeding the Maximum Number of Shares; and that can be sold without exceeding the Maximum Number of shares has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to Register pursuant to written contractual piggy-back registration rights with such persons and that can be sold without exceeding the Maximum Number of Shares;

(b) If the registration is a "demand" registration undertaken at the demand of persons other than either the holders of Registrable Securities, (A) first, the shares of Common Stock or other securities for the account of the demanding persons that can be sold without exceeding the Maximum Number of Shares; (B) second, to the extent that the Maximum Number of Shares has not been reached under the foregoing clause (A), the shares of Common Stock or other securities that the Company desires to sell that can be sold without exceeding the Maximum Number of Shares; (C) third, to the extent that the Maximum

Number of Shares has not been reached under the foregoing clauses (A) and (B), collectively the shares of Common Stock or other securities comprised of Registrable Securities, Pro Rata, as to which Registration has been requested pursuant to the terms hereof, that can be sold without exceeding the Maximum Number of Shares; and (D) fourth, to the extent that the Maximum Number of Shares has not been reached under the foregoing clauses (A), (B) and (C), the shares of Common Stock or other securities for the account of other persons that the Company is obligated to Register pursuant to written contractual arrangements with such persons, that can be sold without exceeding the Maximum Number of Shares.

2.2.3 <u>Withdrawal</u>. Any holder of Registrable Securities may elect to withdraw such holder's request for inclusion of Registrable Securities in any Piggy-Back Registration by giving written notice to the Company of such request to withdraw prior to the effectiveness of the Registration Statement. The Company (whether on its own determination or as the result of a withdrawal by persons making a demand pursuant to written contractual obligations) may withdraw a Registration Statement at any time prior to the effectiveness of such Registration Statement. Notwithstanding any such withdrawal, the Company shall pay all expenses incurred by the holders of Registrable Securities in connection with such Piggy-Back Registration as provided in Section 3.3.

2.2.4 <u>Unlimited Piggy-Back Registration Rights</u>. For purposes of clarity, any Registration effected pursuant to Section 2.2 hereof shall not be counted as a Registration pursuant to a Demand Registration effected under Section 2.1 hereof.

2.3 Registrations on Form S-3. At any time and from time to time on or after the Closing Date of the Business Combination the holders of a majority-in-interest of the Registrable Securities held by the Investors or their affiliates, or the transferees of the Investors may request in writing that the Company register the resale of any or all of such Registrable Securities on Form S-3 or any similar short-form registration which may be available at such time ("Form S-3"); provided, however, that the Company shall not be obligated to effect such request through an underwritten offering. Upon receipt of such written request, the Company will promptly give written notice of the proposed registration to all other holders of Registrable Securities, and, as soon as practicable thereafter, effect the registration of all or such portion of such holder's or holders' Registrable Securities as are specified in such request, together with all or such portion of the Registrable Securities or other securities of the Company, if any, of any other holder or holders joining in such request as are specified in a written request given within fifteen (15) days after receipt of such written notice from the Company; provided, however, that the Company shall not be obligated to effect any such registration pursuant to this Section 2.3: (i) if Form S-3 is not available for such offering; or (ii) if the holders of the Registrable Securities, together with the holders of any other securities of the Company entitled to inclusion in such registration, propose to sell Registrable Securities and such other securities (if any) at any aggregate price to the public of less than \$500,000. Registrations effected pursuant to this Section 2.3 shall not be counted as Demand Registrations effected pursuant to Section 2.1.

2.4 <u>Mandatory Registration</u>. Promptly following the Closing Date of the Business Combination, the Company shall file with the Commission a resale Registration Statement on any form for which the Company then qualifies or which counsel for the Company shall deem appropriate and which form shall be available for the registration under the Securities Act and sale, of all of the Registrable Securities held by the Investors that are Immunovant Stockholders. The Company shall cause the same to become effective and to maintain the effectiveness of such Registration Statement until the earlier of (i) the date that all of the securities registered thereunder have been sold by such Investors or (ii) the date such securities are freely saleable under Rule 144 under the Securities Act without volume limitations.

3. REGISTRATION PROCEDURES.

3.1 <u>Filings</u>; <u>Information</u>. Whenever the Company is required to effect the Registration of any Registrable Securities pursuant to Section 2, the Company shall use its best efforts to effect the registration and sale of such Registrable Securities in accordance with the intended method(s) of distribution thereof as expeditiously as practicable, and in connection with any such request:

3.1.1 <u>Filing Registration Statement</u>. The Company shall use its best efforts to, as expeditiously as possible after receipt of a request for a Demand Registration pursuant to Section 2.1, prepare and file with the Commission a Registration Statement on any form for which the Company then qualifies or which counsel for the Company shall deem appropriate and which form shall be available for the sale of all Registrable Securities to be Registered thereunder in accordance with the intended method(s) of distribution thereof, and shall use its best

efforts to cause such Registration Statement to become effective and use its best efforts to keep it effective for the period required by Section 3.1.3; provided, however, that the Company shall have the right to defer any Demand Registration for up to thirty (30) days, and any Piggy-Back Registration for such period as may be applicable to deferment of any Demand Registration to which such Piggy-Back Registration relates, in each case if the Company shall furnish to the holders a certificate signed by the Chief Executive Officer or Chairman of the Company stating that, in the good faith judgment of the Board of Directors of the Company, it would be materially detrimental to the Company and its shareholders for such Registration Statement to be effected at such time; provided further, however, that the Company shall not have the right to exercise the right set forth in the immediately preceding proviso more than once in any 365-day period in respect of a Demand Registration hereunder.

3.1.2 <u>Copies</u>. The Company shall, prior to filing a Registration Statement or prospectus, or any amendment or supplement thereto, furnish without charge to the holders of Registrable Securities included in such registration, and such holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the prospectus included in such Registration Statement (including each preliminary prospectus), and such holders may request in order to facilitate the disposition of the Registrable Securities owned by such holders.

3.1.3 <u>Amendments and Supplements</u>. The Company shall prepare and file with the Commission such amendments, including post-effective amendments, and supplements to such Registration Statement and the prospectus used in connection therewith as may be necessary to keep such Registration Statement effective and in compliance with the provisions of the Securities Act until all Registrable Securities and other securities covered by such Registration Statement have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or such securities have been withdrawn.

3.1.4 Notification. After the filing of a Registration Statement, the Company shall promptly, and in no event more than two (2) business days after such filing, notify the holders of Registrable Securities included in such Registration Statement of such filing, and shall further notify such holders promptly and confirm such advice in writing in all events within two (2) business days of the occurrence of any of the following: (i) when such Registration Statement becomes effective; (ii) when any post-effective amendment to such Registration Statement becomes effective; (iii) the issuance or threatened issuance by the Commission of any stop order (and the Company shall take all actions required to prevent the entry of such stop order or to remove it if entered); and (iv) any request by the Commission for any amendment or supplement to such Registration Statement or any prospectus relating thereto or for additional information or of the occurrence of an event requiring the preparation of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of the securities covered by such Registration Statement, such prospectus will not contain an untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein not misleading, and promptly make available to the holders of Registrable Securities included in such Registration Statement any such supplement or amendment; except that, before filing with the Commission a Registration Statement or prospectus or any amendment or supplement thereto, including documents incorporated by reference, the Company shall furnish, to the holders of Registrable Securities included in such Registration Statement and to the legal counsel for any such holders, copies of all such documents proposed to be filed sufficiently in advance of filing to provide such holders and legal counsel with a reasonable opportunity to review such documents and comment thereon, and the Company shall not file any Registration Statement or prospectus or amendment or supplement thereto, including documents incorporated by reference, to which such holders or their legal counsel shall object.

3.1.5 <u>State Securities Laws Compliance</u>. The Company shall use its best efforts to (i) Register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be Registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the holders of Registrable Securities in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this paragraph or subject itself to taxation in any such jurisdiction.

3.1.6 <u>Agreements for Disposition</u>. The Company shall enter into customary agreements (including, if applicable, an underwriting agreement in customary form) and take such other actions as are reasonably required in order to expedite or facilitate the disposition of such Registrable Securities. The representations, warranties and covenants of the Company in any underwriting agreement which are made to or for the benefit of any Underwriters, to the extent applicable, shall also be made to and for the benefit of the holders of Registrable Securities included in such registration statement. No holder of Registrable Securities included in such registration statement. No holder of Registrable Securities included in such registration statement or warranties or warranties in the underwriting agreement except, if applicable, with respect to such holder's organization, good standing, authority, title to Registrable Securities, lack of conflict of such sale with such holder's material agreements and organizational documents, and, with respect to written information relating to such holder. that such holder has furnished in writing expressly for inclusion in such Registration Statement.

3.1.7 <u>Cooperation</u>. The principal executive officer of the Company, the principal financial officer of the Company, the principal accounting officer of the Company and all other officers and members of the management of the Company shall cooperate fully in any offering of Registrable Securities hereunder, which cooperation shall include, without limitation, the preparation of the Registration Statement with respect to such offering and all other offering materials and related documents, and participation in meetings with Underwriters, attorneys, accountants and potential investors.

3.1.8 <u>Records</u>. The Company shall make available, for inspection by the holders of Registrable Securities included in such Registration Statement, any Underwriter participating in any disposition pursuant to such registration statement and any attorney, accountant or other professional retained by any holder of Registrable Securities included in such Registration Statement or any Underwriter, all financial and other records, pertinent corporate documents and properties of the Company, as shall be necessary to enable them to exercise their due diligence responsibility, and cause the Company's officers, directors and employees to supply all information requested by any of them in connection with such Registration Statement.

3.1.9 <u>Opinions and Comfort Letters</u>. Upon request, the Company shall furnish to each holder of Registrable Securities included in any Registration Statement a signed counterpart, addressed to such holder, of (i) any opinion of counsel to the Company delivered to any Underwriter and (ii) any comfort letter from the Company's independent public accountants delivered to any Underwriter. In the event no legal opinion is delivered to any Underwriter, the Company shall furnish to each holder of Registrable Securities included in such Registration Statement, at any time that such holder elects to use a prospectus, an opinion of counsel to the Company to the effect that the Registration Statement containing such prospectus has been declared effective and that no stop order is in effect.

3.1.10 <u>Earnings Statement</u>. The Company shall comply with all applicable rules and regulations of the Commission and the Securities Act, and make available to its shareholders, as soon as practicable, an earnings statement covering a period of twelve (12) months, which earnings statement shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder.

3.1.11 Listing. The Company shall use its best efforts to cause all Registrable Securities included in any registration to be listed on such exchanges or otherwise designated for trading in the same manner as similar securities issued by the Company are then listed or designated or, if no such similar securities are then listed or designated, in a manner satisfactory to the holders of a majority of the Registrable Securities included in such registration.

3.1.12 <u>Road Show</u>. If the Registration involves the registration of Registrable Securities involving gross proceeds in excess of \$25,000,000, the Company shall use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any underwritten offering, with all out-of-pocket costs and expenses incurred by the Company or such officers in connection with such attendance and participation to be paid by the Company.

3.2 <u>Obligation to Suspend Distribution</u>. Upon receipt of any notice from the Company of the happening of any event of the kind described in Section 3.1.4(iv), or, in the case of a resale registration on Form S-3 pursuant to Section 2.3 hereof, upon any suspension by the Company, pursuant to a written insider trading compliance program adopted by the Company's Board of Directors, of the ability of all "insiders" covered by such program to transact in the Company's securities because of the existence of material non-public information, each holder of Registrable Securities included in any Registration shall immediately discontinue disposition of such Registrable

Securities pursuant to the Registration Statement covering such Registrable Securities until such holder receives the supplemented or amended prospectus contemplated by Section 3.1.4(iv) or the restriction on the ability of "insiders" to transact in the Company's securities is removed, as applicable, and, if so directed by the Company, each such holder will deliver to the Company all copies, other than permanent file copies then in such holder's possession, of the most recent prospectus covering such Registrable Securities at the time of receipt of such notice.

3.3 Registration Expenses. The Company shall bear all costs and expenses incurred in connection with any Demand Registration pursuant to Section 2.1, any Piggy-Back Registration pursuant to Section 2.2, and any registration on Form S-3 effected pursuant to Section 2.3, and all expenses incurred in performing or complying with its other obligations under this Agreement, whether or not the Registration Statement becomes effective, including, without limitation: (i) all registration and filing fees; (ii) fees and expenses of compliance with securities or "blue sky" laws (including fees and disbursements of counsel in connection with blue sky qualifications of the Registrable Securities); (iii) printing expenses; (iv) the Company's internal expenses (including, without limitation, all salaries and expenses of its officers and employees); (v) the fees and expenses incurred in connection with the listing of the Registrable Securities as required by Section 3.1.11; (vi) Financial Industry Regulatory Authority fees; (vii) fees and disbursements of counsel for the Company and fees and expenses for independent certified public accountants retained by the Company (including the expenses or costs associated with the delivery of any opinions or comfort letters requested pursuant to Section 3.1.9); (viii) the reasonable fees and expenses of any special experts retained by the Company in connection with such registration; and (ix) the reasonable fees and expenses of one legal counsel selected by the holders of a majorityin-interest of the Registrable Securities included in such registration. The Company shall have no obligation to pay any underwriting discounts or selling commissions attributable to the Registrable Securities being sold by the holders thereof, which underwriting discounts or selling commissions shall be borne by such holders. Additionally, in an underwritten offering, all selling shareholders and the Company shall bear the expenses of the Underwriter pro rata in proportion to the respective amount of shares each is selling in such offering.

3.4 <u>Information</u>. The holders of Registrable Securities shall provide such information as may reasonably be requested by the Company, or the managing Underwriter, if any, in connection with the preparation of any Registration Statement, including amendments and supplements thereto, in order to effect the registration of any Registrable Securities under the Securities Act pursuant to Section 2 and in connection with the Company's obligation to comply with federal and applicable state securities laws.

4. INDEMNIFICATION AND CONTRIBUTION.

4.1 Indemnification by the Company. The Company agrees to indemnify and hold harmless each Investor and each other holder of Registrable Securities, and each of their respective officers, employees, affiliates, directors, partners, members, attorneys and agents, and each person, if any, who controls an Investor and each other holder of Registrable Securities (within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act) (each, an "Investor Indemnified Party"), from and against any expenses, losses, judgments, claims, damages or liabilities, whether joint or several, arising out of or based upon any untrue statement (or allegedly untrue statement) of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to such Registration Statement, or arising out of or based upon any omission (or alleged omission) to state a material fact required to be stated therein or necessary to make the statements therein not misleading, or any violation by the Company of the Securities Act or any rule or regulation promulgated thereunder applicable to the Company and relating to action or inaction required of the Company in connection with any such registration; and the Company shall promptly reimburse the Investor Indemnified Party for any legal and any other expenses reasonably incurred by such Investor Indemnified Party in connection with investigating and defending any such expense, loss, judgment, claim, damage, liability or action; provided, however, that the Company will not be liable in any such case to the extent that any such expense, loss, claim, damage or liability arises out of or is based upon any untrue statement or allegedly untrue statement or omission or alleged omission made in such Registration Statement, preliminary prospectus, final prospectus, or summary prospectus, or any such amendment or supplement, in reliance upon and in conformity with information furnished to the Company, in writing, by such selling holder expressly for use therein. The Company also shall indemnify any Underwriter of the Registrable Securities, their officers, affiliates, directors, partners, members and agents and each person who controls such Underwriter on substantially the same basis as that of the indemnification provided above in this Section 4.1.

4.2 Indemnification by Holders of Registrable Securities Each selling holder of Registrable Securities will, in the event that any registration is being effected under the Securities Act pursuant to this Agreement of any Registrable Securities held by such selling holder, indemnify and hold harmless the Company, each of its directors and officers and each Underwriter (if any), and each other selling holder and each other person, if any, who controls another selling holder or such Underwriter within the meaning of the Securities Act, against any losses, claims, judgments, damages or liabilities, whether joint or several, insofar as such losses, claims, judgments, damages or liabilities (or actions in respect thereof) arise out of or are based upon any untrue statement or allegedly untrue statement of a material fact contained in any Registration Statement under which the sale of such Registrable Securities was registered under the Securities Act, any preliminary prospectus, final prospectus or summary prospectus contained in the Registration Statement, or any amendment or supplement to the Registration Statement, or arise out of or are based upon any omission or the alleged omission to state a material fact required to be stated therein or necessary to make the statement therein not misleading, if the statement or omission was made in reliance upon and in conformity with information furnished in writing to the Company by such selling holder expressly for use therein, and shall reimburse the Company, its directors and officers, and each other selling holder or controlling person for any legal or other expenses reasonably incurred by any of them in connection with investigation or defending any such loss, claim, damage, liability or action. Each selling holder's indemnification obligations hereunder shall be several and not joint and shall be limited to the amount of any net proceeds actually received by such selling holder.

4.3 Conduct of Indemnification Proceedings. Promptly after receipt by any person of any notice of any loss, claim, damage or liability or any action in respect of which indemnity may be sought pursuant to Section 4.1 or 4.2, such person (the "Indemnified Party") shall, if a claim in respect thereof is to be made against any other person for indemnification hereunder, notify such other person (the "Indemnifying Party") in writing of the loss, claim, judgment, damage, liability or action; provided, however, that the failure by the Indemnified Party to notify the Indemnifying Party shall not relieve the Indemnifying Party from any liability which the Indemnifying Party may have to such Indemnified Party hereunder, except and solely to the extent the Indemnifying Party is actually prejudiced by such failure. If the Indemnified Party is seeking indemnification with respect to any claim or action brought against the Indemnified Party, then the Indemnifying Party shall be entitled to participate in such claim or action, and, to the extent that it wishes, jointly with all other Indemnifying Parties, to assume control of the defense thereof with counsel satisfactory to the Indemnified Party. After notice from the Indemnifying Party to the Indemnified Party of its election to assume control of the defense of such claim or action, the Indemnifying Party shall not be liable to the Indemnified Party for any legal or other expenses subsequently incurred by the Indemnified Party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that in any action in which both the Indemnified Party and the Indemnifying Party are named as defendants, the Indemnified Party shall have the right to employ separate counsel (but no more than one such separate counsel) to represent the Indemnified Party and its controlling persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought by the Indemnified Party against the Indemnifying Party, with the fees and expenses of such counsel to be paid by such Indemnifying Party if, based upon the written opinion of counsel of such Indemnified Party, representation of both parties by the same counsel would be inappropriate due to actual or potential differing interests between them. No Indemnifying Party shall, without the prior written consent of the Indemnified Party, consent to entry of judgment or effect any settlement of any claim or pending or threatened proceeding in respect of which the Indemnified Party is or could have been a party and indemnity could have been sought hereunder by such Indemnified Party, unless such judgment or settlement includes an unconditional release of such Indemnified Party from all liability arising out of such claim or proceeding.

4.4 Contribution.

4.4.1 If the indemnification provided for in the foregoing Sections 4.1, 4.2 and 4.3 is unavailable to any Indemnified Party in respect of any loss, claim, damage, liability or action referred to herein, then each such Indemnifying Party, in lieu of indemnifying such Indemnified Party, shall contribute to the amount paid or payable by such Indemnified Party as a result of such loss, claim, damage, liability or action in such proportion as is appropriate to reflect the relative fault of the Indemnified Parties and the Indemnifying Parties in connection with the actions or omissions which resulted in such loss, claim, damage, liability or action, as well as any other relevant equitable considerations. The relative fault of any Indemnified Party and any Indemnifying Party shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by such Indemnified Party or such Indemnifying Party and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission.

4.4.2 The parties hereto agree that it would not be just and equitable if contribution pursuant to this Section 4.4 were determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in the immediately preceding Section 4.4.1.

4.4.3 The amount paid or payable by an Indemnified Party as a result of any loss, claim, damage, liability or action referred to in the immediately preceding paragraph shall be deemed to include, subject to the limitations set forth above, any legal or other expenses incurred by such Indemnified Party in connection with investigating or defending any such action or claim. Notwithstanding the provisions of this Section 4.4, no holder of Registrable Securities shall be required to contribute any amount in excess of the dollar amount of the net proceeds (after payment of any underwriting fees, discounts, commissions or taxes) actually received by such holder from the sale of Registrable Securities which gave rise to such contribution obligation. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any person who was not guilty of such fraudulent misrepresentation.

5. <u>RULE 144</u>.

5.1 <u>Rule 144</u>. The Company covenants that it shall file any reports required to be filed by it under the Securities Act and the Exchange Act and shall take such further action as the holders of Registrable Securities may reasonably request, all to the extent required from time to time to enable such holders to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission.

6. MISCELLANEOUS

6.1 <u>Other Registration Rights</u>. The Company represents and warrants that no person, other than the holders of the Registrable Securities, has any right to require the Company to register any shares of the Company's capital stock for sale or to include shares of the Company's capital stock in any registration filed by the Company for the sale of shares of capital stock for its own account or for the account of any other person.

6.2 Assignment; No Third Party Beneficiaries. This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part. This Agreement and the rights, duties and obligations of the holders of Registrable Securities hereunder may be freely assigned or delegated by such holder of Registrable Securities in conjunction with and to the extent of any transfer of Registrable Securities by any such holder. This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties, to the permitted assigns of the Investors or holder of Registrable Securities. This Agreement is not intended to confer any rights or benefits on any persons that are not party hereto other than as expressly set forth in Article 4 and this Section 6.2.

6.3 <u>Notices</u>. All notices, demands, requests, consents, approvals or other communications (collectively, "<u>Notices</u>") required or permitted to be given hereunder or which are given with respect to this Agreement shall be in writing and shall be personally served, delivered by reputable air courier service with charges prepaid, or transmitted by hand delivery, telegram, telex or facsimile, addressed as set forth below, or to such other address as such party shall have specified most recently by written notice. Notice shall be deemed given on the date of service or transmission if personally served or transmitted by telegram, telex or facsimile; provided that, if such service or transmission is not on a business day or is after normal business hours, then such notice shall be deemed given on the next business day. Notice otherwise sent as provided herein shall be deemed given on the next business day following timely delivery of such notice to a reputable air courier service with an order for next-day delivery.

To the Company:

Health Sciences Acquisitions Corporation 412 West 15th Street, Floor 9 New York, NY 10011 Attention: Roderick Wong, M.D. Phone: (646) 343-9280 Email: rw@rtwfunds.com

with a copy to (which shall not constitute notice):

Loeb & Loeb LLP 345 Park Avenue New York, NY 10154 Attention: Giovanni Caruso, Esq. Phone: (212) 407-4866 Email: gcaruso@loeb.com

To an Investor, to the address set forth below such Investor's name on the signature pages hereto.

6.4 <u>Severability</u>. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible that is valid and enforceable.

6.5 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts, each of which shall be deemed an original, and all of which taken together shall constitute one and the same instrument.

6.6 Entire Agreement. This Agreement (including all agreements entered into pursuant hereto and all certificates and instruments delivered pursuant hereto and thereto) constitute the entire agreement of the parties with respect to the subject matter hereof and supersede all prior and contemporaneous agreements, representations, understandings, negotiations and discussions between the parties, whether oral or written.

6.7 <u>Modifications and Amendments</u>. No amendment, modification or termination of this Agreement shall be binding upon any party unless executed in writing by such party.

6.8 <u>Titles and Headings</u>. Titles and headings of sections of this Agreement are for convenience only and shall not affect the construction of any provision of this Agreement.

6.9 Waivers and Extensions. Any party to this Agreement may waive any right, breach or default which such party has the right to waive, provided that such waiver will not be effective against the waiving party unless it is in writing, is signed by such party, and specifically refers to this Agreement. Waivers may be made in advance or after the right waived has arisen or the breach or default waived has occurred. Any waiver may be conditional. No waiver of any breach of any agreement or provision herein contained shall be deemed a waiver of any preceding or succeeding breach thereof nor of any other agreement or provision herein contained. No waiver or extension of time for performance of any obligations or acts shall be deemed a waiver or extension of the time for performance of any other agreement.

6.10 <u>Remedies Cumulative</u>. In the event that the Company fails to observe or perform any covenant or agreement to be observed or performed under this Agreement, the Investor or any other holder of Registrable Securities may proceed to protect and enforce its rights by suit in equity or action at law, whether for specific performance of any term contained in this Agreement or for an injunction against the breach of any such term or in aid of the exercise of any power granted in this Agreement or to enforce any other legal or equitable right, or to take any one or more of such actions, without being required to post a bond. None of the rights, powers or remedies conferred under this Agreement shall be mutually exclusive, and each such right, power or remedy shall be cumulative and in addition to any other right, power or remedy, whether conferred by this Agreement or now or hereafter available at law, in equity, by statute or otherwise.

6.11 <u>Governing Law</u>. This Agreement shall be governed by, interpreted under, and construed in accordance with the internal laws of the State of New York applicable to agreements made and to be performed within the State of New York, without giving effect to any choice-of-law provisions thereof that would compel the application of the substantive laws of any other jurisdiction.

6.12 <u>Waiver of Trial by Jury</u>. Each party hereby irrevocably and unconditionally waives the right to a trial by jury in any action, suit, counterclaim or other proceeding (whether based on contract, tort or otherwise) arising out of, connected with or relating to this Agreement, the transactions contemplated hereby, or the actions of the Investor in the negotiation, administration, performance or enforcement hereof.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

66

IN WITNESS WHEREOF, the parties have caused this Registration Rights Agreement to be executed and delivered by their duly authorized representatives as of the date first written above.

COMPANY:
HEALTH SCIENCES ACQUISITIONS CORPORATION
By:
Name:
Title:

INVESTOR:	
Print Name of Investor	
Signature	
By:	
Title:	
Address:	
68	

EXHIBIT C

Form of Sponsor Restricted Stock Agreement

RESTRICTED STOCK AGREEMENT

This Restricted Stock Agreement (this "<u>Agreement</u>") is entered into as of the 29th day of September, 2019, by and between Health Sciences Acquisitions Corporation, a Delaware corporation (the "<u>Company</u>"), and Health Sciences Holdings, LLC (the "<u>Holder</u>").

WITNESSETH:

WHEREAS, in December 2018, the Holder purchased 2,875,000shares (the "<u>Shares</u>") of the Company's common stock (the "<u>Common Stock</u>");

WHEREAS, the Shares are currently held in escrow pursuant to the terms of that certain Stock Escrow Agreement, dated May 9, 2019 (the "<u>Escrow Agreement</u>"), between the Company, the Holder and Continental Stock Transfer & Trust Company, a New York corporation (the "<u>Escrow Agreem</u>");

WHEREAS, concurrently with the execution of this Agreement, the Company is entering into that certain Share Exchange Agreement, dated as of September 29, 2019 (the "Share Exchange Agreement"), by and among the Company, Immunovant Sciences Ltd., a Bermuda exempted limited company ("Immunovant"), the stockholders of Immunovant and Roivant Sciences Ltd., a Bermuda exempted limited company, to effect the consummation of a business combination with Immunovant (the "Business Combination"); and

WHEREAS, Holder is entering in to this Agreement as a condition of, and as a material inducement for Immunovant to enter into and consummate the transactions contemplated by the Share Exchange Agreement.

NOW, THEREFORE, in consideration of the foregoing and of the mutual covenants herein contained and other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the parties hereto hereby agree as follows:

1. <u>Cancellation of Shares</u>. Concurrently with the Closing (as defined in the Share Exchange Agreement) the Company shall instruct the Escrow Agent to cancel a number of Shares (the "<u>Canceled Shares</u>") equal to: (a) 1,800,000, *multiplied by* (b) (i) the number of shares of Common Stock validly redeemed by holders thereof in connection with the Business Combination as reflected in the records of the Company's transfer agent, *divided by* (ii) 11,500,000.

2. <u>Share Restriction</u>. Concurrently with the Closing, the Company shall instruct the Escrow Agent that a number of Shares (the "<u>Restricted Shares</u>") equal to 1,800,000 *minus* the number of Canceled Shares, shall be held in escrow and subject to potential forfeiture until vested in accordance with Section 4 below.

3. <u>Representations</u>. The Company and the Holder hereby represent and warrant as follows:

a. Company Representations.

i) Authority; Due Execution. The Company has all requisite power and authority and the legal capacity to execute, deliver and perform this Agreement. The execution, delivery and performance of this Agreement has been duly authorized by all necessary corporate action on the part of the Company. This Agreement has been duly executed and delivered by or on behalf of the Company and constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except that such enforceability may be limited by bankruptcy, insolvency, moratorium or other similar laws affecting or relating to creditors' rights generally, and is subject to general principles of equity.

i) No Conflicts. The execution and delivery of this Agreement by the Company does not, and the performance of this Agreement by the Company will not (A) conflict with or violate any law applicable to the Company of which the Company is aware, or (B) result in the creation of a lien or encumbrance on the Company's assets pursuant to any note, bond, mortgage, indenture, contract, agreement, lease, license, permit, franchise or other instrument or obligation to which the Company is a party or by which the Company or any of the Company's assets is bound or affected.

b. Holder Representations.

i) The Holder has all requisite power and authority and the legal capacity to execute, deliver and perform this Agreement. This Agreement has been duly executed and delivered by the Holder and constitutes a legal, valid and binding obligation of the Holder, enforceable against the Holder in accordance with its terms, except that such enforceability may be limited by bankruptcy, insolvency, moratorium or other similar laws affecting or relating to creditors' rights generally, and is subject to general principles of equity.

ii) The Holder owns, of record and beneficially, and has good, valid and indefeasible title to the Shares free and clear of any and all mortgages, pledges, security interests, encumbrances, liens or charges of any kind, except for those imposed on the Holder in connection with the Company's initial public offering. Except for those agreements the Company is a party to, there are no options, rights, voting trusts, stockholder agreements or any other contracts or understandings to which the Holder is a party or by which the Holder or the Shares are bound with respect to the issuance, sale, transfer, voting or registration of the Shares.

4 Treatment of Restricted Shares.

a. *Registration.* The Restricted Shares shall remain in the name of the Holder registered in book entry form at the Company's transfer agent. Unless and until the Restricted Shares are forfeited as provided herein, Holder shall be entitled to vote such shares. Any distributions of Common Stock declared with respect to the Restricted Shares, including, but not limited to, shares of Common Stock issued as a result of a stock dividend, stock split, combination of shares or otherwise, shall be set aside and not paid until the Restricted Shares have been vested and released to the Holder or, if the Restricted Shares are not vested and released in accordance with this Agreement, then all such distributions declared on such Restricted Shares shall be forfeited.

b. *Nontransferability*. Except as otherwise required by law, Restricted Shares that have not vested in accordance with the terms of this Agreement may not be sold, assigned, exchanged, transferred, pledged, hypothecated or otherwise disposed of, except to the Company as provided herein.

c. Vesting. The Restricted Shares shall vest and no longer be subject to forfeiture in accordance with the following schedule:

i) 50% of the Restricted Shares shall vest and no longer be subject to forfeiture upon the date of final determination pursuant to Section 3.3(a) of the Share Exchange Agreement that Milestone #1 has been achieved and the applicable Earnout Shares (as defined therein) have become deliverable thereunder; and

ii) 50% of the Restricted Shares shall vest and no longer be subject to forfeiture upon the date of final determination pursuant to Section 3.3(a) of the Share Exchange Agreement that Milestone #2 has been achieved and the applicable Earnout Shares have become deliverable thereunder.

Notwithstanding the foregoing, in the event that, prior to the vesting of all Restricted Shares pursuant to clauses (i) and (ii) above, there is an Acceleration Event (as defined in the Share Exchange Agreement), then all Restricted Shares shall immediately vest in full and no longer be subject to forfeiture upon the consummation of such Acceleration Event; *provided* that, the Restricted Shares, if any, that remain subject to the vesting conditions set forth in clauses (i) and (ii) above shall not be deemed vested if the value of the consideration to be received in exchange for each share of Common Stock in the event of an Acceleration Event that is a Change of Control (as defined in the Share Exchange Agreement) is lower than the applicable stock price thresholds referenced thereby.

d. *Delivery following Vesting*. Upon the achievement of the vesting conditions set forth above, the Company shall instruct the Escrow Agent to remove any legend reflecting the limitation of transferability, the risk of forfeiture and other restrictions under this Agreement from such vested Restricted Shares, and such Shares will be eligible for release from escrow. For the avoidance of doubt, to the extent then-applicable, such Shares will remain subject to the restrictions set forth in Section 3 of the Escrow Agreement.

e. *Cancellation of Unvested Restricted Shares.* To the extent that all Restricted Shares have not vested pursuant to the terms of this Agreement by the date it is finally determined that the stockholders of Immunovant are not entitled or eligible to receive any further Earnout Shares under the Share Exchange Agreement, the Company and the Holder shall instruct the Escrow Agent to cancel such Restricted Shares that have not thenvested.

5. <u>Notice</u>. All notices, request, demands, waivers and communications required or permitted to be given hereunder shall be in writing and shall be delivered in person or mailed, certified or registered mail with postage prepaid, or sent by facsimile, as follows:

If to Company, to it at:

Health Sciences Acquisitions Corporation 412 West 15th Street, Floor 9 New York, NY 10011 Attention: Roderick Wong, M.D. Phone: (646) 343-9280 Email: rw@rtwfunds.com

with a copy (which shall not constitute notice) to:

Loeb & Loeb LLP 345 Park Avenue New York, NY 10154 Attention: Giovanni Caruso Phone: (212) 407-4866 Email: gcaruso@loeb.com

If to Holder, to Holder at Holder's last known mailing address specified in the Company's records,

or to such other address as either party hereto shall specify by notice in writing to the other party in accordance with this Section. All such notices, requests, demands, waivers and communications shall be deemed to have been received on the date when given unless mailed, in which case on the third business day after the mailing.

6. <u>Assignment</u>. Neither party may assign any of its respective rights or obligations hereunder, except by operation of law.

7. <u>Amendments</u>. This Agreement may not be amended, modified, or terminated without the written agreement of both parties hereto.

8. <u>Governing Law</u>. The validity, interpretation, construction and performance of this Agreement shall be governed by the internal laws of the State of New York, without regard to the principles of conflicts of law.

9. <u>Counterparts</u>. This Agreement may be executed in two counterparts, each of which shall be an original, but both of which together shall constitute one and the same agreement.

[Signature pages follow.]

IN WITNESS WHEREOF, the Company and Holder have entered into this Agreement as of the grant date specified above.

HEALT	HEALTH SCIENCES ACQUISITIONS CORPORATION					
By:						
Name:	Roderick Wong, M.D.					
Title:	President					
HEALT	HEALTH SCIENCES HOLDINGS, LLC					
By:						
Name:	Naveen Yalamanchi, M.D.					
Title:	Director					
7:	2					

EXHIBIT D

Purchaser Charter Documents

The form of amended and restated certificate of incorporation and bylaws will be determined following the date of this Agreement.

EXHIBIT E

R&W Insurance Policy

Immunovant to Merge with Health Sciences Acquisitions Corporation, Creating New Publicly Listed FcRn-Focused Company

- Top shareholders of Immunovant post-closing to include Roivant Sciences and blue-chip biotech investors, including RTW Investments, BVF Partners, Adage Capital Management, Cormorant Asset Management, Eventide Asset Management, and Perceptive Advisors
- Immunovant is developing IMVT-1401, a fully human antibody to FcRn that delivered a mean IgG reduction of nearly 80% in a Phase 1 study of healthy volunteers
 receiving 4 weekly 680 mg subcutaneous injections
- Immunovant is expected to have more than \$100 million at closing to fund development of IMVT-1401 into 2H 2021
- Top-line data from ongoing Phase 2a trial in Graves' ophthalmopathy expected by Q1 2020
- Top-line data from ongoing Phase 2a trial in myasthenia gravis expected by Q2 2020

NEW YORK and LONDON, October 2, 2019 /PRNewswire/ – Health Sciences Acquisitions Corporation ("HSAC," NASDAQ: HSAC), a special purpose acquisition company sponsored by RTW Investments, and Immunovant Sciences Ltd. ("Immunovant"), a clinical-stage biopharmaceutical company focused on enabling normal lives for patients with autoimmune diseases, today announced that they have entered into a definitive share exchange agreement ("SEA"). HSAC will acquire 100% of the issued and outstanding shares in Immunovant. Upon closing, the combined company will be called Immunovant, Inc.

"We are thrilled to have the opportunity to partner with the team at Immunovant. We believe IMVT-1401 is a uniquely compelling asset within the FcRn drug class, which we expect will become a cornerstone therapy for treating many auto-antibody driven diseases," said Roderick T. Wong, M.D., President, Chief Executive Officer and Chairman of HSAC and Managing Partner and Chief Investment Officer of RTW Investments.

In addition to the merger described above, Immunovant also announced today that it completed a \$35 million private bridge financing with RTW Investments, BVF Partners, and Roivant Sciences Ltd. ("Roivant"). The notes issued in this financing will convert into common shares of Immunovant immediately prior to the closing of the business combination.

The combined company is expected to have more than \$100 million at closing and will be led by Immunovant's experienced management team, headed by Chief Executive Officer Pete Salzmann, M.D. HSAC has received written commitments from certain HSAC shareholders to vote in favor of the transaction and to not redeem their shares, which, in aggregate, more than satisfies the minimum cash closing condition set forth in the SEA. Current HSAC shareholders supporting the transaction include RTW Investments, Adage Capital Management, Cormorant Asset Management, Eventide Asset Management, and Perceptive Advisors.

"I am proud of the many milestones delivered by the Immunovant team this year, including completion of a comprehensive Phase 1 program demonstrating robust IgG reductions with simple subcutaneous injections and initiation of a broad Phase 2 program with both first-in-class and best-in-class potential in multiple diseases with high unmet patient need. We believe the potency of IMVT-1401 and the ability to administer IMVT-1401 as a simple subcutaneous injection represent important potentially differentiating features of this product candidate. Today's financing transaction will allow us to continue to pursue our vision of enabling normal lives for patients with autoimmune diseases," said Pete Salzmann, M.D., Chief Executive Officer of Immunovant.

Immunovant is developing IMVT-1401, a fully human anti-FcRn monoclonal antibody with the potential to treat IgG-mediated autoimmune diseases. IMVT-1401 is the result of a multi-year research program by Immunovant's partner, HanAll Biopharma, to engineer a highly potent anti-FcRn antibody specifically optimized for subcutaneous injection with a small gauge needle. In a Phase 1 study of healthy volunteers receiving 4 weekly subcutaneous injections, IMVT-1401 delivered a mean IgG reduction of 63% at a dose of 340 mg and a mean IgG reduction of 78% at a dose of 680 mg.

Proceeds from this transaction are expected to finance Phase 2 development of IMVT-1401 in three indications. IMVT-1401 is currently being tested in a Phase 2a trial for Graves' ophthalmopathy (potentially a first-in-class anti-FcRn), with top-line data expected by Q1 2020, and in a Phase 2a trial for myasthenia gravis, with top-line data expected by Q2 2020. Immunovant plans to file an IND for a third indication, warm autoimmune hemolytic anemia, later this year. The company also intends to pursue additional indications in the future.

A corporate presentation describing Immunovant's development plans can be found at www.immunovant.com.

Summary of Transaction

On September 29, 2019, HSAC entered into the SEA with Immunovant and its shareholders, including Roivant. Upon the closing of the transactions contemplated in the SEA, HSAC will acquire all of the shares of Immunovant for the consideration described below, and Immunovant will become a wholly owned subsidiary of HSAC.

Upon the closing of the transactions, the current Immunovant shareholders will sell to HSAC, and HSAC will purchase from the current Immunovant shareholders, all of the issued and outstanding Immunovant shares, and HSAC will issue (or reserve for issuance upon the exercise of options) approximately 43 million HSAC shares to the current Immunovant shareholders. The aggregate value of the consideration to be paid by HSAC in the business combination is \$395 million, before giving effect to Immunovant's bridge financing.

Assuming no redemption from HSAC shareholders, it is estimated that the current security holders of Immunovant will own approximately 77% of the combined company.

Immunovant shareholders may, subject to the terms of the SEA, receive up to an additional 20 million HSAC shares (the "Earnout Shares"): 10 million shares if the share price exceeds \$17.50 by March 31, 2023 and an additional 10 million shares if the share price exceeds \$31.50 by March 31, 2025.

In connection with the transactions, HSAC's sponsor has agreed to cancel all 10 million of its private warrants. Furthermore, subject to terms of the SEA, 1.8 million of the sponsor's founder shares will be cancelled unless HSAC's common stock exceeds certain stock prices on substantially identical terms and conditions as the Earnout Shares.



The description of the transaction contained herein is only a high-level summary and is qualified in its entirety by reference to the definitive agreement relating to the transaction. A copy of the definitive agreement, this press release and a corporate presentation will be filed today by HSAC with the Securities and Exchange Commission (the "SEC") as exhibits to a Current Report on Form 8-K, which can be accessed through the SEC's website at www.sec.gov.

Chardan is acting as HSAC's M&A and capital markets advisor. Leerink is acting as financial advisor to Immunovant. Loeb & Loeb LLP is representing HSAC. Cooley LLP is representing Immunovant.

About Immunovant

Immunovant, a member of the Roivant family of companies, is a clinical-stage biopharmaceutical company focused on enabling normal lives for patients with autoimmune diseases. Immunovant is developing IMVT-1401, a novel, fully human anti-FcRn monoclonal antibody, as a subcutaneous injection for the treatment of autoimmune diseases mediated by pathogenic IgG antibodies. For further information about Immunovant, please visit www.immunovant.com.

About Roivant

Roivant Sciences aims to improve health by rapidly delivering innovative medicines and technologies to patients. Roivant does this by building Vants – nimble, entrepreneurial biotech, and healthcare technology companies with a unique approach to sourcing talent, aligning incentives, and deploying technology to drive greater efficiency in R&D and commercialization. For further information about Roivant, please visit www.roivant.com.

About Health Sciences Acquisitions Corporation

HSAC is a Delaware company established for the purpose of entering into a merger, share exchange, asset acquisition, share purchase, recapitalization, reorganization or similar business combination with one or more businesses or entities. On May 14, 2019, HSAC raised \$115 million to achieve this goal. As of June 30, 2019, there was approximately \$115.3 million in HSAC's trust account. HSAC is sponsored by RTW Investments.

About RTW Investments

RTW Investments, LP ("RTW") is a New York-based investment firm that focuses on identifying transformational and disruptive innovations in biopharmaceutical and medical technologies. As a leading partner of industry and academia, RTW utilizes deep scientific expertise and a rigorous and comprehensive process to support emerging medical therapies. For further information about RTW, please visit www.rtwfunds.com.



Important Notice Regarding Forward-Looking Statements

This press release contains certain "forward-looking statements" within the meaning of the Securities Act of 1933 and the Securities Exchange Act of 1934, both as amended. Statements that are not historical facts, including statements about the pending business combination between HSAC and the stockholders of Immunovant and the transactions contemplated thereby, and the parties' perspectives and expectations, are forward-looking statements. Such statements include, but are not limited to, statements regarding the proposed transaction, including the anticipated initial enterprise value, cash available at closing, the anticipated use of the combined company's cash and cash equivalents, initiation, timing, progress, and reporting of results of Immunovant's development programs, the potential benefits of Immunovant's product candidates, the benefits of the proposed transaction, integration plans, anticipated future financial and operating performance and results, including estimates for growth, the expected management and governance of the combined company, and the expected timing of the transactions. The words "expect," "believe," "estimate," "intend," "plan" and similar expressions indicate forward-looking statements. These forward-looking statements are not guarantees of future performance and are subject to various risks and uncertainties, assumptions (including assumptions about general economic, market, industry and operational factors), known or unknown, which could cause the actual results to vary materially from those indicated or anticipated.

Such risks and uncertainties include, but are not limited to: (i) risks related to the expected timing and likelihood of completion of the pending transaction, including the risk that the transaction may not close due to one or more closing conditions to the transaction not being satisfied or waived, such as regulatory approvals not being obtained, on a timely basis or otherwise, or that a governmental entity prohibited, delayed or refused to grant approval for the consummation of the transaction or required certain conditions, limitations or restrictions in connection with such approvals; (ii) risks related to the ability of HSAC and Immunovant to successfully integrate the businesses; (iii) the occurrence of any event, change or other circumstances that could give rise to the termination of the applicable transaction agreements; (iv) the risk that there may be a material adverse change with respect to the financial position, performance, operations or prospects of Immunovant or HSAC; (v) risks related to disruption of management time from ongoing business operations due to the proposed transaction; (vi) the risk that any announcements relating to the proposed transaction could have adverse effects on the market price of HSAC's common stock; (vii) risks associated with the financing of the proposed transaction; and (viii) risks related to the timing, cost and results of Immunovant's clinical trials and regulatory submissions. A further list and description of risks and uncertainties can be found in HSAC's Registration Statement field on Form S-1 filed with the SEC which you are encouraged to read. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those indicated or anticipated by such forward-looking statements. Accordingly, you are cautioned not to place undue reliance on these forward-looking statements to reflect events or circumstances after the date they were made, and HSAC, Immunovant, and their subsid

Disclaimer

This communication shall neither constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which the offer, solicitation, or sale would be unlawful prior to the registration or qualification under the securities laws of any such jurisdiction.

Participants in Solicitation

HSAC, Immunovant, and their respective directors, executive officers and employees and other persons may be deemed to be participants in the solicitation of proxies from the holders of HSAC common stock in respect of the proposed transaction. Information about HSAC's directors and executive officers and their ownership of HSAC's common stock is set forth in HSAC's Registration Statement filed on Form S-1 filed with the SEC on May 3, 2019, as modified or supplemented by any Form 3 or Form 4 filed with the SEC since the date of such filing. Other information regarding the interests of the participants in the proxy solicitation will be included in the proxy statement pertaining to the proposed transaction when it becomes available. These documents can be obtained free of charge from the sources indicated below.

Additional Information and Where to Find It

In connection with the transaction described herein, HSAC has filed and will file relevant materials with the SEC, including a proxy statement on Schedule 14A. Promptly after filing its definitive proxy statement with the SEC, HSAC will mail the definitive proxy statement and a proxy card to each stockholder entitled to vote at the special meeting relating to the transaction. Investors and security holders of HSAC are urged to read these materials (including any amendments or supplements thereto) and any other relevant documents in connection with the transaction that HSAC will file with the SEC when they become available because they will contain important information about HSAC, Immunovant and the transaction. The preliminary proxy statement, the definitive proxy statement and other relevant materials in connection with the transaction (when they become available), and any other documents filed by HSAC with the SEC, may be obtained free of charge at the SEC's website (www.sec.gov) or by writing to HSAC at 412 West 15th Street, Floor 9. New York, NY 10011.

Contacts:

Sandeep Kulkarni, M.D. Chief Operating Officer Immunovant, Inc. info@immunovant.com

Stephanie A. Sirota Vice President of Corporate Strategy and Corporate Communications Health Sciences Acquisitions Corporation hsac@rtwfunds.com



Corporate Overview

October 2019

This presentation has been prepared by Immunovant Sciences Ltd. ("we," "us," "our," "Immunovant" or the "Company") and contains forward-looking statements. In some cases, you can identify forward-looking statements by the words "anticipate," "believe," "continue," "could," "estimate," "expect," "intend," "may," "might," "objective," "ongoing," "plan," "potential," "predict," "project," "should," "will," and "would," or the negative of these terms, or other comparable terminology intended to identify statements about the future. These statements are based on management's current beliefs and expectations. These statements include but are not limited to statements regarding our business strategy, our plans to develop and commercialize our product candidates, our plans to enter into commercial transactions or strategic partnerships, the safety and efficacy of our product candidates, our expectations regarding timing, the design and results of clinical trials of our product candidates, our plans and expected timing with respect to regulatory filings and approvals, the size and growth potential of the markets for our product candidates, and our ability to serve those markets, and our plans and expected timing with respect to regulatory filings and approvals. These statements involve substantial known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from the information expressed or implied by these forward looking statements. We may not actually achieve the plans, intentions or expectations disclosed in our forward-looking statements, and you should not place undue reliance on our forward-looking statements. Actual results or events could differ materially from the plans, intentions and expectations disclosed in the forward-looking statements we make as a result of various important factors, including, without limitation, those inherent in the preclinical and clinical development process and the regulatory approval process, the risks and uncertainties in commercialization and gaining market acceptance, the risks associated with protecting and defending our intellectual property rights, our reliance on third-parties to conduct clinical and preclinical trials, our reliance on third-party suppliers to manufacture clinical, preclinical and any future commercial supplies of our product candidates, increased regulatory requirements, our ability to provide the financial support and resources necessary to develop our product candidate on the expected timeline, our ability to identify and acquire or in-license new product candidates and competition from others developing products for similar uses. These statements are subject to the risk that clinical study data are subject to differing interpretations, and regulatory agencies, medical and scientific experts and others may not share our view of the clinical study data. There can be no assurance that the clinical programs for our product candidate will be successful in demonstrating safety and/or efficacy, that we will not encounter problems or delays in clinical development, or that any of our product candidates will ever receive regulatory approval or be successfully commercialized. The forward-looking statements in this presentation represent our views as of the date of this presentation. We anticipate that subsequent events and developments will cause our views to change. While we may elect to update these forward-looking statements at some point in the future, we have no current intention of doing so except to the extent required by applicable law. You should, therefore, not rely on these forward-looking statements as representing our views as of any date subsequent to the date of this presentation.

HSAC, Immunovant, and their respective directors, executive officers and employees and other persons may be deemed to be participants in the solicitation of proxies from the holders of HSAC common stock in respect of the proposed transaction. Information about HSAC's directors and executive officers and their ownership of HSAC's common stock is set forth in HSAC's Registration Statement filed on Form S-1 filed with the SEC on May 3, 2019, as modified or supplemented by any Form 3 or Form 4 filed with the SEC since the date of such filing. Other information regarding the interests of the participants in the proxy solicitation will be included in the proxy statement pertaining to the proposed transaction when it becomes available. These documents can be obtained free of charge from the sources indicated below.

In connection with the transaction described herein, HSAC has filed and will file relevant materials with the SEC, including a proxy statement on Schedule 14A. Promptly after filing its definitive proxy statement with the SEC, HSAC will mail the definitive proxy statement and a proxy card to each stockholder entitled to vote at the special meeting relating to the transaction. Investors and security holders of HSAC are urged to read these materials (including any amendments or supplements thereto) and any other relevant documents in connection with the transaction that HSAC will file with the SEC when they become available because they will contain important information about HSAC, Immunovant and the transaction. The preliminary proxy statement, the definitive proxy statement and other relevant materials in connection with the transaction with the transaction with the transaction to the SEC, may be obtained free of charge at the SEC's website (www.sec.gov) or by writing to HSAC at 412 West 15th Street, Floor 9. New York, NY 10011.

By attending or receiving this presentation you acknowledge that you will be solely responsible for your own assessment of the market and our market position and that you will conduct your own analysis and be solely responsible for forming your own view of the potential future performance of our business and the transactions. The information herein does not purport to be all-inclusive. The data and estimates contained herein was derived from various internal and external sources, and involves a number of assumptions and limitations. You are cautioned not to give undue weight to such data or estimates. Neither HSAC nor Immunovant nor any other person makes any representation as to the accuracy or completeness of such data or undertakes any obligation to update such data after the date of this presentation. This presentation does not constitute an offer to sell or the solicitation of an offer buy securities, nor is it a solicitation of any vote or approval of the transactions, nor will there be any sale of these securities in, any state or jurisdiction in which such offer, solicitation or sale would be unlawful. Any offering of securities or solicitation to vote regarding the proposed transactions described herein will be made only by means of a proxy statement on Schedule 14A filed with the SEC.

Transaction

Immunovant and HSAC to merge

TRANSACTION SUMMARY	 Immunovant Sciences Ltd. ("Immunovant") and Health Sciences Acquisitions Corporation ("HSAC," Nasdaq: HSAC) have entered into a definitive business combination agreement Immunovant is a clinical-stage biopharmaceutical company focused on enabling normal lives for patients with autoimmune diseases HSAC is a special purpose acquisition company sponsored by RTW Investments Upon the closing of the transactions, HSAC will change its name to Immunovant, Inc. Expected post transaction equity value of \$556 million, assuming HSAC share price of \$10/share and no redemptions from the HSAC shareholders Transaction expected to close in December 2019
PREMIER INVESTORS AND ALIGNMENT OF INTEREST	 Provides Immunovant with a blue-chip investor base and cash resources to continue development of IMVT-1401, a compelling asset within the FcRn drug class Shareholders of the combined company expected to include Roivant Sciences, RTW Investments, and leading biotech investors including BVF Partners, Adage Capital Management, Cormorant Asset Management, Eventide Asset Management, and Perceptive Advisors
USE OF PROCEEDS	 At the time of closing, the combined company is expected to have more than \$100 million in cash and cash equivalents, including proceeds from the completed \$35 million private bridge financing Funding expected to enable completion of Phase 2 program in myasthenia gravis, Graves' ophthalmopathy, and warm autoimmune hemolytic anemia Expected to provide runway into second half of 2021
KEY MANAGEMENT AND BOARD	 Combined company to be led by Immunovant Chief Executive Officer, Pete Salzmann, M.D., M.B.A. Anticipated directors: Frank Torti (Chairperson), Andrew Fromkin, Douglas Hughes, George Migausky, Atul Pande, Myrtle Potter, Pete Salzmann
ү іммилоv	ANT

At a glance: terms of transaction

Pro forma valuation		Sources of funds		Uses of funds			
Illustrative share price (per share)	\$	10.00	HSAC Cash in Trust \$115,341,558		Equity Issued to Immunovant Shareholders (inclusive of	\$430,000,000 \$106,989,301	
Non-redeemable shares outstanding ²	55,575,000		Immunovant Shareholder Equity Rollover	\$395,000,000	bridge financing) Cash to Balance Sheet		
		Bridge Financing	idge Financing \$35,000,000		\$8,352,257		
Equity Value	\$ 5:	55,750,000	Sponsor Promote	\$10,750,000	Sponsor Promote	\$10,750,000	
Cash provided by transaction	\$ 11	15,341,558 ¹	Total Sources	\$556,091,558	Total Uses	\$556,091,55	

Pro forma ownership with earn out to Immunovant and % total ownership

	Pro forma share price, per share						
_	\$10.00		\$17.50		\$31.50		
Immunovant	Shares	%	Shares	%	Shares	%	
 Immunovant Shareholders and Employees² 	39,500,000		40,688,136		41,392,216		
 Bridge Financing Investors 	3,500,000		3,500,000		3,500,000		
 Earnout Shares, cumulative³ 			10,000,000		20,000,000		
Immunovant Total	43,000,000	77%	54,188,136	78%	64,892,216	78%	
HSAC Sponsors⁴	1,075,000	2%	1,975,000	3%	2,875,000	3%	
HSAC Public Shareholders	11,500,000	21%	13,471,429 ^⁵	19%	15,150,7945	18%	
Pro Forma Diluted Shares Outstanding	55,575,000	100%	69,634,565	100%	82,918,010	100%	



Assuming no redemptions from the HSAC shareholders Calculated using \$395M pre-maney equity value, on a fully-diluted basis, includes imputed dilution from employee stack options and 10,000 shares of Series A Preferred Stack. Rolvant Scionces will record shares of Safets A Preferred Stack that provides for certain proportional voting rights for the election of directors. Bridge investors and immunovant current shareholders, but not employees, are entilled to receive a pro rata pation of earnouts. 1.8M sponsor shares are cancelled at \$10,000 (\$90,000 shares are canceled at \$17.50); no shares are canceled at \$31.50 Giving effect to warrant exercisable at \$11.50 per share, using treasury stock method to calculate fully diluted shares outstanding

Immunovant



Our vision: Normal lives for patients with autoimmune diseases

Our asset: IMVT-1401, a novel, fully human monoclonal antibody inhibiting FcRn-mediated recycling of IgG

Our strategy for IMVT-1401:

- <u>Be best-in-class</u> in target indications where anti-FcRn mechanism has already established clinical proof-of-concept
- <u>Be first-in-class</u> in target indications with clear biologic rationale and no known in-class competition

Our near-term value drivers: Four anticipated data readouts over the next 20 months

7

MIMMUNOVANT

Immunovant Leadership

Management Team





UCSF Medical Center

Pete Salzmann, MD, MBA









Robert Zeldin, MD Chief Medical Officer





Brad Middlekauff, JD General Counsel PDS Biotechnology kolltan MEDAREX

8

Board

- Frank Torti, MD, MBA Immunovant Chairperson, Vant Investment Chair, Roivant Pharma
- Myrtle Potter Vant Operating Chair, Roivant Pharma
- Andrew Fromkin



- George Migausky
- Atul Pande, MD

Y IMMUNOVANT

IMVT-1401: Program Highlights

IMVT-1401: A novel, fully human monoclonal antibody inhibiting FcRn

 Early evidence suggests that anti-FcRn agents could transform the treatment of autoimmune diseases mediated by pathogenic IgG antibodies

In Phase 1, IMVT-1401 generated compelling pharmacodynamic activity

- Clinically meaningful IgG reductions observed (78% IgG reduction at 680mg dose level)
- No difference observed between intravenous and subcutaneous formulations at equivalent doses

IMVT-1401 has been well tolerated to date

- No headaches reported in the highest dose multiple dose cohort tested
- · No treatment-related serious adverse events (SAEs) or dose limiting toxicities reported
- No confirmed cases of anti-drug antibodies in any subject in multiple dose cohorts

IMVT-1401 was designed from inception for subcutaneous (SC) injection

- · Requirement during development process
- · Phase 1 data suggest every other week or less frequent dosing achievable for chronic use



IMVT-1401: A Pipeline in a Product

Target Indication	Pre- clinical	Phase 1	Phase 2	Phase 3	Status
Myasthenia Gravis (MG)	ASC	END-MG			Phase 2a open for enrollment
Graves' Ophthalmopathy (GO)	ASC	END-GO			 Phase 2a open for enrollment Phase 2b open for enrollment
Warm Autoimmune Hemolytic Anemia (WAIHA)	ASCEND	-WAIHA			 IND submission expected in 2H 2019



IMVT-1401

IgG antibodies implicated in certain autoimmune diseases

Antibodies in healthy individuals

- Antibodies play an important role in immune defense against pathogens¹
 - Clearing bacteria, viruses, and other harmful organisms and substances
 - Eliciting an immune response that leads to inflammation
- IgG antibody subclass accounts for ~75% of antibodies in the plasma of healthy people¹

1.

2.

W IMMUNOVANT

Antibodies in autoimmune disease

- In many autoimmune diseases, IgG antibodies develop that can recognize and bind to normal tissues²
 - Targets may include cell-surface receptors or circulating proteins
 - Result is a harmful immune response that damages critical tissues and organs
- Predisposing factors may include genetic susceptibility, environmental triggers, and factors not yet known³



Leusen J.H.W., The Role of IgG in Immune Responses. Molecular and Cellular Mechanisms of Antibody Activity, 2013 Isabela S., et al. The role of autoantibodies in health and disease. romanian journal of morphology and embryology, 2016 Mariani SM. Genes and autoimmune diseases - a complex inheritance. Med Gen Med, 2004

IMVT-1401's mechanism shown to promote IgG degradation¹

FcRn prolongs the half-life of IgG²

- · FcRn intercepts IgG, which would otherwise be degraded in lysosomes
- The FcRn–IgG complex is then recycled to the cell surface and free IgG is released back into circulation

Inhibiting FcRn promotes IgG degradation²

- IMVT-1401 binds to FcRn, thereby preventing it from recycling IgG antibodies back to circulation
- As a result, IgG is increasingly delivered to lysosomes for degradation



See Phase 1 MAD/SAD data on slide 16 Derry C., et al. FcRn: the neonatal Fc receptor comes of age. Nat Rev Immunol, 2007 1. 2.

Broad range of potential applications for anti-FcRn mechanism

IgG-mediated autoimmune diseases where FcRn mechanism may be relevant:



Y IMMUNOVANT



Phase 1 SAD/MAD study design
IMVT-1401 produced clinically meaningful IgG reductions in Phase 1 study

Preliminary results from Phase 1 SAD/MAD cohorts

Single-dose administration produced dose-dependent IgG reductions

Mean total IgG reduction after single dose in healthy volunteers

Repeat dosing at 680mg SC resulted in a 78% IgG reduction without the need for IV induction

Mean total IgG reduction after 4 weekly doses in healthy volunteers



🏋 IMMUNOVANT

IMVT-1401 reduced levels of all four IgG subtypes

Preliminary results from Phase 1 MAD cohorts



Generally well-tolerated in Phase 1 study

Preliminary results from Phase 1 SAD/MAD cohorts

- · 99 subjects dosed to date through SAD and MAD portions of Phase 1
 - IMVT-1401: 77 subjects
 - Placebo: 22 subjects
- · Most common AEs were mild erythema and swelling at injection site
 - Injection site reactions were not dose or frequency related
 - Occurred at similar incidence for drug and placebo treated subjects
- · No headaches observed in 680mg SC MAD cohort
- · Albumin changes:
 - Dose-dependent, reversible, and asymptomatic albumin reductions observed
 - At day 28, mean albumin levels were 37.5 g/L in the 340 mg cohort, and 32.4 g/L in 680mg cohort (normal range 36-51 g/L)
- 2 SAEs observed in two separate SAD cohorts, both ruled unrelated to treatment by study investigator (cancer, appendicitis)
- Treatment-emergent ADA confirmed in 8% of IMVT-1401-treated subjects and 6% of placebo-treated subjects
 - No subject in MAD cohorts has developed a confirmed ADA response to IMVT-1401

Y IMMUNOVANT

Adverse events reported in Phase 1

Preliminary results from Phase 1 SAD/MAD cohorts

						Single-/	Ascending	Dose						Multi	ple-Asc Dose	ending
	Intravenous Infusion					Subcutaneous Injection				Subcutaneous Injection						
	0.1 mg/kg	100 mg	340 mg	765 mg	1530 mg	Placebo	0.5 mg/kg	1.5 mg/kg	5 mg/kg	340 mg	500 mg	765 mg	Placebo	340 mg	680 mg	Placeb
	n=4	n=6	n=6	n=6	n=6	n=8	n=3	n=6	n=6	n=6	n=6	n=6	n=10	n=8	n=8	n=4
MedDRA Preferred Term																
Abdominal pain									1					1		
Abdominal pain upper													2	1		
Abnormal sensation in eye					1					1						
Back pain						2					1		1	1		
Constipation						1								1		
Cough											1		2			
Diarrhea														2		
Dizziness						1							1			1
Dry skin													1		1	
Erythema							1								1	
Fatigue	1			1	1	1	1			1			1			
Headache	1	1	1	1	1			1	1	4	1		1	2		
Injection site erythema									5	1	5	6	7	8	7	4
Injection site pain											1			2		1
Injection site swelling									3		2	4	3	7	6	2
Insomnia									1					4		
Myalgia														1	1	
Nasal congestion									1		1		1	1		
Nausea									1	1			1		1	1
Ocular hyperaemia															2	
Oropharyngeal pain	1			1	2				1		1		1	2	-	
Pain in extremity					_	1							1	_		
Procedural complication								1		1						
Procedural dizziness					2						1					
Pyrexia			1	1	-				1							
Rash					2				2				2		1	
Rhinorrhea									1				2			
Sinusitis			1										1			
Somnolence		1							1							
Upper respiratory tract infection	1	1	1				3		1	1				1		
Vision blurred					1					4						

IMVT-1401 has been given as a convenient SC injection



MIMMUNOVANT

IMVT-1401 designed from inception to be a potentially class-leading SC injection

- Fully human monoclonal antibody Antibody development Antibody development by OmniRat™ by phage library screening Generated from Ligand/OMT's Screening from hybridoma OmniAb transgenic rat platform Screening from phage-display library - Affinity-matured 1A library (~3600 clones) - Affinity-matured 11H library (~3600 clones) shFcRn-immunized transgenic rats (~2500 hybridoma) - >400 antibody campaigns • ELISA and Nucleotide sequencing • FACS and Nucleotide sequencing ongoing that use OmniAb 32 Selected scFv molecules 18 Selected rat Fc-IgG molecules technology¹ Whole IgG conversion In silico immunogenicity - 12 clinical-stage antibodies in 44 Whole IgG molecules development¹ ELISA and SPR · FACS binding and blocking IgG1 backbone Fc-engineered to 17 Lead molecules reduce effector function IgG catabolism test in FcRn-humanized mice
 PK/PD study in cynomolgus monkey Optimized for SC delivery 2 Selected molecules + Fc engineering/ VH deglycosylation mutagenesis Current clinic formulation is 2 Candidate molecules 170mg/mL
 - Delivered by 27-gauge needle



MVT-1401

MMUNOVANT 1. Ligand 2019 Analyst Day presentation, presented March 12, 2019

Somni Ab

IMVT-1401 has the potential to deliver a class-leading profile

WIMVT-1401 attribute	Potential patient benefit
Clinically meaningful IgG reductions	 680mg SC weekly: 78% reduction after four dose 340mg SC weekly: 63% reduction after four dose
SC injection	Fast and minimally invasive
Simple dosing schedule	 No requirement for IV induction doses or lengthy SC infusions Provides option for at-home administration Fixed dosing, vs. weight-based, reduces potentia for dose miscalculations
Fully human antibody	Low risk of immunogenicity
Fc-engineered to reduce effector function	Low potential for unintended immune responses

IMVT-1401 and competitors' programs with subcutaneous (SC) injection or infusion

Company		argenx	ueb	ALEXION Varribooy	MOMENTA	Syntimmune
Anti-FcRn candidate	IMVT-1401	Efgartigimod (ARGX-113)	Rozanolixizumab (UCB7665)	ABY-039	M281	ALXN1830 (SYNT001)
SC administration regimen	SC injection	IV induction, followed by SC injection ¹	SC infusion given over 30-60 minutes ^{3,4}	SC injection		
IV induction dosing	N/A		N/A	N/A	Not in clinic	Not in clinic
SC dose	340mg SC weekly 680mg SC weekly	300mg SC weekly ¹	7mg/kg SC weekly⁵	200mg SC single dose ⁶	e	
Mean IgG reduction observed	63-78% after 4 doses	"Approximately" 50% after two IV induction doses followed by 8 SC doses ²	56% after 3 doses⁵ 68% after 6 doses⁵	~45%		

Note: data as of 9/24/2019 and is not based on head to head comparison studies



4.

5. 6.

UCB, ASH presentation, December 2017 UCB, press release, issued October 18, 2018 Affibody, PEGS conference presentation, April 2019

 1.
 Argenx, corporate presentation, August 2018

 2.
 Argenx, press release, issued June 14, 2018

 3.
 Kiessling P. et al, The FcRn inhibitor rozanolixizumab reduces human serum IgG concentration: A randomized phase 1 study Science Translational Medicine, 2017

IMVT-1401 for Myasthenia Gravis

Myasthenia Gravis overview

- Rare autoimmune disorder affecting an estimated 65,000 people in the US¹
- Characterized by weakness of voluntary muscles including ocular, facial, oropharyngeal, limb, and respiratory muscles¹
- 15-20% of MG patients will experience at least one myasthenic crisis over their lifetimes, a potentially life threatening acute complication²
- Disease caused by autoantibodies targeting the neuromuscular junction (NMJ)¹
- ~93% of patients have an identified autoantibody¹

1.

W IMMUNOVANT

- Anti-acetylcholine receptor (AChR) antibodies (~85%)
- Anti-muscle-specific tyrosine kinase (MuSK) antibodies (~8%)

Med Sci., 2016



Meriggioli M.N. & Sanders D.B. Muscle autoantibodies in myasthenia gravis: beyond diagnosis? Expert Rev Clin

Immunol, 2012 Sudulagunta S.R. et al. Refractory myasthenia gravis – clinical profile, comorbidities and response to rituximab. Ger

Unmet need persists despite availability of treatment options

Current treatment paradigm¹

	1 st Line	2 nd Line	3 rd Line	4 th Line
•	Acetylcholinesterase inhibitors Corticosteroids	Immunosuppressive agentsThymectomy	 IVIg Plasma exchange Immunoadsorption Rituximab (off-label) 	• Eculizumab

Unmet need

- ~10% of MG patients refractory to current treatments, while 80% fail to achieve complete stable remission¹
- · Existing therapies associated with significant side effects
 - Early line agents can lead to disease exacerbation and do not always prevent disease progression
 - Treatment for more advanced disease often requires invasive and burdensome infusions
- · Patients with anti-MuSK antibodies more likely to become refractory¹
 - ~50% of the refractory MG population, despite comprising <10% of the overall MG population
 - Newest treatment option, eculizumab, only indicated for anti-AChR positive patients

Mantegazza R & Antozzi C. When myasthenia gravis is deemed refractory: clinical signposts and treatment strategies. Ther Adv Neurol Disord., 2018

ASCEND-MG: Phase 2a study design





IMVT-1401 for Graves' Ophthalmopathy

Graves' Ophthalmopathy overview

- · Also called Graves' orbitopathy or thyroid eye disease (TED)
- · 15,000-20,000 patients with active GO in the United States per year
- Clinical features¹:
 - Proptosis
 - Eye pain
 - Double vision
 - Light sensitivity
- Can be sight-threatening²
- · Caused by autoantibodies that activate cell types present in tissues surrounding the eye2
- · Close temporal and pathobiologic relationship with Graves' disease



Bahn, 2010 Figure 1. Patients with Graves' Ophthalmopsthy Panel A shows a 59-year-old woman with excess proptosis, moderate eyelid edema, and erythema with moderate eyelid retraction affecting all four eyelids. Conjunctival chemosis (edema) and erythema with biateral edema of the caruncles, with prolapse of the right caruncle, are evident. Panel B shows a 40-year old woman with excess proptosis, minimal bifetral injection, and chemosis with slight erythema of the eyelids. She also had evidence, on slit-lamp examination, of moderate superior limbic keratoconjunctivitis.

IMMUNOVANT
 Davies T & Burch HB. Clinical features and diagnosis of Graves' orbitopathy (ophthalmopathy), UpToDate, 2018
 McAlinden C. An overview of thyroid eye disease. Eye and Vision, 2014

Anti-TSHR autoantibodies drive progression of GO and Graves' disease



- adipocytes1
- Activation leads to inflammation and proliferation .
- Autoantibodies against Insulin-like growth factor-1 receptor (IGF-1R) also identified²



- TSHR highly expressed on thyrocytes (cells that make up the thyroid gland)³
- Activation leads to increased production of • thyroid hormone³

IMVT-1401 could address GO and Graves' disease caused by any IgG autoantibody, whether against TSHR or IGF-1R

¥ IMMUNOVANT

Smith T.J., et al. Role of IGF-1 pathway in the pathogenesis of Graves' orbitopathy. Best Pract Res Clin Endocrinol Metab, 2013 Liaboe C.A., et al. An Introductory Tutorial and Overview of Disease - Thyroid Eye Disease (TED), 2016 Varewijck A.J., et al. Circulating IgGs may modulate IGF-I receptor stimulating activity in a subset of patients with Graves' ophthalmopathy. J Clin Endocrinol Metab., 2013 2. 3.

GO characterized by an active phase, followed by a static phase

Rundle's Curve describes natural history of disease¹



Limited treatment options for GO

Current treatment paradigm¹

1 st Line	2 nd Line	3 rd Line	Inactive disease
Corticosteroids	 Orbital radiotherapy Immunosuppressive agents 	• Rituximab (off-label)	 Orbital surgery

Unmet need

- · Currently no FDA-approved therapies for GO
- · Corticosteroids are not effective in all patients, and approximately one-third of patients will relapse
- Sight-threatening disease may occur in 3-5% of patients with Graves' disease²
 - Medical emergency requiring immediate hospitalization and evaluation for surgery²
- Up to 20% of GO patients require surgical invervention²



ASCEND-GO Phase 2 clinical program



ASCEND-GO 1

Phase 2a

- Trial ongoing in Canada
- Single arm, open label
- N=8
- 6 weeks of dosing
 - 680mg weekly x 2 doses
 - 340mg weekly x 4 doses

ASCEND-GO 2

Phase 2b

- Trial ongoing in USA and Europe
- Double masked, placebo controlled, randomized
- N=77
- 3 drug arms vs placebo
- 12 weeks of dosing

Y IMMUNOVANT

ASCEND-GO 1: Phase 2a study design





ASCEND-GO 2: Phase 2b study design





IMVT-1401 for Warm Autoimmune Hemolytic Anemia

Warm Autoimmune Hemolytic Anemia overview



- · Blood disorder marked by red blood cell destruction
- Estimated prevalence of 42,000 patients in US and 66,000 patients in EU¹
- · Presentation typically non-specific and occurs over several weeks to months
 - · Fatigue, weakness, skin pallor, shortness of breath
- Severe cases can be fatal²



Park S.H. Diagnosis and treatment of autoimmune hemolytic anemia: classic approach and recent advances. Blood Res., 2016 Roumier M., et al. Characteristics and outcome of warm autoimmune hemolytic anemia in adults: new insights based on a single-center experience with 60 patients. Am J Hematol., 2014

Limited options for treating WAIHA

Current treatment paradigm^{1,2}

1 st Line 2 nd Line		3 rd Line 4 th Line	
Corticosteroids RBC transfusion	 Immunosuppressive agents 	Rituximab (off-label) Splenectomy	

Unmet need

- Currently no FDA-approved therapies for WAIHA
- Only one-third of all patients maintain sustained disease control once steroids are discontinued
 - Majority of patients will require either long-term steroid treatment or additional therapies¹ -
- No clear guidelines on choice of treatment in patients failing treatment with corticosteroids ٠
- RBC transfusions are indicated in patients who require immediate stabilization, despite the fact that ٠ autoantibodies present in WAIHA patients may react against RBCs in the transfusion product^{1,2}



Salama A. Treatment Options for Primary Autoimmune Hemolytic Anemia: A Short Comprehensive Review. Transfus Med Hemother., 2015
 Park S.H. Diagnosis and treatment of autoimmune hemolytic anemia: classic approach and recent advances. Blood Res., 2016

ASCEND-WAIHA: Phase 2 study design





Immunovant Recap



Our vision: Normal lives for patients with autoimmune diseases

Our asset: IMVT-1401, a novel, fully human monoclonal antibody inhibiting FcRn-mediated recycling of IgG

Our strategy for IMVT-1401:

- <u>Be best-in-class</u> in target indications where anti-FcRn mechanism has already established clinical proof-of-concept
- <u>Be first-in-class</u> in target indications with clear biologic rationale and no known in-class competition

Our near-term value drivers: Four anticipated data readouts over the next 20 months

MIMMUNOVANT

IMVT-1401: Multiple anticipated near-term value inflection points

MG	 Phase 2a open for enrollment Top-line results of Phase 2a study expected in 1H 2020 Pivotal Phase 3 study initiation expected in 2020
GO	 Phase 2a open for enrollment Initial results of Phase 2a study in Q1 2020 Phase 2b proof-of-concept study open for enrollment Top-line results of Phase 2b study expected in early 2021
WAIHA	☐ IND submission expected in 2H 2019



